Essays on policy instruments and incentives in health care in Norway

Anastasia Mokienko



2020

Thesis submitted for the Degree of Doctor of Philosophy (PhD)

Department of Health Management and Health Economics

Institute of Health and Society

Faculty of Medicine

University of Oslo

Norway

© Anastasia Mokienko, 2020

Series of dissertations submitted to the Faculty of Medicine, University of Oslo

ISBN 978-82-8377-584-6

All rights reserved. No part of this publication may be reproduced or transmitted, in any form or by any means, without permission.

Cover: Hanne Baadsgaard Utigard. Print production: Reprosentralen, University of Oslo.

Summary

Equal access to quality health care services, efficient resource use and cost containment are Norway's health care policy objectives (1). To reach these goals, policymakers use financial incentives and organizational structures. Efficient use of these policy instruments requires that policymakers understand how health care providers and users react to them. This thesis contributes to this knowledge.

The first two papers present research into financial incentives. In 2008, policymakers changed the reimbursement scheme for radiology providers to cut costs. Paper 1 examines how the change supplemented the general practitioner (GP) gatekeeping role. Paper 2 evaluates how the change affected the provision of radiology at the municipality level in different regions and centralities, depending on difference in travel time between private and public radiology providers. Paper 3 studies the organizational structure of the Norwegian regular GP scheme where patients can change GPs twice a year. This paper identifies patterns in disenrollment among patients with chronic diseases. Such patterns could indicate otherwise unobserved care quality.

Paper 1 concludes that the payment system for specialist providers might serve as a rationing tool and supplement gatekeeping. Paper 2 indicates the reimbursement change contributed to reduced services for populations with only private providers nearby, patient reallocation from private to public providers, and a reduction in the difference between municipality centralities in their consumption pattern, but an increase in the difference between Regional Health Authorities. Paper 3 suggests that most patient groups tend to remain with GPs with a greater share of arthritis, asthma, and depression patients, which can indicate high quality care. The results are relevant for both researchers and policymakers interested in policy instrument development.

Acknowledgements

Today, upon completion of my PhD dissertation, I feel both relieved and thankful. I am grateful to all those people who were by my side in this uneasy, nevertheless interesting process of creating a piece, a contribution to science.

I remember when I was working on my bachelor degree, I always looked up to the PhD students. They were something out of this world for me. When I myself was admitted to a PhD program, I understood that to be a researcher you didn't need to be from a different planet, but you had to be curious and a very hard working and dedicated person. Another thing I have learned was that researchers are people who look on this world in a very detailed manner. Deep into the roots. You have to be able to ask questions, often outside of the box, and most importantly, you have to be patient. Patience is a virtue that researchers acquire. It takes time to answer a question. To write an article may take many months, sometimes years of remodeling and reassessing. To publish an article might take several journals and several rounds of reviews, and thus many months of work on top of writing the draft. The PhD period was a good school for me to develop the necessary skills and the ability to complete my projects in order to see them published. My supervisors, Tor Iversen and Knut Reidar Wangen were excellent role models for me on how to complete the projects.

My first two publications (papers 1 and 3), were written in co-authorship with each of them and it taught me a lot about the way you approach scientific writing, analysis, and the publication process so that I could apply this knowledge on my third publication (paper 2). I am thankful to Tor and Knut for their generosity in sharing knowledge, for support, feedback and discussions, and for introducing me to the science as a process and craft.

I have met many good people during my work at university. I also want to thank all of my fellow colleagues for being there for me and for the conversations and time spent together. In particular, Anita, Yvonne, Mathyn, Søren, Gudrun, Camilla and Liv for being warm and supportive and Hans Olav Melberg, Terje Hagen and Ole Berg for always being kind and helpful. I was blessed to have Anita as an awesome officemate. She has become a close friend and I am thankful, not only for the fun we had and for the uplifting energy we created together, but also for all our conversations, her support and brainstorming together, especially towards the end.

I also wish to express my gratitude to the Department of Economics, University of Oslo, for providing a strong academic basis on topics such as microeconomics and econometrics, and the Department of Health Management and Health Economics, University of Oslo for employing me, providing me with the opportunity to be part of insightful conferences and workshops and to participate in courses abroad. Furthermore, this PhD thesis would not have been possible without funding from the Research Council of Norway.

Finally, I wish to thank my family and friends for their enormous support; I am particularly grateful to my husband, Javier, and my mom for their patience and love.

List of Papers

Paper 1: Supplementing gatekeeping with a revenue scheme for secondary care providers. Iversen, T. & Mokienko, A. Int J Health Econ Manag. (2016) 16: 247. https://doi.org/10.1007/s10754-016-9188-2

Paper 2: Effect of a funding change and travel times on delivery of private and public radiology services in Norway: Register-based longitudinal study of Norwegian claims data. Mokienko. Submitted to BMC Cost Effectiveness and Resource Allocation.*

Paper 3: Disenrollment from general practitioners among chronic patients: a registerbased longitudinal study of Norwegian claims data. Mokienko, A., Wangen, K.R. *BMC Fam Pract* **17**, 170 (2016). <u>https://doi.org/10.1186/s12875-016-0571-3</u>

^{*}Paper 2 was published 16th October 2019: Effects of a reimbursement change and travel times on the delivery of private and public radiology services in Norway: a register-based longitudinal study of Norwegian claims data. Mokienko, A. *Cost Eff Resour Alloc* **17**, 22 (2019). https://doi.org/10.1186/s12962-019-0190-7

Table of Contents

Summary	3
Acknowledgements	5
List of Papers	6
Table of Contents	7
Abbreviations	10
Preface	11
1 Introduction	12
2 Institutional background	15
2.1 Organization of health care system	15
2.1.1 Health care in Norway	15
2.1.2 Primary care	16
2.1.3 Specialist care	17
2.2 Financing	
2.2.1 Patient copayments	
2.2.2 Private health insurance	19
2.2.3 Primary care physician	19
2.2.4 Secondary care physicians	19
2.2.5 Hospitals and secondary care	20
2.3 Radiology services	20
2.3.1 Structure	20
2.3.2 Financing	21
2.3.3 Regional variation over time	23
3 Concepts, theory and empirical literature	26
3.1 Policy instruments	26
3.2 Financial incentives	27
3.3 Organizational strategies	30
3.3.1 Competition	
3.3.2 GP market	31
3.3.3 GP switching	32

4 Aims of the papers
4.1 Context of paper 1 and 2
4.2 Aims of Paper 1
4.3 Aims of Paper 2
4.4 Context and aims of paper 3
5 Discussion
5.1 Discussion of the data
5.1.1 Paper 1
5.1.2 Paper 2
5.1.3 Paper 3
5.2. Discussion of the methods
5.2.1 Paper 1
5.2.2 Paper 2
5.2.3 Paper 3
5.3 Discussion of the results
5.3.1 Paper 1
5.3.1.1 Results
5.3.1.2 Discussion of the results
5.3.2 Paper 2
5.3.2.1 Results
5.3.2.2 Discussion of the results
5.3.3 Paper 3
5.3.3.1 Results
5.3.3.2 Discussion of the results71
6 Concluding remarks
Appendix
Table N1. Overview of the three studies completed for the dissertation: Aims and
Hypotheses*78
Table N2. Overview of the three studies completed for the dissertation: Data and
Methods*
Table N3. Overview of the three studies completed for the dissertation: Results
and policy implications*

xes
xes

Abbreviations

- ABF: Activity-based funding
- CAT: Computerized axial tomography

DRG: Diagnosis-related group

DT1: Type 1 diabetes

DT2: Type 2 diabetes

FFS: Fee for service

GP: General Practitioner

HELFO: Norwegian Health Economics Administration (Helseøkonomiforvaltningen)

KUHR: Control and payment of reimbursements to health service providers (Kontroll

og Utbetaling av HelseRefusjon)

MCs: Marginal costs

MRI: Magnetic resonance imaging

NHI: National health insurance

RHA: Regional Health Authority

X-rays: Radiography

Preface

I have been interested in the topic of intrinsic and extrinsic motivation for behaviour for a long time. In my early career, when I had to lead and motivate a team of colleagues, I questioned the definition of leadership. I continued this questioning in broader perspective when I started my PhD studies at University of Oslo while researching policy implications in Norwegian health care. I saw a commonality between leading a team of people and creating policies. In both instances, knowing one's own and one's team's strengths and weaknesses is necessary, as is creating incentives for certain actions or behaviours. In terms of health care policies, this idea means knowing the implications of various policy instruments and how the whole system works to create incentives for health care actors to behave in certain way.

1 Introduction

'Whilst we must assume that financial incentives are effective instruments, the Directorate finds that good solutions must also be promoted using other policy instruments, such as management and organization, clear lines of responsibility and correct prioritization on the basis of medical factors. it is an acknowledgement of the fact that one instrument alone cannot make a direct contribution to achievement of goals in all areas'. (Ministry of Health and Care Services of Norway pp. 8-9 in (2))

Norway's health policy is directed toward both efficiency and redistribution and fairness (p. 4 in (3)). The financing system in Norway supports three main goals for health care policies: increasing the quality of health care services (including accessibility regardless of where patients live), cost containment and effective use of resources (1, 4, 5). High-quality services are defined as those that are effective, safe, user-centred, and coordinated and are characterized by continuity, resource utilization, availability and fair distribution (4, 6).

Creating and implementing health care policies is complicated. The process starts when the government reports on policies to Parliament by issuing white papers or parliamentary reports (in Norwegian *melding til Stortinget* or 'St. Meld'). White papers are used when the government would like to present cases to Parliament without a proposal for specific decisions or new or amended legislation. These papers are usually used to report analyses, plans and ambitions within a particular policy area (7). For instance, the Ministry of Health and Care Services issued 'Future primary health care proximity and wholeness' (White paper 26 in 2015), yearly papers 'Health Care Quality and Patient Safety' (White Papers 11, 12, 13, 6 for 2013, 2014, 2015, 2016 correspondingly), and White Paper 23 from 1997 about Regular General Practitioner Scheme (5, 8-12).

After Parliament approves the white paper from the Ministry of Health and Care Services, the Ministry and the Norwegian Directorate of Health 'translate' it to the language of legislation proposals, incentives, produces decisions and guidelines, and creates incentives for the health institutions to behave in a desired way.

To determine which incentives to use, the policymakers should clarify the effects and outcomes of the actions: when they change some constraints, financial or organizational, the market for health care services adjusts. The aim of this thesis is to contribute to the evidence on the impact of financial and organizational incentives and policy instruments. I dedicate the first two papers to studying the outcomes of the cost containment policy for radiology providers and the third paper to studying disenrollment patterns of the patients with chronic diseases as a possible indicator of high-quality care.

The first two papers in this thesis focus on extrinsic or financial incentives and their outcomes. These sections study the change in the remuneration scheme for radiology providers in Norway in 2008. The goal of policymakers was to cut costs in radiology. Thus, according to the Royal proposition for the state budget for 2008, the radiology budget was cut and the reimbursement scheme was changed from a 50/50 split to a 60/40 split (basic and variable parts accordingly) (13).

Paper 1 investigated whether the change may supplement the gatekeeping mechanism in regulating utilization of radiology services and studied outcomes for different modalities of radiology. Paper 2 examined how the change affected the provision of radiology on the municipal level in different regions and centralities depending on travel time between private and public radiology providers.

The second part of the thesis, the third paper, is dedicated to organizational structures or incentives. This section studies disenrollment patterns among patients with chronic diseases because they could indicate otherwise unobserved GP quality. In 2001, the introduction of the Regular General Practitioner Scheme aimed to increase quality and continuity of care and control over costs. To increase quality and competition between GPs, the patients were allowed to change GPs up to two times a year (12, 14). Patients choose their GPs and remain with him (her) or switch for various reasons, such as patient-GP relationships, GPs' qualifications, and patients' satisfaction with services, access, and GP referrals. All these factors affect patients' perception of service quality. We considered using patients' disenrollment patterns as quality indicators. As such, the third study elucidates disenrollment choices of patients with chronic diseases and contributes to potential quality identification in general.

The remainder of this thesis is structured as follows. Chapter 2 describes the institutional healthcare settings for primary and secondary care in Norway, including radiology providers and their financing options. The third chapter presents the concepts, theory and empirical literature as a background for the study. Chapter four provides a summary of the three essays' aims. Chapter 5 discusses the approaches in all three essays with regard to data, methods, and results. Chapter 6 provides a conclusion to the research.

14

2 Institutional background

2.1 Organization of health care system

2.1.1 Health care in Norway

Norway offers universal health insurance (i.e., all residents are insured through the taxation system). Residents pay a small consultation fee until they reach the threshold of 2,369 NOK (around 235 EUR) per year (15), above which treatment is free. Health expenditures amount to approximately 9.9 % of gross domestic product (GDP) (data for 2015) (16). National health insurance covers planned and acute primary, hospital and ambulatory care, rehabilitation and some drugs, as well as dental care for children up to 18 years and for certain disadvantaged groups.

This section is dedicated to policy-making and structure of health care in Norway. Policy is created when Parliament issues political decisions. The Ministry of Health and Care Services sets these decisions into the action with the help of legislation and documents guiding the work of the Directorate of Health¹ (16-18) (its executive agency), Regional Health Authorities and other agencies² under it (19-27). Figure 1 presents a flow chart depicting the organization of the health care system in Norway

¹ The Directorate is responsible for clinical guidelines, new health technologies, national quality indicators, reporting and learning system for adverse events in hospitals, fee setting for diagnosis-related groups (DRG) and projects on the strategies to ensure quality of care. In 2016, the Directorate of eHealth was established; its role is to develop information technology in health care.

² These agencies include the Medicine Agency that decides which medications to reimburse patients for up to a certain threshold; the Norwegian Institute of Public Health, which works with research and surveillance of public health, holds several registries (merged with Norwegian Knowledge Center for health services in 2016); Norwegian Board of Health Supervision audits the health care system in regard to complaints against both institutions and individuals, the Radiation Protection Agency System of Patient Injury Compensation, the Biotechnological Advisory Board, the Norwegian Registration Authority for Health Personnel (under HD) and the Health and care services ombudsmen who helps patients who do not receive care they need.

(16). In the next sections, I review the structure of primary and specialist health care, in particular the provision of radiology health services. Due to the thesis's focus, I do not address mental health care and long-term health care.



Figure 1. Organization of the health care system in Norway (borrowed from (16))

2.1.2 Primary care

The municipality provides primary healthcare. According to the Regular General Practitioner Scheme, implemented in 2001, each resident has the opportunity to be listed with a GP in the municipality (although some small municipalities share GPs). Almost all five million Norwegian residents are listed with a GP. The responsibility for

the Regular GP Scheme lies with the Norwegian Health Economics Administration (Helfo), which is a subordinate institution directly linked to the Norwegian Directorate of Health³ (28).

General practitioners decide the maximum number of patients they list and accept additional patients until the maximum number is reached. Each GP had on average 1,127 patients in 2015 (16). Patients are able to find a GP's availability and the maximum and current lengths of their patient lists. Patients can switch to another GP online, according to availability, up to two times annually. On average, 3% of patients choose to switch annually (29, 30).

General practitioners provide initial medical services other than those involving emergencies. Additionally, primary healthcare functions as a gatekeeping system for secondary healthcare. Thus, to receive coverage for specialist treatment or undergo an examination, a patient needs a referral from a GP (16, 31).

2.1.3 Specialist care

Norwegian Regional Health Authorities (RHAs) are responsible for specialist healthcare, including radiology diagnostics⁴ (32). In 2007, the number of RHAs was reduced from five to four, South-Eastern Norway, Northern Norway, Western Norway, and Central Norway. In total, RHAs are responsible for 19 public hospital trusts.

Inpatient specialist care is mostly provided by hospital trusts, although some is provided by contracted private facilities (32). Patients have free choice of specialists and

³ In addition, this directorate directs payments to health care providers, handles individual reimbursement for certain medicines, dental and health services abroad and issues European Health Insurance cards.

⁴ The responsibilities for ownership and financing were moved from 19 counties to the central government, and hospitals were organized as hospital trusts within RHAs during the hospital reform in 2002.

hospitals (33, 34). Outpatient care is delivered by (1) public providers, which are part of hospitals, and by (2) private physician specialists and private institutions with a lifelong RHA contract. In addition, some private providers contract with RHAs after a tendering process. Private specialists and private institutions account for around onethird of outpatient consultations. To see a public specialist or a specialist who contracts with an RHA, patients need a referral from the GP.

2.2 Financing

2.2.1 Patient copayments

The patient copayments constitute 155 NOK (15 euro) for a visit to an ordinary GP, 204 NOK (21 euro) to a GP specializing in general medicine, and 351 NOK (35 euro) to a specialist per visit in 2019 (35). X-rays copayment is 250 NOK (25 euro), while blood tests are 55 NOK (6 euro). A few groups are granted exemptions from these payments: patients with communicable diseases, children under 16 years old, mothers and children undergoing antenatal and postnatal follows up, patients with work related injuries, and young people under 18 years old who are under psychotherapeutic care (16, 35).

The patients pay their copayments directly to the provider until they receive an exemption card (in Norwegian *Frikort*). There are two exemption card schemes in Norway: one for user fee group 1 and one for user fee group 2. The first covers consultations at a GP, psychologist, hospital or laboratories and some medicines. The exemption card for user fee group 1 is generated automatically once a patient has paid more than 2369 NOK (in 2019) in user fees. The second card covers physiotherapy,

rehabilitation, some particular teeth and gum diseases and treatment trips abroad. This card is generated automatically when a patient has paid more than 2085 NOK in user fees. Afterward, the patient no longer needs to pay the provider because Helfo pays providers directly (15, 36).

2.2.2 Private health insurance

About 9% of the Norwegian population has private health insurance: 91% through the employer, and the rest buy it privately. Private health insurance ensures quicker access to specialists and a broader choice of private providers, as well as providing access to some treatments not offered in the public health service (16, 37). However, to see a specialist, except physiotherapists and psychologists, patients still need a referral from a GP if the patient wants it to be covered by private health insurance. Patient who choose to pay in full out of pocket can visit specialists without referral.

2.2.3 Primary care physician

There are two types of GPs: 5% are salaried and 95% are self-employed. Salaried GPs are more common in sparsely populated areas, as the salary removes their financial risk and improves GP recruitment at the periphery. Self-employed GPs receive a capitation fee from the municipality, a fee for service (FFS) from the state, and copayment from patients (each constituting approximately one-third of the GP's income) (38).

2.2.4 Secondary care physicians

Specialists based in hospitals are salaried. Privately practicing specialists that have contract with RHA are paid in the following way: (1) lump sum (35%), (2) fee for service (35%), and (3) patient fee (30%). The specialists that do not have agreements

with RHA are not regulated and set their prices themselves. Patients pay them directly and in full either out of pocket or with the help of private health insurance.

2.2.5 Hospitals and secondary care

Public hospitals are financed through RHAs. Somatic services are funded 50% by block grants and 50% through activity-based funding (ABF) based on diagnosis-related groups (DRG) (for 2015 (16)). Norway has used ABF for secondary healthcare providers since 1997 when the system of global budgeting was replaced by, partly, ABF. The remuneration schemes have changed several times since 1997; ABF reimbursement percentages varied between 40% and 60%: 55% in 2002, 60% in 2003 and 2005, and 40% in 2004 and 2006 (2, 32, 39). The purpose of ABF is to encourage achievement of activity targets. If these targets are not met, the RHA loses income. If the activity level exceeds the target, costs are partially compensated (p. 12 in (2)).

2.3 Radiology services

2.3.1 Structure

Since a greater part of the thesis is dedicated to radiology providers, I present the organization and financing of radiology services in this section. There are two types of radiology providers in Norway: private and public. Private providers operate as for-profit institutions, while public providers are hospital radiology departments.

Regional health authorities choose a number of private radiology providers via a tendering process and sign contracts with them for a number of services. These contracts specify the volume of the services and reimbursement; some specify an aggregated budget for services (40), while others are more detailed and specify a budget

for each type of service (e.g., ultrasound imaging, magnetic resonance imaging (MRI), computerized axial tomography (CAT) scans and X-rays) (41).

These private providers are part of the National Health Insurance (NHI) Scheme but also accept privately paying patients. Contracts with private providers specify that patients pay the same copayments to private and public providers if covered by the NHI; private laboratories can receive self-paying patients, but they should not be prioritized or examined at the expense of NHI services (42). According to interviews with managers of private radiology providers, these providers treat patients according to wait time and severity.

A patient has to receive referral to radiology examination from a GP or specialist to be covered by the NHI scheme. The referral can be for any available provider, public or private. Specialists and GPs can also recommend that the laboratory accepts patients without adhering to the typical wait time if they suspect serious issues. However, severely ill patients are often directed to hospital laboratories.

Generally, patients are added to a wait list when laboratories receive referrals. If patients choose to pay private providers entirely out of pocket, they receive examinations without adhering to the typical wait time. Usually, private providers' wait lists are relatively shorter than those of public providers, but there are variations according to region and service type.

2.3.2 Financing

For radiology services, the change to ABF occurred first in 2005 ((2, 43, 44)). According to Ministry of Health and Care Services of Norway, 'the purpose of the changes in the financing scheme is [...] that the regional health authorities should be given increased responsibility and a better opportunity to plan and prioritize the total provision of laboratory and radiology services in their own health region' (p. 248 in (45)).

On the 1st of September 2005, the financing of all radiology was changed to a 50/50 split between block grants (from RHA) and ABF (from Helfo). Afterward, private providers had to also enter into agreements with RHA to receive refunds and were required to produce an agreed number of services. However, they would still receive refunds from Helfo and patient copayments if they produced more services than agreed (43).

On the 1st of January of 2008, the proportion covered by the grant was changed from 50% to 60% for block grant and from 50% to 40% for ABF from Helfo. The aim of the change in 2008 was to harmonize financing of radiology provisions and specialist health care, encouraging achievement of activity targets and reducing spending (43). According to Norwegian Directorate of Health, 'On average, an activity-based funding share of 40% is regarded as being lower than the marginal cost. Nevertheless, the possibility of activity-based reimbursement at 40% covering the marginal costs in some cases cannot be ruled out. However, it is in conflict with the intentions behind the activity-based funding scheme if this consideration steers decisions as to which patients should be offered treatment first' (p. 13 in (2)).

Regarding public providers, RHA pays a fixed budget to the hospitals for their laboratories, which function as public providers. That portion of the budget cannot be distinguished from other budget components for each hospital (29). This system complicates control over the number of examinations provided. Therefore, relative to private providers, public providers have softer budget constraints (31, 46, 47).

Patients copay the same amount for services regardless of whether they visit a private or public provider as long as the provider is under the NHI scheme.

2.3.3 Regional variation over time

Figures 1A, 1B, and 1C illustrate the provision of radiology services in Norway in 2002 versus 2008. Figure 1A describes the total number of radiology examinations in years 2002 and 2008 in different health regions normalized per 1000 inhabitants (48). Figure 1B provides the same data but stratified by modality (48). Figure 1C depicts distribution between the private and public sectors, stratified by health region in 2002 and 2008 (48). The use of radiology services in 2008 became more uneven in different RHAs compared to 2002, increasing especially in the central RHA. The distribution of the modalities indicates that X-rays are the most popular service type in all RHAs, followed by CT, MRI and ultrasound. The distribution of the shares of the total number of the services performed at private providers has changed from 2002 to 2008, especially for South East RHA (a decline from 79% to 62%), West RHA (an increase from 15% to 27%) and Central RHA (an increase from 3% to 8%). Figure 2 depicts distribution of market share between private and public radiology providers on a national level over the years 2002 to 2009 (43), indicating that spending for private radiology continually increased until 2007 and started to decrease in 2008.



Figure 1A. Total number of private and public examinations per 1000 inhabitants in four different health regions: North, Central, West and South-East in 2002 and 2008 (Figure 3.3 in (48))



Figure 1B. Number of radiology examinations at private and public providers per modality (x-rays, computed tomography, ultrasound and magnetic resonance imaging), normalized per 1000 individuals per health region (North, Central, West and South-East) in 2008 (Figure 3.4 in (48))



Figure 1C. Share of total number of private and public radiology examinations in different regions (North, Central, West and South-East) that were performed by private providers in 2002 and 2008. (Figure 3.9 in (48))



Figure 2. Market share in the costs between private (blue) and public (red) radiology providers in percent (Figure 3.4 in (43))

3 Concepts, theory and empirical literature

3.1 Policy instruments

This chapter introduces some theoretical background and empirical literature about policy instruments and incentives in health care.

According to the recent guidelines for the development of Norwegian health care⁵, the main priority in Norwegian health care is to reduce costs and increase efficiency, improving the quality of care and making health care equally accessible regardless of patient group or geographic location (4, 5, 49-51). These goals are implemented through legislation and by creating incentives for the providers using policy instruments and by coordinating health care on national and local levels.

Public policy instruments are a set of techniques by which governmental authorities wield their power to support or prevent certain social changes ((52), p. 21). Organization (or organizational strategy) is a prerequisite for the application of the policy instruments ((52), p. 38). For instance, the fact that the patients are allowed to change GPs or to freely choose health care providers is an organizational strategy to promote competition.

Policy instruments are divided into (1) regulations (requirements and prohibitions), (2) economic means (financial incentives and disincentives and tools) and (3) information (transfer of knowledge, communication). Each of these categories can be affirmative or negative ((52), p. 250). Financial tools are rewards and penalties, or different structures

⁵ White Paper 34 (2015–2016) Values in the patient's health service - Report on prioritization, White Paper 11 (2015-2016) National health and hospital plan (2016-2019), White Paper 26 Future of Primary Health Care, National strategy for quality improvement for 2005-2015 and National strategy to reduce social inequality in health (2007)

of reimbursement (53-57). For example, the introduction of the ABF payment for radiology providers was a financial tool to cut spending. Non-financial tools, such as reputational incentives or elimination of informational barriers, may appeal to intrinsic motivation (58).

Both financial and non-financial incentive tools can be directed toward individual providers, institutions or patients (59, 60). Thus, policymakers implement policy goals by creating incentives for providers using payment mechanisms or changing certain organizational constraints (61-64).

3.2 Financial incentives

'The funding schemes are complex and fragmented. Objectives and policy instruments must be adapted to trends and developments in society and medicine. It is only natural that the funding schemes are under constant review and assessment'. (Ministry of Health and Care Services of Norway, pp. 8-9 in (2))

Kazungu et al. (2018) highlight that purchasing decisions are the base of universal health care. Purchasing decisions include three main areas: (1) what health services to buy, (2) what providers to use, (3) how to buy these services (i.e., payment mechanisms, price, contracts). Provider payment mechanisms (PPM) are crucial because they create incentives to attain access, quality, quantity and efficiency goals (53). Jegers et al. (2002) classify PPM in two dimensions: variable versus fixed and prospective versus retrospective (65). Kazungu et al. (2018) elaborate and define six main PPM categories: 1) global budget (a prospective payment with total flexibility on how to

spend it to deliver a set of services); 2) line-item budget (a prospective payment to spend on specific itemized services); 3) fee-for-service (FFS) (a retrospective activity-based reimbursement method for each provided service); 4) capitation (a fixed amount of money prior to service delivery to provide agreed services for each registered individual over a fixed period); 5) case-based or diagnosis-related groups (DRG groups) (a fixed amount per case such as for each diagnosis, admission or discharge); 6) pay for performance (a payment after the providers meet certain performance thresholds based on predetermined measures) (53, 66-68).

In Norway like in most countries there is a combination of the PPMs. For example, Norwegian hospitals recieve a global budget and DRG-based payment (with recent initialization of pay for performance); primary and secondary care physicians in Norway use FFS and capitation.

Empirical evidence indicates that healthcare providers react to financial incentives through PPMs (66). For example, two review studies by Kazungu et al. (2018) and Gosden et al. (1999) found that quantity of health care services (like hospitalization, number of procedures, number of diagnostic consultations, and number and time of clinical consultations) is reduced under capitation but increases under FFS (53, 61) Thus, regarding payments to GPs and specialist care, most research papers indicate that FFS creates greater incentives than salary does for providing services (64, 69-77) and capitation creates more incentives for referrals to specialist and hospital care (53, 78). In addition, Holte et al. (2016) in an experimental study found that GPs value losses from their current income level around three times higher than equivalent gains (79). Pedersen and Jarbol (2012) and Pedersen and Gyrd-Hansen (2014) also found that GPs are willing to change their practice location (for example to move to rural areas) or

practice organization (such as to shared practice) if they are compensated financially and with non-pecuniary benefits (80, 81). However, another study found that GPs appreciate improvements in non-pecuniary attributes more than increases in income (82). There is also a difference between genders: while male GPs prefer the ABF system, female GPs prefer salaries (83, 84).

A new systematic review from 2019 indicates that GPs often respond to reimbursement incentive when delivering cancer care (85). For example, they may increase the volume of surgical procedures when surgical fees are increased (86, 87) or favour higher priced anti-cancer agents when reimbursed at a higher rate (88-90).

Changes to the reimbursement schedule also affect patient treatment for mental health care providers. Douven et al. (2015) found that after the introduction of a new reimbursement schedule (discontinuous discrete step function), self-employed mental healthcare providers treated patients longer to reach the next threshold and obtain a higher fee (91).

Considerable research has been conducted on financial incentives in hospitals, and although they are not the topic of this dissertation, I mention a few previous results because they also indicate that financial tools create certain incentives in providers' behaviour.

A study of the change to prospective payment in Medicare for inpatient rehabilitation facilities in 1997 to 1999 indicates that changes in payment lead to a reduction in costs and hospitalization duration (92). In Scandinavian countries, switching to ABF resulted in increased technical efficiency, and in some instances increased patient satisfaction, but not increased cost efficiency (93-95).

The results of the research on changes in reimbursement and diagnosis-specific prices for DRGs suggest that hospitals upgraded patients to diagnoses that provided greater reimbursements (96-99). This effect was particularly strong in for-profit hospitals (see review study of private and public sector in (100)), which supports earlier findings from Medicare about patients' upcoding to diagnoses with larger reimbursement, but provides no evidence that hospitals admitted more patients, increased the intensity of care or changed duration of hospitalization or actual costs (99).

Recent evidence from Norway from 2016 indicates that a 10% increase in DRG prices leads to a 0.8% to 1.3% increase in the number of patients treated with medical DRGs. However, the number of patients treated with surgical DRGs remained the same (98). An earlier Norwegian study from 2010 on the change to ABF funding in 1997 found that ABF change did not favor the most efficient hospitals, but it has contributed to reducing wait time and increasing the number of services in less efficient hospitals (39).

3.3 Organizational strategies

3.3.1 Competition

One of the organizational strategies is competition framework. Within health care, competition occurs both between providers and between insurers. Competition between insurers is more relevant for markets like in the USA, while competition between providers can occur on several levels: primary, secondary, tertiary health care; inpatient and outpatient (101).

American and European studies on competition have diverse outcomes. Results based on American data indicate that competition and hospital care quality are negatively correlated (102-104), while results based on European data suggest competition and hospital care quality are either insignificantly or positively correlated (31, 105-110). These results differ primarily because of the difference in the organization and regulation of the health care systems between the two regions. The key is the price regulation. As theoretical evidence demonstrates when price is regulated, the competition raises its quality (or any of its available indicators) (101, 102, 111-113).

In the tax-financed health care systems, like Norway, there is little natural competition, and therefore, policymakers impose organizational structures to promote competition, for example by allowing patients to freely choose specialist or hospital and to change GPs up to twice a year. The next two sections are dedicated to the GP market and GP switching, as this dissertation focuses on the GP market in two essays.

3.3.2 GP market

The Norwegian model of GP system organization, allowing patients to switch GPs up to twice a year, provides the foundation for the competition between GPs, both to keep existing patients and to increase patient lists. The main incentive of GP competition is to improve performance and quality of services (105, 114-117). In addition, from a policy point of view, competition between GPs is an instrument to reduce growth in health care costs by inducing substitution of expensive hospital or specialist care through less costly primary care (101, 105).

Norwegian studies over time have found several important outcomes of the competition between GPs in regard to gatekeeping.

Gatekeeping is detrimental to patient satisfaction (118). The GPs' gatekeeping role operates when the only way to receive specialist healthcare is by obtaining a referral from a GP. Research suggests that gatekeeping reduces healthcare costs and 'unnecessary' interventions. Additionally, GPs have a better overview of quality and
availability of secondary healthcare and thus can be better agents (64, 119-121). Simultaneously, increased competition between GPs could result in less strict gatekeeping because of the capitation system, as GPs attempt to satisfy patients to ensure these individuals remain on their patient lists (31). A study by Carlsen and Norheim (2003) indicates that GPs under the patient list system are less concerned with the gatekeeper role and more with providing better services to keep the patients (109). Several continuous studies by Iversen and Lurås, from 2000 to 2005, indicate that GPs with patient shortages likely provide more services per patient and have higher incomes per listed person (114, 115, 122, 123). Lurås (2007) found a negative association between patient shortage and patient satisfaction with general practitioners (124). A study by Kann et al. (2010) found that GPs prescribe 3% more reimbursement drugs in the areas with higher competition than in the areas with lower competition; a shortage of patients on GP lists also had impact. However, the longer the patient list, the lower the impact (125).

A study by Godager et al. (2015) suggests that competition has a small positive or negligible effect on overall referrals. The researchers did not support the policy claim that increased competition increases the number of the referrals. One arguments was that GPs earn more by treating the patients themselves (106). Supporting it, a newer study by Islam and Kjerstad (2017) found that intensified GP competition may reduce inpatient hospital admissions by inducing GPs to provide more services and may increase outpatient admissions (105).

3.3.3 GP switching

One way to increase competition between GPs in Norway is to allow patients to switch GPs, incentivizing the GPs to improve the quality of their services. This section

describes patients' and GPs' characteristics regarding patients' preferences and reasons for switching.

A survey study by Billinghurst Whitfield (1993) about the reasons patients change to a GP indicated convenience as the most common reason (53%), followed by recommendation or reputation (36%) and positive expectation of service (37%) (126). The most common reasons to switch away from a GP was dissatisfaction with the doctor, such as loss of confidence, lack of interest in patients and their concerns or rudeness. Criticisms of practice organization were mainly focused on the lack of continuity, long wait lists and unhelpful receptionists (126).

Patients reported greater satisfaction with providers characterized with 'personal' care (64, 127-129). Patients also preferred GPs with observable characteristics similar to them, like age group and gender (130). Patients disenrolled less often from younger GPs and female GPs (124, 131).

In general, patients tended to switch more often from GPs who were already short on their number of patients (124, 132). However, patients without chronic diseases preferred GPs with shorter lists (more availability), while patients with chronic diseases preferred GPs with full lists (more popular and associated with higher disease detection) (131, 133-135).

Three groups of patients were found to switch GPs more frequently: patients belonging to racial and ethnic minority groups, those who use information to choose their physicians and those who switched physicians during the preceding five years because of dissatisfaction (136). Other factors influencing GP switching include sex, educational level, self-assessed health status and GP capacity at the municipality level. Older and wealthier patients switched their physicians less frequently than younger and less wealthy patients. Female patients and patients with chronic diseases, higher educational levels or fair or poor health tended to switch GPs more frequently than other patients (137).

A Japanese study indicated that chronic illness and patient-physician relationships are the main contributors to GP switching behaviour (138). Patients with chronic diseases visit their GPs frequently and are therefore well informed about healthcare quality. This pattern could be a further incentive to switch GPs if the patient is dissatisfied (the positive relationship between dissatisfaction with quality and GP switching is supported in (124, 139, 140)). However, this idea contradicts the theory that patients with chronic diseases experience higher costs, relative to those of other patients, when switching GPs and, therefore, are supposed to be less prone to changing their GPs. These costs include the cost of learning about new physicians, psychological costs resulting from disloyalty and costs related to uncertainty regarding the quality of untested brands (136, 137, 141).

Patients without chronic diseases use fewer healthcare services, are less informed about quality and benefit less from high-quality GPs than patients with chronic diseases. Therefore, these patients might be less concerned than patients with chronic diseases about the doctor they visit and less motivated to switch GPs (137).

4 Aims of the papers

4.1 Context of paper 1 and 2

The first and second studies are set in the context of the reimbursement change of 2008 for private and public radiology providers in Norway (described in the Section 2.3.2).

The reimbursement scheme was changed to keep spending on radiology within a certain budget and harmonize the financial scheme and system for financing outpatient medical services in Norway. Between 2005 and 2008, the ratio of ABF and block grants for radiology providers was approximately a 50/50 split and changed to a 40/60 split thereafter (13, 32). The new 40% ABF was intended to set spending limits by discouraging laboratories from exceeding contracted volumes (2, 13).

Both papers study the implications of the reimbursement change of 2008. Paper 1 investigates whether this reimbursement change supplements gatekeeping and involves service rationing to private radiology providers. Paper 2 examines the variation in the impact of the reimbursement change, depending on the difference in travel times to private and public providers in different municipalities.

These studies rely on existing research on the reimbursement changes for secondary healthcare providers (discussed in the previous chapter). While there is vast research on reimbursement changes for the hospitals, outpatient service providers do not benefit from the same attention. Papers 1 and 2 fill this gap. In addition to it, to my best knowledge, no other paper has studied this particular change.

4.2 Aims of Paper 1

Paper 1 studies the result of this reimbursement change and its connection to gatekeeping and competition in the GP market. The study relies on the existing research on (1) reimbursement changes and (2) GP competition and gatekeeping and contributes to the knowledge of whether reforms in reimbursement can contribute to the roles of GPs as gatekeepers.

Information box 1The overall objective of Paper 1 is to determinewhether the change in the revenue system forsecondarycareproviderssupplementsgatekeeping mechanisms.

The theory model argues how individuals are rationed and that individuals with least expected benefit are rationed first. Based on this theory model, we develop hypotheses.

The hypotheses are as follows:

- 1) With less FFS, we expect a reduction in the volume of radiology services performed by private providers
- 2) If post-2008 rationing occurred according to declining benefits, we would expect a greater reduction in number of examinations by private providers in municipalities with more competition for patients among GPs than in municipalities with less competition for patients among GPs.
- 3) We hypothesize an increase in the number of examinations performed by public providers and a greater increase in number of examinations by public providers in municipalities with more competition for patients among GPs than in municipalities with less competition for patients among GPs.
- 4) Because MRI examinations to a greater extent than the other modalities are located within the gray-area, we hypothesize a greater decline in the total number of MRI scans than for the other modalities and also a greater decline in municipalities with more competition for patients among GPs than in municipalities with less competition for patients among GPs.

Table N1 in the appendix provides an overview of the aims and hypotheses for all three papers.

4.3 Aims of Paper 2

The number of radiology services varies within regions and municipalities. Understanding how this variation works and how it affects political and financial changes helps policymakers make more thorough decisions (Paper 2 (29)). Considering three factors is important. First, one of the policy goals in health care is greater geographical equitability in the range of services (51). Second, Norway's sparse population is unevenly distributed. Third, the distribution of private and public radiology providers is also uneven.

Paper 2 relies not only on the earlier research on financial incentives (described in the previous chapter) but also on research about geographical variation in the consumption of the health care services. Pedersen et al (2012) indicate that distance to the health care providers is an important factor for patients (142). McGrail et al (2015) suggest that geographical tolerance and the distances to providers affect the consumption of health care services because geographical tolerance differs between densely and sparsely populated communities (143). Residents from densely populated areas are less willing to travel to access a GP than residents from sparsely populated areas. For example, 41% of residents in sparsely settled communities were willing to travel for one hour to see a GP, while only 3% of residents from densely populated communities were willing to do the same (143). Research indicates that greater travel distances lead to reduced utilization of healthcare services (144-147). These two findings combined indicate that populations living greater distances from healthcare institutions visit these institutions less frequently but are willing to travel farther than those living in populated areas.

Moreover, research suggests that patients consume more services from nearby providers (31).

Paper 2's research fills a gap in the literature by studying relative distances to the private and public providers based on their institutional differences and differences in the reactions to the reimbursement change. This is an important topic in light of the priority in Norwegian health care to create greater geographical equitability in health care.

Information box 2

The overall **objective of Paper 2** is to *examine how the change in the remuneration system for radiology providers can contribute to a change in the radiology supply in different geographical regions, depending on the difference in the proximity of private and public radiology providers.*

The hypotheses are as follows:

1) There will be a larger decrease in the number of private services than public services based on the differences in their budget constraints.

2) The stream of patients who move between providers and the effect on the total number of services will be different depending on the difference in the proximity of private and public radiology providers. The changes at private, public and both providers will be following:

2A) Patients use private radiology more when these providers are relatively closer (i.e. *Time_difference* is negative or equal to zero), which means that, after 2008, the greatest reduction in the *Priv_Serv* will be in these areas. The reduction diminishes with the increase in *Time_difference*.

2B) The change for public providers consists of two effects. The first involves a reduction in the original public service users. The greater usage was before 2008, the greater the reduction in the number of services will become after 2008. In general, patients use public radiology more when these providers are closer (that is, when *Time_difference* is zero or positive). The second effect relates to users switching from private radiology. These patients are more likely to switch if they live closer to a public provider compared to a private provider (i.e. the greater the value of *Time_difference*). Depending on what effect is greater, the change will be positive, negative, or equal to zero.

2C) Since private providers are more affected, the greatest reduction in the total number of services occurs in the areas with negative *Time_difference*. This reduction will diminish with an increase in *Time_difference* because patients can more easily switch to a public provider.

4.4 Context and aims of paper 3

The third study relies on existing research for disenrollment patterns, and its connection to perceived quality and GP attributes. However, existing research does not cover the connection between disenrollment and patient list characteristics, except for list length. Paper 3 contributes to existing research by investigating how chronic patients' switching behaviour is connected to GP list composition, indirectly shedding light on the assessment of primary care quality. The aim was to investigate patterns of chronic patient disenrollment.

Information box 3

The overall **objective of Paper 3** is to *identify* patterns in disenrollment among patients with chronic diseases because such patterns could indicate otherwise unobserved GP quality.

The published version of the paper does not have a defined hypothesis. However, we proposed expectations, which functioned like hypotheses:

1) If patients switch between GPs until their demands are met, we expect these patients to be disproportionally distributed across GPs.

2) If disenrollment patterns of special groups of patients align with the pattern of other patient groups, then they might be used as quality indicators.

5 Discussion

5.1 Discussion of the data

Literature indicates many different approaches regarding data level choice. Most importantly, an appropriate data level, whether state, country, municipality or provider, depends on the research question (148-152). Thus, in this dissertation, I used several levels of data for the convenience of approaching research questions. An overview of the data and variables for all three studies is provided in the Table 2N in the appendix.

In the first paper the aim was to investigate whether the reimbursement change could complement gatekeeping, and thus, the choice fell on data on the GP level that most accessibly describe the GP market.

In the second paper, the aim was to examine how the proximity of the providers affects the number of services consumed in different municipalities after the reimbursement change, and as well to discuss whether the reimbursement change would contribute to or reduce geographical variation between municipalities and RHAs. Thus, I chose data on the municipality level. I could eventually use data on patient level to study characteristics of the patients as well. However, I did not have precise distances from the patients' homes to the health care providers, only the distances between the patients' municipalities to nearest private and public providers' municipalities. This limitation helped me to focus on the municipality level.

In the third paper, the aim was to study the connection between the patient's disenrollment choices and different characteristics of the GP's patient list. Thus, the most appropriate was to use data on patient and GP levels.

5.1.1 Paper 1

The study uses monthly claims data regarding private and public radiology services between 2007 and 2010 that were provided by the Norwegian Directorate of Health. Only examinations covered by NHI were included. Data were at the GP level, and information regarding both self-employed (95%) and salaried (5%) GPs were included. Claims data, data on GPs' characteristics and patient lists, and data on travel distances between municipalities of GPs and of providers were merged. Travel distance data was provided by Info Map Norway (153) and connected to our data through the municipality codes. After careful consideration, we decided to focus on self-employed GPs and remove salaried GPs since salaried GPs have the same salary regardless of the number of patients. We were left with 176,709 monthly observations of 4486 self-employed GPs. Our data covered the entire population of self-employed GPs who had contracts with municipalities in Norway.

The data provide information concerning consumption of radiology services according to modality (number of services consumed per 1,000 patients listed with the GP), which were our dependent variables.

For the main independent variables, we used two types of competition indicators based on the previous empirical literature.

The empirical literature provided two indicator types introduced first by Iversen and Lurås (2002) (and later used in Iversen and Ma (2011) and Godager et al. (2015)) (31, 106, 116). Iversen and Lurås called the first indicator a 'micro' indicator and labelled it 'Shortage'. Shortage was a dummy variable that indicated whether a GP needed more than 100 extra patients to complete his (her) desired list size (e.g., shortage = 1 if (desired list size – actual list size) > 100). The second indicator was called a 'macro'

indicator and was labelled 'Supply'. Supply characterized the municipality, indicating the ratio between the sum of the desired number of patients for all the GPs in the municipality and the population of that municipality.

Thus, in Paper 1, we used the micro indicator in the same way as previous research: as a dummy variable Shortage, where 1 indicates the GP needs more than 100 patients to complete his (her) desired list size. We found that approximately 19% of the GPs had more than 100 free spots on their lists.

Regarding the macro indicator, we made several modifications, which were also used in Iversen and Ma (2011) and Godager et al. (2015) (31, 106). Overall, we used two types of macro indicators: number of open practices (#Open) and number of open practices relative to population size (#Open/Capita).

We also used several variables describing the GPs and their patient lists as control variables because these variables often influence the GPs' practice styles. For example, GPs who have more women on their lists would issue referrals more often connected to childbirth, prenatal, and postnatal periods than GPs without many female patients. Likewise, if an area has excess capacity among GPs, the referral rate might increase (31, 106). We used the following control variables for GP-patient characteristics: GP's age and sex, GP's specialization status, the GP's share of female patients and patients over 70, the presence or absence of patient shortage and GP's municipality (Table 1 in Paper 1 provide more details). We also controlled for travel time to the closest private and public providers. Table N2 in the appendix provides an overview of the data and methods for all three papers.

5.1.2 Paper 2

The second study used the same data set as the first study (claims data regarding private and public radiology services between 2007 and 2010 that the Norwegian Directorate of Health provided us, merged with GP data (both salaried and self-employed) and with the data on travel time between municipalities of patients and providers (provided by Info Map Norway (153)). This time, I aggregated all the claims at the municipality level (I removed GP characteristics data, as there was no use for them in the second study and left only the data on the municipality level, like centrality, region, distances and service consumption). I was left with 422 municipalities in 48 periods (monthly observations during 2007 to 2010), totalling 19,867 observations. Stratified summary statistics are displayed in Table 1 in Paper 2.

I used the monthly number of private and public radiology consultations per 1,000 individuals in the municipality as the dependent variable. This variable was calculated by accumulating services referred to and claimed in every municipality. Thus, if a patient from Municipality A went to take X-rays in Municipality B, the service was classified as a service to Municipality A.

The main independent variable for the regression analysis was $Time_difference$: $Time_difference = Privtime - Pubtime$ (i.e., the difference between travel times to the nearest private and public providers). Travel times were measured as the travel time by car between the patient's municipality and the municipality of each type of provider in hours. Thus, if the patient had a private provider in his (her) own municipality, *Privtime* was set to zero. If the patient had the public provider in another municipality a half an hour drive away, *Pubtime* was set to 0.5 hours. In this example, *Time_difference* = 0 -0.5 = -0.5 hours. To choose main independent variable was not easy. I considered and tried several options: 1) Using discrete intervals for travel times. I was attracted to this option because I could study different combinations of distances. However, there were two difficulties with this choice: choosing the thresholds for the periods and interpreting the analysis was confusing; 2) Using Publime or Privtime. This option had a simple interpretation. However, from earlier studies, I already knew that the closest providers are the most used ones. In addition, this option was not suitable for researching importance of relative proximity of providers (i.e. which provider is closer affects outcome too); 3) Using *Time_difference*. It was chosen to be main independent variable used in the regression analysis, because when deciding between private or public providers in the settings of unevenly distributed providers, patients often choose the more available provider in terms of proximity and, since private and public providers have different institutional settings, it affects outcome. Thus, according to the aims of the study, using *Time_difference* gave the clearest interpretation of the regression results. 4) Using *Time_difference* and one of the travel times (*Pubtime* or *Privtime*). This option was appealing but lead to confusion with regard to interpretation because Time_difference already included both Publime and Privtime. Thus, option 3) was used and the main independent variable for the regression analysis became *Time_difference*.

I complemented the analysis with stratified summary statistics and calculations over variation change, using municipalities' characteristics, such as centrality and the health region they belonged to. The *Centrality* variable was constructed by Statistics Norway and indicates how central the municipality is, where 1 is the smallest and most remote municipality and 7 are regional centres with well-developed infrastructures (154). Thus, this variable indicates, first, the level of infrastructure development (i.e., ease of access to care) and, second, the municipality size and, thus, whether there is a choice

of providers. Some literature indicates that the patients in the most densely populated areas visit health care providers more often than patients in the least populated areas, but patients from sparsely populated areas are willing to travel longer distances (143). Therefore, *Centrality* might be correlated with patients' willingness to travel. *Regions 1* through *4* are dummy variables describing whether the municipality belongs to South East (1), West (2), Central (3) or North (4) Regional Health Authorities.

5.1.3 Paper 3

The third study used a different dataset, consisting of combined panel data for six semiannual periods between 2009 and 2011, merged using the GPs' IDs: GP data (national register of regular GPs, covering the entire GP population) and patient data (claims data obtained from the KUHR registry (Kontroll og Utbetaling av Helse Refusjon) covering the entire Norwegian patient population). Figure 3 presents a flow diagram of the data and sample selection (155).



Figure 3. Flow diagram of sample selection in Paper 3 (155)

In general, we chose to focus on the choices of chronically ill patients because they need continuous follow up and their GP preferences might reveal unobservable

characteristics of GPs. We had seven patient groups: type 1 diabetes (DT1), type 2 diabetes (DT2), asthma, arthritis, schizophrenia, depression and epilepsy. These patient groups were chosen due to their variation both in represented numbers of patients and in utilization of healthcare. We used two samples. Sample 1 included patients with at least one of the mentioned diagnoses; Sample 2 included a comparison group containing the entire cohorts of patients born in 1940 and 1970, excluding patients in Sample 1 (see more in Table N2 and Figure 3). Thus, after excluding cases in which disenrollment was irrelevant to our study purposes (for instance, connected to a GP's or patient's move or GP's replacement) and cases in which patients were from municipalities with less than 50,000 inhabitants (i.e., without a wide choice of GPs), we were left with 313,659 patients in Sample 1 and 30,212 patients in Sample 2 (343,871 in total).

We are aware that the present comparison group gives different age distribution from the treatment group. It would have being ideal to use all cohorts or a random sample of these as a comparison group. However, due to privacy concerns from the registers, we could not access all the data. Alternatively, using only Sample 2 and splitting it would give us too few observations of chronic patients. Nevertheless, our samples were rich and contained all the patients registered with the chosen chronic diseases in Norway (Sample 1). While in Sample 2, cohort 1970 represented a random sample of young adults and cohort 1940 represented a random sample of elder adults.

The main dependent variable was *Switch out*, a dummy variable that indicated a patient switched GPs from one semi-annual period to another. In our data, 4.52% of patients in Sample 1 (chronic patients) and 3.76% in Sample 2 (comparison group) switched GPs between the first and second halves of 2009 (see Table 2 in Paper 3 (155)).

The independent variables include information about the GPs' age, sex, specialization, and list length and patients' sex, birth year, and number of visits, as these variables may affect a GP's practice profile and a patient's preferences. These variables were chosen because earlier research indicated their importance for patient-GP relationships (130, 131, 133-135, 137, 138).

Table 1 in Paper 3 provides descriptive statistics on the patient level, where GP's characteristics denote the characteristics of the GPs with which particular groups of patients (e.g., DT1, DT2) are enrolled. (155)

In addition, we created a control variable *Pat_Comorb*, indicating number of the patients' comorbidities (0 to 6). For this variable, we counted how many of the remaining six diagnoses the patient had. Earlier research indicates that number of comorbidities is an important variable because it underlines the importance of appropriate management of chronic conditions and care coordination since those conditions often must be addressed by different specialists (136, 156, 157).

The variable *Diagnosis_Share* (where *diagnosis* is any of the seven diagnoses from the list; see Table 1 in Paper 3) was the main research variable. This variable indicated the share of patients with certain chronic diagnoses on the GP's list from each patient's perspective (i.e., for every GP-patient pair, we excluded the patient when calculating the GP's share). The other possibility was to calculate this figure from the GP's perspective, i.e. to find the share of the patients with a certain diagnosis from the total number of patients.

For example, a patient with epilepsy, which is a rare diagnosis, might be the only epileptic patient enrolled with a GP who has 100 patients on her list. From the GP's perspective, the share of patients with epilepsy is 1%, but this figure is of little relevance

to the patient because, besides him, there are no other epilepsy patients on that list. Thus, from this patient's perspective, the share of other epilepsy patients on his GP's list is zero.

In the data analysis, we used patient level data, and therefore, we chose the patient's perspective to calculate this variable. A potential limitation is that the same GP might have been registered with different values for *Diagnosis_share*, depending on the diagnoses of the patient we examined. Thus, the values for *Diagnosis_share* could vary in the entries for the Sample 1, while for Sample 2, these values would be identical for identical GPs.

5.2. Discussion of the methods

5.2.1 Paper 1

In this section, I chronologically present and discuss the methods used in each study. The overview of all three papers' methods is displayed in Table 2N for easier understanding.

In Paper 1, in the regression analysis, we employed the fixed effect model with GP competition variables at the municipality level: #Open and #Open/capita. We performed regressions for all modalities combined and for each type of modality. Since many GPs in the same municipalities and hospital catchment areas communicated on formal and informal levels, their practice profiles could have been correlated. Therefore, to calculate standard errors, we clustered them by hospital catchment areas (21 areas) and checked for robustness by changing the cluster level to that of municipalities (395), changing weights from populations of municipalities to patient lists, and then dropping weights entirely.

I had a few considerations about weighing since robustness check showed some variation in the results. In general, the weighting gave the observations with smaller variance more weight because these observations provided more accurate information than those with large variation. However, since we used clusters on a group level, the weighting could have reduced precision. This reduction can occur when the group average effect is large and fairly homoscedastistic. In this case, weighting can impose heteroscedasticity and unnecessarily increase standard errors (158, 159). Thus, using weighting might not have been efficient for our model specification. The other thought was given to types of weights: first, municipality population, then GP's list length and then no weights at all. The model specification with #Open/capita included municipality population. To obtain a result by weighting municipality population

increases precision in this particular model specification. Therefore, using other weights or no weights on this specification would naturally reduce precision.

5.2.2 Paper 2

Paper 2 employs several approaches to demonstrate geographical variation in the provision of radiology services and the reaction of radiology providers to the reimbursement change.

First, the paper presented stratified summary statistics for the number of services at private and public radiology providers and time difference before and after 2008 (12 month before and 36 month after) according to (1) centrality and (2) to RHAs (see Tables 1BC in Paper 2).

Second, the study presented fixed effect linear regression estimations (see Table 2 in Paper 2). The fixed effect model was based on standard assumptions that the errors were uncorrelated with the independent variables and that the errors were conditionally homoscedastic and not serially correlated (160). I expected that the relationship between the number of the services and the time difference might not be linear because patients have different consumption patterns with regard to different distances to providers and their own centrality (143-147). After trying several polynomial functions, I chose a quadratic function. I estimated three regressions for number of services at private, public and both private and public providers. I also considered the random effect model, but after testing it against the fixed effect model with Hausman test, I chose the fixed effect model. I could have also tried clustering the variables by hospital catchment areas for a robustness check, to control whether the municipalities had

certain common policy programs or other traits shared in the hospital catchment areas. Also, controlling for life expectancy and socio-economic factors in the municipalities may have been beneficial since, in Norway, there is up to 10 to 12 years difference in life expectancy between men living in municipalities with the highest and lowest life expectancies, and much of this difference is due to variations in education, living standards, income, institutions (schools, workplaces) and proximity to natural resources (161). However, during this short time span, these characteristics were included in municipality fixed effects.

Last, the study presents a numeric analysis to measure the variation between different centralities and RHAs before and after 2008, based on the stratified summary statistics using 'mean value range'. A limitation of the method is that I did not consider the entire distribution of the radiology services. This method is not sophisticated, but it illuminates change in the variation. The method was inspired by 'observations range', or the difference between the highest and lowest observations. Observations range is easy to compute and understand, but it ignores all but two of observations, does not weight observations and is skewed by outliers (162-164). Mean value range, however, includes all the variables in the measure and is still easy to compute and understand and it describes change in the variation between centralities and RHAs.⁶

⁶ Other popular measures under consideration were range ratio, the coefficient of variation and the Gini coefficient. However, they were ruled out because they were more difficult to interpret in the study's context.

5.2.3 Paper 3

We used both graphical and statistical analyses for Paper 3. First, we used graphs to analyse whether the distributions of the patients with chronic diseases were disproportionate across GPs. To accomplish this analysis, we used GP level data for the first quarter of 2009 and examined the proportion of patients with a particular diagnosis and a 95% confidence interval (we did the same procedure for every diagnosis, but illustrated it for DT2 in the paper).

To explain the results of the distribution, we subsequently calculated Spearman's correlation coefficients between GPs' characteristics (age, sex, list length, specialist status) and shares of patients with different diagnoses. We used data on GP level for the first quarter of 2009 (Table 4 in Paper 3).

We also calculated proportions for comorbidities (i.e., the probability of having one more chronic diagnoses when a patient already has a certain chronic diagnosis). We used patient level data for the first half of 2009 (Table 3 in Paper 3).

Finally, we performed a random effect (RE) logistic regression analysis of voluntary disenrollment for each diagnosis subsample (from Sample 1) and for the control cohorts (Sample 2) (for an overview of the random effect logistic regression method see (165, 166)). We used standard assumptions for a random model that patient specific effects were normally distributed and not correlated with independent variables. There we used the data on patient level described in Table 1 in Paper 3. We focused on patients in municipalities with more than 50,000 inhabitants, to make sure that patients have a sufficient choice of GPs.

We also considered other options for the regression analysis. First, we considered the fixed effect model, which allows patient-specific effects to be non-normally distributed or correlated with independent variables. Testing this model against the RE model with the Hausman test indicated that we could use the RE model. In addition, the RE model allowed us to include such time invariant variables as sex and birth year, which would have been excluded in the fixed effect model.

Second, we could have used multilevel mixed effects logit regression by clustering patients with GPs and GPs with municipalities (160). Thus, we could account for hierarchy in data, which could help us explore whether the clustering of different types of chronic patients could be ascribed to a GP effect or a municipality effect. This knowledge could have helped us determine whether clusters of patients with particular diagnoses were due to specific characteristics of the GPs (unobservable GP qualities) or of the municipalities (for example, certain policies, funding or educational programs or certain age distribution in population or particular climate that made the population more prone to certain diseases).

5.3 Discussion of the results

5.3.1 Paper 1

5.3.1.1 Results

The first study's result indicate that the number of private radiology services reduced after 2008, while the number of public services either stayed the same or increased (in accordance with Hypothesis 1 and 3). Figure 3 illustrates the averages of referrals to private and public laboratories by self-employed GPs, with monthly fluctuations due to seasonal variations. In addition, high referral rates and competition levels observed prior to 2008 indicate a larger reduction in private examinations conducted after 2008 (in accordance with Hypothesis 2). At the aggregate level, we are not able to reject that the total reduction in the number of investigations is unchanged irrespective of the level of competition (Paper 1 (29)).

Examination rates were also analysed according to modality (see Table 4 in Paper 1). The results suggest that for all modalities other than ultrasound, the absolute value of the reduction in the rate of examinations performed by private providers increased with increases in competition between GPs; in addition, a competition-dependent reduction was observed in the total number of MRI services (supporting Hypothesis 4).

Regarding the other variables, shortage of the patients, and higher proportion of elderly and female population are correlated with an increase in the number of the services for both providers. Variables reflecting travel distance to providers exerted a significant effect on examination rates in the first study. Greater travel distances to providers are associated with a lower number of services (i.e. increased distance to the nearest public provider reduces the use of that public provider; this result was also found for private providers). Increased distance to a private provider increases the use of a public provider, and vice versa. A summary of all three papers' results is also presented in Table N3 to provide an easier overview.



Figure 3. Referrals to private and public laboratories by self-employed GPs: Monthly average examination rates per 1,000 listed patients (vertical axis) and months from January 2007 to December 2010 (horizontal axis). Source: Paper 1

Information box 4: Results of Paper 1

- The number of private radiology services reduced after 2008, while the number of public services either stayed the same or increased.
- A greater reduction in the number of examinations by private providers in municipalities with more competition for patients among GPs than in municipalities with less competition for patients among GPs.
- 3) The hypothesis was not supported for total (private and public together) numbers, indicating a greater flow of patients from private to public providers in the most competitive areas.
- The hypothesis was, however, supported for the MRI services, indicating a reduction in the highly competitive areas for private and public providers together.

Thus, regarding policy implications, the revenue system has the potential to be supplemented for gatekeeping for certain types of treatments and when the providers face hard budget constraints. In this case, the result is valid for MRI (29).

5.3.1.2.1 Rationing

The first results in Paper 1 indicated that the number of private and public radiology examinations after the reimbursement change decreased and increased, respectively. We find a greater reduction in the number of examinations by private providers in municipalities with more competition for patients among GPs than in municipalities with less competition for patients among GPs. The overall effect was inconclusive. The results of the competition indicators for overall rates suggest an increase in the number of examinations performed by public providers compensated for the reduction in the number of examinations performed by private providers, and competition did not exert a negative effect on total examination rates. In other words, public providers performed some of the examinations that had previously been performed by private providers in areas with high competition levels. The result underlines the importance of hard budget constraints for reimbursement changes to have effect. This implication could have been more clearly spelled out in the paper.

In regard to rationing, the results with the exemption on MRI suggest that rationing by private providers does not necessarily carry over to rationing in the system as a whole. It is not necessarily efficient for public providers to examine excess referrals previously made in high competition areas at private providers.

5.3.1.2.2 Modalities

When examination rates in the first study were analysed according to modality (see Table 4 in Paper 1), only MRIs had an overall reduction in areas with high levels of competition between GPs, indicating the rationing of patients with low expected benefit. MRI is considered a grey area of healthcare services, since MRIs have less clear indications, implying discretionary referrals; it is also called supply sensitive care (167). Regarding other services, we found that patients shifted from private to public providers, but the average number of services was constant.

This result could indicate that when outpatient care is set under the constraint, only services in the grey area of medicine with the least clear indications are rationed. We should also take into consideration that there is a higher threshold for both providers and patients to engage in MRI diagnostics than in the rest of the modalities due to several reasons. For example, MRI services are the most expensive radiology services in terms of the provider's resources (i.e., marginal costs are highest). They also are most resource taking for the patients: patients have to spend time in the device and not all the patients can tolerate it. Furthermore, wait lines for MRI are longest (168-170). Therefore, MRI's high marginal costs might be a reason that MRI services were cut for both providers after the change in the reimbursement.

Table 4 in Paper 1 also illustrates that X-rays and CAT scans shifted in the market from private to public providers, while ultrasound, the cheapest of all the modalities and an established diagnostic, did not have a significant effect on the competition variables. With regard to ultrasound, the reimbursement price may still be larger than the marginal costs for this service because ultrasound does not require as many resources from the provider as the rest of the modalities. Although the price reduction was an equal percentage for all modalities, it might have not reached the marginal costs for ultrasound.

5.3.1.2.3 Gatekeeping

Gatekeeping both saves on costs, preventing unnecessary public spending, and prevents the use of unnecessary specialist health services, thus preventing overtreatment.

60

Paper 1 studied gatekeeping only from the perspective of preventing unnecessary spending. To research overall overtreatment and undertreatment, we needed data on all the services, including private health insurance and out of pocket payments. Since the wait lines are long, there is a reason to believe that some of the patients from GP competitive areas might appeal to private health insurance or pay directly. Therefore, the overall conclusion is that although the change in the payment system for specialist providers might serve to ration and/or strengthen gatekeeping in the sense of the preventing unnecessary spending, it mostly affects the cases in which providers have hard budget constraints, services are highly supply sensitive ('grey zone') or marginal costs are high relative to price of the reimbursement. Although the study is inconclusive about overall undertreatment and overtreatment, the results suggests MRI overtreatment might have occurred before 2008.

5.3.1.2.4 Other findings

The results of robustness check using weighting indicated that the competition indicators might provide different information depending on specification. During changes in weighting, we observed that the results were still valid for #Open, but the significance changed for #Open/capita, which should be considered.

The variation in significance indicated some heteroscedasticity, as well as that we might pay more attention to the results of the regression, where we used '#open' as competition indicator because it gave the same results across all the regression with and without weighting.

The distance results are not surprising: increased distance to the provider contributed to greater reduction in number of services by the provider. This finding supports earlier research that nearby providers are used more frequently than distant providers (31, 144-

147). I discuss the question of distances in more detail in Paper 2's discussion of the results.

5.3.2 Paper 2

5.3.2.1 Results

The second study contributed results on the geographical variation in the provision of radiology services after the 2008 reimbursement change.

After the change, private services were reduced from 9.59 to 8 services per 1000 inhabitants per municipality, while public services increased from 24.35 to 26.16 nationally. However, the stratified statistics before and after the change are of the most interest: the services changed unevenly in different centralities and different health regions. For example, after the reimbursement change, total services stayed basically the same in South East and North RHA, dropped by one in West RHA, and increased by 2.5 in Central RHA.

The regression analysis indicates that patients received fewer services from the private providers in 2008 through 2010 than in 2007 (in accordance with Hypothesis 1). First, in accordance to Hypothesis 2A the number of services was reduced more for negative values of *Time_difference* (where private providers are closer than public), with a diminishing reduction until *Time_difference* equalled 10 hours. Second, patients received more services at public providers along with an increase in *Time_difference* in accordance with Hypothesis 2B indicating that there were more patients switching to public providers than patients rationed from public providers. Third, total number of services reduced for the patients in the municipalities where *Time_difference* less than 40 minutes and increased along with *Time_difference* for municipalities characterized by *Time_difference* greater than 40 minutes. Thus, hypothesis 2C was partially supported, with the only difference in the threshold: I was expecting that the reduction

would occur at negative *Time_difference*, while it lasted until *Time_difference* equalled 40 minutes.

Variation range in the number of services between different centralities (i.e., 'sizes of municipalities') was reduced by approximately 52% after 2008. The largest municipalities with the top centrality notably were not the greatest consumers of radiology services per capita as expected; instead, they are average. The greatest variation came from the small- and middle-sized municipalities. However, the variation range between different RHAs increased by approximately 20% after 2008. In the Central RHA, the total number of radiology services at both providers increased from 32 to 35, while in the West RHA, the number reduced from 31 to 30.

Information box 5. Results of Paper 2

- I. After the 2008 reimbursement change, patients received fewer services at private providers and more services at public providers (Hypothesis 1).
 - a. Private services reduced more for negative values of time difference, diminishing in reduction until *Time_difference* was around 10 hours (Hypothesis 2A),
 - b. Public services increased with an increase in *Time_difference*.
 This result indicates that the stream of the patients switching from private to public providers is greater than the reduction in public services due to reimbursement change (Hypothesis 2B),

- c. Total services reduced for *Time_difference* below 40 minutes and increased for *Time_difference* above 40 minutes (Hypothesis 2C).
- II. Variation between different centralities reduced by 51.8% and between different RHAs increased by 19.5%.

5.3.2.2 Discussion of the results

5.3.2.2.1 Effect of the distances

The reimbursement change had different implications for the number of supplied services to the municipalities depending on the relative distance to the provider. When the 2008 reimbursement changes took effect, private providers offered fewer services, and many patients chose to switch to the relatively more accessible public providers. Patients who had private and public providers in similar proximity switched more frequently from private to public providers. Approximately 30% of municipalities had either both providers within the municipality itself or at an equally close distance; the total population of these municipalities was around 52% (124 municipalities with average populations of 19,500 people, ranging from 345 to 573,000 people) of the sample and included the largest cities, such as for example Oslo, Tromsø, Bergen.

In the municipalities where there were only private providers nearby, the patients did not have the same opportunity to switch to public providers to compensate for the private providers' reduced offers. Thus, those municipalities had the greatest reductions in the number of total services. It concerned not a large percent of population, just 5% of municipalities with 6% of the Norwegian population. However, this total still consists of approximately 280,000 people (21 municipalities with average populations of 13,000 people, ranging from 2500 to 45,000).

In areas where patients had public providers relatively close, the least change in the number of services after 2008 occurred. In addition, patients switched from private to public providers even more in these areas. The total population of this group is around 42% of the total Norwegian population.

5.3.2.2.2 A need for further research about equity and equality

Some potential issues with regard to the effects of the reimbursement change should be investigated further. First, initially, private providers were supposed to help public providers share the patient load, but after the change, that role diminished, and public providers serviced patients previously served by the private sector. Thus, investigating the optimal workload for both types of providers would be useful because optimal distribution could contribute to the most efficient work for the public and private sectors, and thus shorter wait times.

Second, it would be useful to further examine what kind of patients were rationed in the areas with the greatest reductions in the total number of services and determine whether those were indeed the patients with the least expected benefit (with reference to Paper 1). In connection, we should discuss equity and equality for health care and social inequality. A study by Grasdal and Monstad (2011) distinguishes between inequality and inequity. 'While inequality simply refers to whether there is a correlation between a person's use of services and the person's ranking in the income distribution, inequity on the other hand takes into account individual need for treatment' (p 4 of 12 (171)). For example, smaller municipalities might have greater percent of elderly people compared to the municipalities with universities that attract younger people, and thus these municipalities would have different needs in radiology services.

Hence, it would be valuable to research whether the reform contributed to the general equity in healthcare access (i.e., whether it increased number of the services in areas with underused services and reduced the number of services in areas of overuse) (172). Therefore, to determine the type of patients who were rationed in the areas with the greatest reduction, we would need data on income distribution, individual need for treatment, age distribution, and number of services paid out of pocket or by insurance, mortality and emergency room visits at the municipality level, which we did not have.

Another limitation in our research was that we did not have patient level data on incomes and education levels of patients. Controlling for these variables would be useful since earlier research suggests that patients' use of health care is connected to their income (171), and research conducted by the Ministry of Health and Care Services indicates that social inequality correlates with differences in health status, life expectancy and place of residence (51).

5.3.2.2.3 Regional variation

In addition to population characteristics, the variation between RHAs could arise from the difference in the composition of municipalities in each RHA with respect to the municipality's size, centrality and distribution of the providers, as well as supply side variation. Regional variation might origin from the demand and supply sides (173). Regional variation might come from the patient demand due to possible heterogeneity of the patients toward care. If the people who demand more treatments tend to live in the same areas, regional variation could occur (174-176). From the supply side, regional variation could be due to supplier induced demand, or overprescription of the treatment
by physicians in order to gain more income. This variation could also arise as a response to organizational pressure to treat patients more intensively, especially for conditions with fewer guidelines (173, 177). Physicians clustering together, by working for the same clinic, for example, could create regional variation in beliefs (173). It is difficult to identify factors for regional differences (178), but some research indicates that half of differences are attributed to demand and half to supply (179). Cutler et al. (2019) suggests that although physician organizational factors matter, the most important factor is the physician believes about treatment, and demand side explains much less of total spending (173). In the supply side, financial considerations meant little, while pressure to accommodate either patient or the referring expectation to keep patients happy had a modest but significant relationship with physician beliefs about appropriate care. The greatest variation was due to physician's believes about efficacy of particular therapies (173). Thus to pinpoint the origins in the variation of radiology provision we would need to research further.

5.3.3 Paper 3

5.3.3.1 Results

The third study produced four types of results. First, Figure 4 (155) presents a scatter plot of GPs' proportions of DT2 patients, patient-list lengths (blue dots) and 95% interval curves (red lines), assuming patient allocation purely by chance. It indicates that patients were not allocated purely by chance (in accordance with expectation 1 in section 4.4). If they had been allocated randomly, 95% of the blue dots would have been positioned within the 95% confidence intervals (red lines). However, only 46.5% of the dots are positioned within these lines.



Figure 4. Scatter plot of GP proportion of DT2 patients and patient list length. Legend: Y-axis percent of DT2 patients, X-axis patient-list length. GP level, data for the first quarter of year 2009, N = 3,965, mean proportion of DT2 patients = 0.045, patient-list lengths of > 60 (Source: Figure 2 in Paper 3 (155)).

Second, regarding comorbidity, a large proportion of patients had at least two chronic conditions. For example, 13.5% of patients with DT2 and 28.7% of patients with schizophrenia had comorbid depression.

Third, the results provide information on correlations between variables. GPs' shares of patients with chronic diseases are positively correlated. This indicates that the majority of GPs have clusters of patients with chronic diseases (in accordance with expectation 1 in section 4.4). Shares of asthma, DT2, epilepsy and schizophrenia are negatively correlated with GP list length. Additionally, GP age and gender (male) were positively correlated with the seven chronic patient groups and negatively correlated with other patients.

Fourth, using logistic regressions, we found a pattern for patient GP disenrollment. Patients with chronic diseases who frequently used primary care (diagnosed with DT2, arthritis, asthma or depression) tended not to disenroll from GPs with large shares of patients with the same diagnosis. This tendency was not observed in patients who infrequently used primary care and frequently used secondary care (those diagnosed with DT1, schizophrenia or epilepsy). General practitioners with large shares of arthritis, asthma or depression patients were more popular in most patient groups (including patients without chronic diseases) (in accordance with expectation 2 in section 4.4). Patients without chronic diseases tended to be listed with GPs who had fewer patients who used primary care less frequently than the group of patients with chronic diseases. This patient group contained a higher number of younger, male patients who used primary care less frequently than the group of patients with specialist degrees in general medicine, longer patient lists or larger shares of arthritis patients. Older patients and patients who infrequently used primary care switch GPs less frequently than patients with chronic diseases.

All patients (with and without chronic diseases) tended to disenroll less from younger GPs or GPs with specialist degrees, longer patient lists or larger shares of arthritis

patients (valid for the majority of chronic diagnoses) relative to other GPs. Moreover, men and older patients tended to disenroll less from their GPs. However, patients who visited their GPs frequently tended to disenroll more often.

Information box 6. Results of Paper 3

- 1. Patients with chronic diagnoses were not allocated to GPs purely by chance (Expectation 1).
- Patients with chronic diseases who used primary care frequently (diagnosed with DT2, arthritis, asthma or depression) tended not to disenroll from GPs with large shares of patients with the same diagnosis.
- General practitioners with large shares of arthritis, asthma or depression patients were more popular in most patient groups (including patients without chronic diseases) (Expectation 2)
- 4. Patients without chronic diseases tended to be listed with GPs who had fewer patients with chronic diseases.

5.3.3.2 Discussion of the results

5.3.3.2.1 Quality indicators

The results indicated that all patient groups tend to disenroll less frequent from GPs with larger shares of arthritis, asthma and depression patients than from other GPs. The fact that patients with arthritis, asthma and depression disenroll less from GPs with a high share of these patients might be due to particular characteristics of GPs who seems

to specialize in those diseases. However, why other patient groups and patients without chronic diagnoses disenroll less from GPs with larger shares of arthritis, asthma and depression patients than from other GPs should be considered. In this and next subsection, I discuss the quality indicators and geographical clustering.

One reason patient preferences might align is unobservable qualities of GPs. Greater shares of patients with arthritis, asthma or depression may be indicators of more empathic GPs who are better communicators or diagnosticians, and thus more patients stay with them. We consider this possibility an indication of better quality GP care.

However, to validate the idea as a quality indicator, we would need to compare the results with existing quality indicators. There is extensive literature on quality indicators both in form of research articles and national guidelines.

Campbell et al. (2002) discusses research methods used in developing and applying quality indicators in primary care (180). A quality indicator is a measurable element of practice performance for which there is evidence or consensus that the element can be used to assess quality (180, 181). The researchers point out that when developing quality indicators, three issues are important. First is which stakeholder perspective the indicator is intended to reflect. Patient, caregivers, managers and professionals all have different perspectives that might not overlap (182-184). Health professionals often wish to follow guidelines, health outcomes and efficiency, while patients focus more on clinical performance, an understanding attitude and communication skills. In contrast, managers' pay more attention to data on efficiency, patient satisfaction, accessibility of care and outcomes. Second, the aspect of care, processes or outcomes, should be determined (185, 186). Outcomes such as mortality, morbidity, health status, health-related quality of life and patient satisfaction could be measured with indexes or with

surveys (187), whereas structural indicators inform about practice organization, for example, personnel, availability of appointments or parking. Process indicators describe medical care (e.g., diagnoses, treatments, referrals, prescriptions). A few examples may illustrate: first, for a diabetic patient whose feet are at risk, such an indicator would be whether he (she) was referred to a chiropodist (188); second, for patients generally an indicator of good preventative care would be low number of emergency visits or hospital stays (189); third, number of health checks performed by physician for patients to reduce risk of life style diseases (190). The third issue is information on structure, process or outcome where indicators are underpinned by evidence (i.e., how the indicators work and are applied in practice (180)).

Similarly, the Norwegian Directorate of Health suggests a few quality indicators that patients can use when choosing a care provider, for example, to what extent the provider follows treatment guidelines, whether the patient rights are met and whether the users are satisfied (6). Like Campbell's, the directorate's indicators are divided into three groups: (1) structural (e.g., frames and resources, expertise, available equipment, records), (2) process (activities in the patient course, such as diagnostics and treatment), (3) performance indicators (e.g., survival, health benefits, satisfaction) (4, 6). These indicators should be still interpreted with care. For example, mortality could indicate a poor treatment site or a site with best experts and thus the most severely ill patients are referred there (6).

An interesting question is what indirect indicators could be used to underpin high quality. A study from 2018 about patients with DT2 found that GPs' specialization in General Practice, being graduate from Western Europe, being female were correlated

with delivery of better care to DT2 patients by performing the recommended procedures more often and keeping better blood values (191).

Earlier research indicates that patients are more satisfied with physicians who have full lists and that patients switch less often from these physicians (124). Thus, Iversen and Lurås (2011) recommend making the number of switches per year available public information as a quality indicator (132). This would support our finding of the least disenrollment from the GPs with a greater share of arthritis, asthma and depression patients being a possible quality indicator.

Thus, a limitation of the study is our lack of data on other indicators for comparison and validation of our results. For example, checking the data to determine whether GPs performs regular controls for their chronically ill patients, or how many times patients ends up in hospital or emergency, would be useful to compare with our suggested indication of quality.

5.3.3.2.2 Cluster and disenrollment patterns

Patient clustering could also indicate that a particular geographical area has a greater share of these groups of patients and the rest of the patients do not have other GP choices. Here, I continue the discussion started in the Section 5.3.2.2.3, Regional variation. The reasons for particular patients' clustering could be GP related, patient related or area related. First, GP-related reasons could be unobservable attributes of GPs, GPs' communication abilities, personal characteristics, specialization and diagnostics patterns. For example, GPs who specialize in a particular disease, such as DT2, would also diagnose more patients with DT2, thus increasing the share of DT2 patients on their own lists. In addition, the GPs would also have patients with several types of chronic diseases because DT2 patients often have comorbidities. Such GPs would probably have certain practice patterns.

Patient-related reasons could be personal choices, patients' socio-economic characteristics or effect from informal conversations. For example, patients with DT2 might take part in an organization for diabetes patients and receive particular information. They also might talk with each other, discuss the care they receive and recommend particular GPs. Patients' networks and informal conversations exert a stronger effect on disenrollment than that exerted by publicly available information (192). Earlier findings indicate that obese patients sought 'obese-friendly' GPs (193), which implies that patients seek GPs with particular personal and practice profiles and consider potential GP-patient relationships (156, 194). The patients might also have their own non-observable preferences. The choice might also be affected by their age, gender and socio-economic status. Some research indicates that patients often prefer GPs similar to themselves (130).

Area-related patient distribution could occur if patients with particular diseases cluster in specific geographic areas (173). For instance, cold and humid climates might contribute to the development of arthritis, and thus such patients might cluster in the areas with these climates (195).

5.3.3.2.3 Other findings

Other important findings are that patients diagnosed with DT2, asthma, arthritis or depression use more primary care and switch GPs more frequently relative to other patients. In contrast to previous findings indicating that the duration and frequency of consultations did not affect GP switching (132), our results indicate patients who frequently visit their GPs tend to switch GPs frequently as well. Our findings suggest

that these patients switch more often until they find a patient-GP relationship that works well for them, and then they stay with their GP. The patients with chronic diseases value continuity of care (196, 197) (continuity of care is measured as proportion of consultations made by the usual GP (198)). Continuity of care is essential for positive outcomes from the treatments (132, 199, 200).

The length of patient lists is negatively associated with disenrollment and shares of chronic patients. This finding supports the results of earlier research that greater accessibility was not always equal to higher quality, and that longer patient lists were associated with superior disease management (135).

There is a gap in the knowledge on patient list combination and patterns on how patients change GPs, such as patients changing within the same group of GPs. For example, DT2 patients may change GPs among GPs with a large share of DT2.

6 Concluding remarks

All three papers provide evidence on how Norwegian health care responds to policy instruments, both financial incentives and organizational constraints. Paper 1 demonstrated that the revenue system for specialist care providers has the potential to supplement GP gatekeeping in restricting access to specialist care when providers face hard budget constraints (29). Paper 2 demonstrated that the reimbursement change contributed to (1) a greater reduction of the number of services for population that has only private providers nearby, (2) a reallocation of patients from private to public providers and (3) a reduction in the difference between different centralities of municipalities in their consumption pattern but an increase in the difference between different RHA regions.

The further national policy development for radiology providers was as follows. The demand for services pressed the providers by increasing waiting lists, which in turn, coerced the authorities to make adjustments. Thus, the reduction was followed by an increase in reimbursement. From 2014 to 2015, there was a 14% increase in the reimbursement to private providers and a 16% increase in the reimbursement to public providers (201).

Results of the paper 3 are restricted to the health care systems with regular GP schemes where patients are allowed to change GPs. Paper 3 demonstrated that indicators of GP quality might be identified through patient actions, such as disenrollment of patients with chronic diseases, who appreciate continuity of care. Large shares of arthritis, asthma or depression patients might be indicators of better quality of primary care since those GPs were more popular in most patient groups (including patients without chronic diseases). However, further research examining objective quality measures is required to determine whether disenrollment patterns could function as quality indicators (155).

From a policy prospective, the thesis provided evidence on the outcomes of financial incentives and organizational structures and how those might help reach the health policy goals of increasing health care services' quality, accessibility balanced with cost containment and effective resource allocation.

77

×
• •
q
g
(C
Q
d

.

Table N1. Overview of the three studies completed for the dissertation: Aims and Hypotheses*

Hypotheses	 With less fee for service (FFS), we expect a reduction in the volume of radiology services performed by private providers. If post-2008 rationing occurred according to declining benefits, we would expect a greater reduction in number of examinations by private providers in municipalities with more competition for patients among GPs. We hypothesize an increase in the number of examinations performed by public providers and a greater increase in the number of examinations by public providers in municipalities with nore competition for patients among GPs. We hypothesize an increase in the number of examinations performed by public providers and a greater increase in the number of examinations by public providers in municipalities with nore competition for patients among GPs. Because MRI examinations, to a greater extent than the other modalities, are located within the grey-area, we hypothesize a greater decline in the total number of MRI scans than for the other modalities and also a greater decline in municipalities with more other modalities and also a greater decline in municipalities with more of the other modalities and also a greater decline in municipalities with more other modalities and also a greater decline in municipalities with more competition for patients annog GPs (pp 12-14 in Paper 1 (29))
Aim	Paper 1 asks whether the revenue system for secondary care providers may supplement the gatekeeping mechanism in regulating utilization of secondary care.
Type/Field of Journal / Target audience	Health economics / Economists and administrato rs
Name	Supplementing gatekeeping with a revenue scheme for secondary care providers. Iversen, Mokienko, Int J Health Econ Manag, 2016. DOI 10.1007/s10754- 016-9188-2.
Papers	

inistrato patterns could indicate otherwise	unobserved quality				
longitudinal study admin of Norwegian rs	claims data. Mokienko,	Wangen. BMC	Family Practice, 2016.	https://doi.org/10.1	<u>186/s12875-016-</u> 0571-3

Γ

*Source: Papers 1, 2, 3 (29, 155)

tbles Methods	1. Theory model with direct and indirect rationing nations per month attents at the level 1. Theory model with direct and indirect rationing nations per month attest total, private nations per month attents at the level 2. Empirical approach: attoning per month GP. 3. Model: Fixed effect attents at the level 3. Model: Fixed effect GP. a. Model: Fixed effect attents at the level 5. Study effects of competition indicators, • Study effects of competition indicators, • Clustered on the level of hospital catchment areas (21 clusters) due to contracts with radiology, nber of GPs in a who accept the municipality, use weighted regression with the municipality populations across • To correct for variation in the size of the municipality, use weighted regression with the municipality populations across Ss. Aggregated numbers vs. folder people • Competition indicators: (1) no indicators: (1) no indicator, (2) #Open and (3) #Open/capita, 10 a dGP, • For number of private, public an total services
Main varia	The dependent variables: • The examination and public examination and public examines of the individual of the individual Independent variables: 1. Competition indicator #Open/capita. a. #Open = num municipality additional pa b. #Open/capita variations in municipalitie 2. GP practice: • The proportion of fem municipalitie 0. Adomny variables: 1. Competition indicator additional pa b. #Open/capita ab b. #Open/capita ab b. #Open/capita b. #Op
Data	 1. 176,709 monthly observations of 4486 self-employed GPs, covering the entire population of self-employed GPs who contract with municipalities in Norway. 2. Main data: Claims data to the NIS (Norwegian Insurance Scheme) from public and private radiology laboratories from 2007 to 2010 from the Norwegian Directorate of Health Number of examinations according to type of imaging (the modalities X-ray, ultrasound, magnetic resonance imaging (MRI) and computerized axial tomography (CAT scan)) and fees for patients covered by NIS; Each claim is assigned to the patient's regular GP through a patient ID number of claims during a month at the level of the GP; Merged with GP characteristics and patient list data;
Name	Supplementing gatekeeping with a revenue scheme for secondary care providers. Iversen, Mokienko, Int J Health Econ Manag, 2016. DOI 10.1007/s10754-016- 9188-2

Table N2. Overview of the three studies completed for the dissertation: Data and Methods st

81

 c. To check robustness Cluster on municipality level due to GP contracts. The significance of the result does not change. Change the weights to patient-list size or drop weights. The significance of the result depends on the competition indicator. 	 Empirical approach: 1. Stratified summary statistics (according to centralities and RHAs) for before and after the reimbursement change of 2008 2. Regression analysis: Model: Fixed effect model To study effect of <i>Time_difference</i> For private, public and both providers Quadratic function V Different polynomial functions were checked but quadratic was chosen. 3. Measure of the change in numbers of radiology services before and after 2008 using values obtained from stratified summary statistics for (a) RHAs and (b) centralities
 patients on his (her) list of above 100 patients compared to the desired list size. 3. Travel time to the nearest private provider and travel time to the nearest public provider. See Table 1 in Paper 1 for measurements and details (29) 	 The dependent variable: A number of services performed at private (<i>Priv_Serv</i>), public (<i>Pub_Serv</i>) or both providers together (<i>Total_serv</i>) per month per 1,000 inhabitants in the municipality; If a patient from Municipality A goes to Municipality B for a radiology examination, that service is classified as a service delivered to Municipality A. The independent variable: <i>Time_difference = Privtime - Pubtime =</i> the difference in travel time between the nearest public provider in hours The travel times according to driving time by car between a patient's residential municipality of the patient's GP) and the municipality of the public radiology
provider by Info Map.	 422 municipalities in 48 periods (monthly observations during 2007 to 2010), thus 19,867 observations. Main data: Claim data to NIS from private and public radiology providers in Norway, obtained from the Norwegian Directorate of Health (data used for paper 1) Aggregated number of radiology services (CAT scans, MRIs, X-rays, and ultrasounds) to municipality level, Merged with travel times between GP's municipality and municipality of provider provided by InfoMap Norway
	unding travel livery of public srvices in sejster- tudinal rwegian st ss and llocation
	Effect of a 1 change and times on de private and radiology sc Norway: Ré based longi study of No claims data, Mokienko, to BMC Co Effectivene Resource A
	0

	 GP level data. Graphs to reveal whether the distribution of chronic patients seemed disproportionate across GPs. See Fig. 2 in Paper 3: proportion of patients with DT2 Spearman's correlation coefficients for the various GP-related variables: the
 provider (<i>Pubtime</i>) or the private radiology provider (<i>Privtime</i>) in hours If the patient has the radiology provider in his (her) own municipality, then the travel time will be set to zero by definition in the data set. Centrality of the municipality (<i>Centrality 1-7</i>, dummy variables) is a variable from Statistics Norway and classifies every municipality in Norway according to one of seven categories, where 7 is most central (e.g., Oslo) and 1 is least central municipality belongs to South-East (1), West (2), Central (3), or North (4) Regional Health Authorities. 	 Dependent variable: 'SwitchOut': measured whether a patient disenrolled from a GP from one semiannual period to the subsequent period. Independent variables (see Table 1 in Paper 3): GPs' age, sex, specialization and list length,
	 Data: 1. Data from two national registers in Norway, administrated by the Norwegian Directorate of Health, from 2009 to 2011, six semiannual intervals, merged using the GPs' IDs. • GP data: national register of regular GPs, which covers the entire GP population, • Patient data: claims data obtained
	3 Disenrollment from general practitioners among chronic patients: a register- based longitudinal study of Norwegian claims data. Mokienko, Wangen. <i>BMC Family</i> <i>Practice</i> , 2016.

 nber shares of patients with different diagnoses, the GP's age and sex. 2. Patient level data. 3. Patient level data. • Comorbidity: percent of patients with a chronic disease that have another chronic disease. • Logistic regressions to model patients' disenrollment from their GP: a. To study the effect of GP patient shares with certain diagnosis, b. Subsamples for each patient diagnosis, b. Subsamples for each patient diagnosis, c. Comparison group from Sample 1, c. Comparison group from Sample 2, d. Five effective observations for each patient. 	
 Patients' sex, birth year and num of visits, Number of patient's comorbidit (range between 0 and 6), Shares of the patients with certa diagnosis on the GP list 	
 from the KUHR registry (Kontroll og Utbetaling av HelseRefusjon), which covers the entire Norwegian patient population. The individual level of data included patient characteristics, their consumption of primary care and the GP with which they were encolled. 2. Two samples of patients were selected among patients who visited a GP at least once from 2009 to 2011: a. Sample 1: patients registered with one or more of the following seven diagnoses at least once during the period 2006 to 2011: DT1, DT2, asthma, arthritis, schizophrenia, depression and epilepsy. b. Comparison group, Sample 2: entire birth year cohorts from 1940 and 1970, except patients already included in sample 1. 3. After certain exclusions to clean the data of irrelevant information, we had 313,659 patients in Sample 2 (343,871 in total) (Fig. 1 in Paper 3) 	
https://doi.org/10.118 6/s12875-016-0571-3	

^{*}Source: Papers 1, 2, 3 (29, 155)

Table N3. Overview of the three studies completed for the dissertation: Results and policy implications*

Policy implications	The revenue system for specialist care providers has the potential for supplementing GP gatekeeping in restricting access to specialist care when providers face hard budget constraints.
Results	 Private providers conducted fewer examinations in 2008 to 2010 compared with 2007, and public providers conducted either the same volume or more (Hypotheses 1 and 3) A greater reduction occurred in the number of examinations by private providers in municipalities with more competition for patients among GPs than in municipalities with the competition for patients among GPs. GPs who practice in municipalities with competition 1 patients also initiated a greater rate of examinations at private providers before 2008. => These results imply a greater absolute value of reductions at private providers for GPs who initially exhibited the greatest referral rate (Hypothesis 2). Some of the reduction in private labs is compensated for by an increase in the number of examinations at public labs. At the aggregate level, we are not able the reject that the total reduction in the number of investigations is unchanged irrespective of the level of competition. From a disaggregated analysis, we find for MRIs that the absolute value of the reduction in the sum of private and public examinations increases with the level of competition => the rationing that occurs after 2008, in particular, is a disadvantage for patients in the grey area (which MRI is considered to be) with a low expected benefit (Hypothesis 4) Other results: Distance to the nearest public provider, a shortage of patients, a high proportion of old people and the proportion of females have a positive effect on the number of services at private providers, whereas distance to the nearest private provider has a negative effect. Distances to the nearest private provider and a shortage of patients have a positive effect on the number of services at public providers, whereas the distance to the nearest public provider has a negative effect.
Name	Supplementin gatekeeping with a revenut scheme for secondary cart providers. Iversen, Mokienko, Int J Health Econ Manag, 2016. DOI 10.1007/s107 4-016-9188-2.
Papers	_

7	Effect of a funding change and travel times on delivery of private and public radiology services in Norway: Register-based longitudinal study of Norwegian claims data, Mokienko, Submitted to BMC Cost Effectiveness and Resource Allocation	· · · · · · · · · · · · · · · · · · ·	 After the 2008 reimbursement change, patients received fewer services at private providers and more services at public providers (Hypothesis 1). This result indicates that the stream of the patients that switched from private to public providers is greater than the reduction in the public services due to reimbursement change. Reduction in private services after 2008, but the reduction diminished along with <i>Time_difference</i> (Hypothesis 2A) Public services increased along with <i>Time_difference</i> (Hypothesis 2B) Total number of services decreased until <i>Time_difference</i> ≤ 40 min and increased for <i>Time_difference</i> ≥ 40 min (Hypothesis 2C) Variation reduced between different centralities by 51.8% and increased between different RHA by 19.5%. The largest municipalities (highest centrality) are notably not the greatest consumers of radiology services per capita; they are just average. The variation comes from the small- and middle-sized municipalities. 	The geographical distribution of the providers and how differently the providers react to reimbursement changes, affect the implications of the reimbursement change for publicly reimbursed providers in 2008. Reimbursement change contributed to 1) a greater reduction of the number of services for population that has only private providers nearby, 2) a reallocation of the patients from private to public providers, 3) a reduction in the difference between different centralities of municipalities in their consumption pattern but increase in the difference between different RHA regions.
ε	Disenrollment from general practitioners among chronic patients: a register-based longitudinal study of Norwegian claims data.	э 5 Т	 The proportion of patients with DT2 varied substantially among GPs. If these patients had been allocated purely by chance, about 95% of the proportions would lie between the curves of confidence interval (Figure 2 in Paper 3), but this was not the case (Expectation 1). a. Only 46.5% of the proportions were positioned within the red curves. b. For the other diagnosis groups, the corresponding patient shares also seemed disproportionally distributed. Patients usually have several comorbidities (Table 3 in Paper 3). a. The correlation coefficient of 'Asthm_share' and 'DT1_share' was 0.648, indicating 	 Patients with chronic diseases are not allocated to GPs only by chance; Chronic patients who use primary care intensively disenroll less often from GPs who have a high share of patients with the same diagnosis;

Wangen. BMC		that GPs with a high]	proportion of patients with asthma also tended to have	a high	3. Most patient groups tend to remain with	
r amuy Practice. 2016.		proportion of patients	s with DT1.		3Ps with a greater share of arthritis, asthma	
https://doi.org/	þ.	The proportion of pat	ients with chronic diseases were all positively correlat	ed and	and depression patients, which may indicate	
<u>10.1186/s1287</u>		negatively correlated	with the proportion of other patients ('Other_Share').		better quality care for these and other patient	
<u>5-1/0-010-0</u>	ં	'Other_Share' was ne	sgatively correlated with 'GP_Age' and 'GP_Sex', ind	icating	groups.	
		that older GPs and ma	ale GPs tended to have fewer patients without the seve	n chronic		
		diseases.				
	4. Estimat	ions:				
	a.	Main effects.				
		i. The 'own sh	are effect': all patient groups tended to remain with Gl	Ps who		
		had a high sl	hare of patients with the same diagnosis.			
		ii. The 'cross sl	hare effect': for instance, a high share of DT1 patients			
		increased the	e switch-out for patients with arthritis (meaning, patier	its with		
		arthritis wer	e more likely to switch if their GPs had more patients	with		
		DT1).				
		iii. Most patient	t groups tended to remain with GPs with a greater shar	e of		
		arthritis, asth	nma and depression patients (Expectation 2)			
	þ.	Other results:				
		i. Patients tend	led to switch less often from GPs who had long patient	t lists or		
		who were sp	occialists in general medicine.			
		ii. Patients borr	n more recently or who had more comorbidities tended	to		
		switch GPs 1	more often.			
		iii. The 1% of p	atients who most frequently used primary care tended	to switch		
		less often th	an patients who had fewer visits. However, among the			
		remaining 90	9% of patients, those with a higher number of primary	care		
		visits tended	l to switch more often.			

*Source: Papers 1, 2, 3 (29, 155)

87

References

1. Meld. St. 44 (1995-1996) Ventetidsgarantien - kriterier og finansiering. Sosialog helsedirektoratet. [White Paper no. 44 (1995-1996) The Waiting-time Guarantee -Criteria and Financing. Ministry of Health and Care Services (ed.)]. 1995.

2. The Norwegian Directorate for Health and Social Affairs. Activity-Based Funding of Health Services in Norway. An Assessment and Suggested Measures.; 2007.

3. Berg O. Fra politikk til økonomikk. Den norske helsepolitikks utvikling det siste sekel.: Den norske lægeforening; 2006.

4. Og bedre skal det bli! Nasjonal strategi for kvalitetsforbedring i Sosial- og helsetjenesten. Sosial- og helsedirektoratet. [And better it will be! National strategy for quality improvement in the social and health service. Ministry of Health and Care Services]. 2005.

5. Meld. St. 26 (2014–2015) Melding til Stortinget. Fremtidens primærhelsetjeneste – nærhet og helhet. Helse- og omsorgsdepartementet [White Paper 26 (2014-2015) Future primary health service - proximity and wholeness. Ministry of Health and Care Services (ed.)]. 2015.; 2015.

6. Om kvalitet og kvalitetsindikatorer. Direktoratet for E-Helse. [About quality and quality indicators. Directorate of E-Health] [Available from: https://helsenorge.no/Kvalitetsindikatorer/om-kvalitet-og-kvalitetsindikatorer.

7. Ulserød T. Hva er en stortingsmelding? Civita2018 [updated 14.04.2018. Available from: <u>https://www.civita.no/politisk-ordbok/hva-er-en-stortingsmelding</u>.

8. Meld. St. 11 (2014 - 2015) Melding til Stortinget. Kvalitet og pasientsikkerhet. Helse- og omsorgsdepartementet. [White Paper 11 (2014 - 2015). Quality and patient safety. Ministry of Health and Care Services]. 2013.

9. Meld. St. 12 (2015 - 2016) Melding til Stortinget. Kvalitet og pasientsikkerhet. Helse- og omsorgsdepartementet. [White Paper 12 (2015 - 2016). Quality and patient safety. Ministry of Health and Care Services]. 2014.

10. Meld. St. 13 (2016-2017) Melding til Stortinget. Kvalitet og pasientsikkerhet. Helse- og omsorgsdepartementet. [White Paper 13 (2016 – 2017). Quality and patient safety. Ministry of Health and Care Services]. 2015.

11. Meld. St. 6 (2017 - 2018) Melding til Stortinget. Kvalitet og pasientsikkerhet. Helse- og omsorgsdepartementet. [White Paper 6 (2017 - 2018). Quality and patient safety. Ministry of Health and Care Services]. 2016.

12. St.meld. nr. 23 (1996-97) Trygghet og ansvarlighet. Om legetjenesten i kommunene og fastlegeordningen. Sosial- og helsedepartementet. [White Paper 23 (1996 – 1997). Security and accountability. About the medical service in the municipalities and the regular GP scheme. Ministry of Health and Care Services]. 1997. 13. The Royal Proposition on State Budget for Budget Year 2008. St.prp. nr. 1

(2007–2008). In Norwegian.: Hearing before the The Royal Finance Ministry(2008).

14. Veggeland N. Fastlegeordningen og reformer. Tidsskrift for velferdsforskning. 2018;21(01):59-68.

15. Frikort for helsetjenester. Direktoratet for E-Helse 2019 [Available from: <u>https://helsenorge.no/betaling-for-helsetjenester/frikort-for-helsetjenester#Frikort-for-egenandelstak-1</u>.

16. International Health Care Systems Profile. Norway. Anne Karin Lindahl. In edited by Elias Mossialos and Ana Djordjevic London School of Economics and Political Science Robin Osborn and Dana Sarnak The Commonwealth Fund. 2017.

17. Helsenorge.no [Available from: <u>https://helsenorge.no/</u>.

18. Direktoratet for e-helse [Available from: <u>https://ehelse.no/</u>.

19. Subordinate Institutions. Ministry of Health and Care Services. [Available from: <u>https://www.regjeringen.no/en/dep/hod/organisation-and-management-of-the-ministry-of-health-and-care-services/etater-og-virksomheter-under-helse--og-</u>

omsorgsdepartementet/Subordinate-institutions/id115200/.

20. The Norwegian Medicines Agency. Statens legemiddelverk. [Available from: <u>https://legemiddelverket.no/english</u>.

21. Authorisation and License for Health Personnel [Available from: <u>https://helsedirektoratet.no/english/authorisation-and-license-for-health-personnel</u>.

22. Norwegian Institute of Public Health [25.02.2019]. Available from: <u>https://www.fhi.no/en/</u>.

23. Norwegian Board of Health Supervision [25.02.2019]. Available from: <u>https://www.helsetilsynet.no/norwegian-board-of-health-supervision/</u>.

24. Norwegian Radiation and Nuclear Safety Authority. [25.02.2019]. Available from: <u>https://www.regjeringen.no/en/dep/hod/organisation-and-management-of-the-</u> ministry-of-health-and-care-services/etater-og-virksomheter-under-helse--ogomsorgsdepartementet/Subordinate-institutions/norwegian-radiation-and-nuclear-

safety-authority/id279760/.

25. The Norwegian System of Compensation to Patients.

26. The Norwegian Biotechnology Advisory Board [Available from: https://www.regjeringen.no/en/dep/hod/organisation-and-management-of-theministry-of-health-and-care-services/etater-og-virksomheter-under-helse--og-

omsorgsdepartementet/Subordinate-institutions/The-Norwegian-Biotechnology-Advisory-Board/id450770/.

27. About the Health and Social Services Ombudsman.

28. Norwegian Health Economics Administration. About Helfo. [Available from: <u>https://helfo.no/english/about-helfo</u>.

29. Iversen T, Mokienko A. Supplementing gatekeeping with a revenue scheme for secondary care providers. International Journal of Health Economics and Management. 2016:1-21.

30. Changing your regular doctor Directoratet for E-Helse [Available from: <u>https://helsenorge.no/foreigners-in-norway/general-practitioner?redirect=false</u>.

31. Iversen T, Ma CT. Market conditions and general practitioners' referrals. International journal of health care finance and economics. 2011;11(4):245-65.

32. Ringard Å, Sagan, A., Saunes, I.S., Lindahl, A.K. Norway Health system review. Health systems in Transition. 2013. Contract No.: No 8 2013.

33. Fritt behandlingsvalg. Helsedirektoratet. [Available from: https://helsedirektoratet.no/fritt-behandlingsvalg.

34. Fritt behandlingsvalg. Direktoratet for E-Helse. [Available from: https://helsenorge.no/rettigheter/fritt-behandlingsvalg.

35. Egenandel hos lege. Direktoratet for E-Helse. [Available from: https://helsenorge.no/betaling-for-helsetjenester/betaling-hos-lege.

36. Exemption card for public health services Direktoratet for E-Helse2019 [Available from: <u>https://helsenorge.no/foreigners-in-norway/exemption-card-for-public-health-services?redirect=false</u>.

37. Moe L. Norske kunder kjøper helseforsikring for 300 millioner mer enn I 2013. Dagens medisin. 24.02.2016.

38. Norwegian Health Economics Administration. Copayments for health care services. In Norwegian. 2016 [Available from: <u>https://helsenorge.no/egenandeler-for-helsetjenester</u>.

39. Biørn E, Hagen T, Iversen T, Magnussen J. How different are hospitals' responses to a financial reform? The impact on efficiency of activitybased financing. 2010;13:1-16.

40. Northern Norway Regional Health Authority. Contract of Purchize of Policlinic Radiology Services between Curato Røntgen AS and North RHA. In Norwegian [Helse Nord Rammeavtale Kjøp av polikliniske radiologiske tjenester mellom Curato Røntgen AS og Helse Nord RHF] 2013 [

41. South east Norway Regional Health Authority. Contract of Purchize of Policlinic Radiology Services with Unilabs. In Norwegian. 2013 [

42. South East Norway Regional Health Authority. Contract of Purchase of Outpatient Radiology Services. In Norwegian. 2014 [

43. Norwegian Health Economics Administration. Analysis of Total Spending on Outpatient Laboratory and Radiology Services. Analysis Report 4 2010. In Norwegian: Analyse av samlede utgifter til polikliniske laboratorie- og røntgentjenester. Analyserapport r 4 2010. Versjon 1.0. 1 juni 2010.

44. St.meld. nr. 5 (2003-2004) Inntektssystem for spesialisthelsetjenesten. Helseog omsorgsdepartementet. [White Paper 5 (2003-2004) Income system for the specialist health service. Ministry of Health and Care Services]. 2004.

45. The Royal Proposition on State Budget for Budget Year 2006. St.prp. No.1 (2005-2006), Ministry of Health and Care Services.

46. Kornai J. Resource-Constrained versus Demand-Constrained Systems. Econometrica. 1979;47(4):801-19.

47. Kornai J. "Hard" and "soft" budget constraint. Acta Oeconomica. 1980;25(3/4):231-45.

48. Almén A, Friberg EG, Widmark A, Olerud HM. Radiology in Norway anno 2008. Trends in examination frequency and collective effective dose to the population. StrålevernRapport 2010:12. Østerås: Norwegian Radiation Protection Authority, 2010.Language: Norwegian.

49. Meld. St. 34 (2015–2016) Verdier i pasientens helsetjeneste — Melding om prioritering. Helse- og omsorgsdepartementet. [White Paper 34 (2015-2016). Values in the patient's health service - about prioritization. Ministry of Health and Care Services]. 2016.

50. Meld. St. 11 (2015–2016) Nasjonal helse- og sykehusplan (2016–2019). Helse- og omsorgsdepartementet. [White Paper 11 (2015–2016) National Health and Hospital Plan (2016–2019). Ministry of Health and Care Services]. 2016.

51. National Strategy to Reduce Social Inequalities in Health, Report No. 20. Ministry of Health and Care Services. 2007.

52. Bemelmans-Videc ML, Rist, R.C., Vedung, E. Policy Instruments: Typologies and Theories in Carrots, Sticks and Sermons. In: McCormick J, editor. Carrots, Sticks and Sermons Policy Instruments and Their Evaluation. 1st ed. New York: Routledge; 1998.

53. Kazungu JS, Barasa EW, Obadha M, Chuma J. What characteristics of provider payment mechanisms influence health care providers' behaviour? A literature review. The International Journal of Health Planning and Management. 2018;33(4):e892-e905.

54. Conrad DA, Perry L. Quality-Based Financial Incentives in Health Care: Can We Improve Quality by Paying for It? Annual Review of Public Health. 2009;30(1):357-71.

55. Diamond George A, Kaul S. Evidence-Based Financial Incentives for Healthcare Reform. Circulation: Cardiovascular Quality and Outcomes. 2009;2(2):134-40.

56. Lavergne MR. Financial incentives for physicians to improve health care. CMAJ : Canadian Medical Association journal = journal de l'Association medicale canadienne. 2017;189(49):E1505-E6.

57. Scott A, Liu M, Yong J. Financial Incentives to Encourage Value-Based Health Care. Medical Care Research and Review. 2016;75(1):3-32.

58. Hunt A. Non-Financial Provider Incentives: Looking Beyond Provider Payment Reform. Altarum Health Care Value Hub. 2018(RESEARCH BRIEF NO. 24).

59. Abduljawad A, Al-Assaf AF. Incentives for better performance in health care. Sultan Qaboos University medical journal. 2011;11(2):201-6.

60. Gorman A. How are healthcare organizations making use of financial incentives to motivate patients? Kaiser Health News. 2017 05.12.2017.

61. Gosden T, Pedersen L, Torgerson D. How should we pay doctors? A systematic review of salary payments and their effect on doctor behaviour. QJM : monthly journal of the Association of Physicians. 1999;92(1):47-55.

62. Rice T. The Physician as the Patient's Agent. In: Jones AM, editor. The Elgar Companion to Health Economics2006.

63. Robinson JC. Theory and practice in the design of physician payment incentives. The Milbank quarterly. 2001;79(2):149-77, iii.

64. Scott A. Chapter 22 Economics of general practice. Handbook of Health Economics. Volume 1, Part B: Elsevier; 2000. p. 1175-200.

65. Jegers M, Kesteloot K Fau - De Graeve D, De Graeve D Fau - Gilles W, Gilles W. A typology for provider payment systems in health care. (0168-8510 (Print)).

66. Cashin C. Assessing health provider payment systems: A practical guide for countries working toward universal health coverage. 2015.

67. Cashin C, Langenbrunner J, O'Dougherty S. Designing and Implementing Health Care Provider Payment Systems: How-to Manual2009.

68. Rosenthal M. Pay for performance: A decision guide for purchasers. Agency Healthcare Res Qual (AHRQ). 2006(6):1-47.

69. Chandra A, Cutler D, Song Z. Chapter Six - Who Ordered That? The Economics of Treatment Choices in Medical Care. In: Mark V. Pauly TGM, Pedro PB, editors. Handbook of Health Economics. Volume 2: Elsevier; 2011. p. 397-432.

70. Paying providers, pp 331-337 from P. Zweifel et al., Health Economics, 2nd ed. Springer-Verlag Berlin Heidelberg. 2009.

71. Allard M, Jelovac I, Léger PT. Treatment and referral decisions under different physician payment mechanisms. Journal of health economics. 2011;30(5):880-93.

72. McGuire TG. Chapter 9 - Physician Agency. In: Anthony JC, Joseph PN, editors. Handbook of Health Economics. Volume 1, Part A: Elsevier; 2000. p. 461-536.

73. van Dijk CE, van den Berg B, Verheij RA, Spreeuwenberg P, Groenewegen PP, de Bakker DH. Moral hazard and supplier-induced demand: empirical evidence in general practice. Health economics. 2013;22(3):340-52.

74. Douven R, Mocking R, Mosca I. The effect of physician remuneration on regional variation in hospital treatments. International Journal of Health Economics and Management. 2015;15(2):215-40.

75. Stearns SC, Wolfe BL, Kindig DA. Physician responses to fee-for-service and capitation payment. Inquiry : a journal of medical care organization, provision and financing. 1992;29(4):416-25.

76. Hickson GB, Altemeier WA, Perrin JM. Physician reimbursement by salary or fee-for-service: effect on physician practice behavior in a randomized prospective study. Pediatrics. 1987;80(3):344-50.

77. Epstein AM, Begg CB, McNeil BJ. The use of ambulatory testing in prepaid and fee-for-service group practices. Relation to perceived profitability. The New England journal of medicine. 1986;314(17):1089-94.

78. Krasnik A, Groenewegen PP, Pedersen PA, von Scholten P, Mooney G, Gottschau A, et al. Changing remuneration systems: effects on activity in general practice. BMJ (Clinical research ed). 1990;300(6741):1698-701.

79. Holte JH, Sivey P, Abelsen B, Olsen JA. Modelling Nonlinearities and Reference Dependence in General Practitioners' Income Preferences. Health economics. 2016;25(8):1020-38.

80. Pedersen L, Jarbol D. [Preferences for user fees in general practice and the establishment of general practitioners in rural areas.]. Ugeskrift for laeger. 2012;174:2940-3.

81. Pedersen L, Gyrd-Hansen D. Preference for practice: A Danish study on young doctors' choice of general practice using a discrete choice experiment. Eur J Health Econ. 2014;15: 611.

82. Holte JH, Kjaer T, Abelsen B, Olsen JA. The impact of pecuniary and nonpecuniary incentives for attracting young doctors to rural general practice. Social Science & Medicine. 2015;128:1-9.

Abelsen B, Olsen JA. Young doctors' preferences for payment systems: the influence of gender and personality traits. Human Resources for Health. 2015;13(1):69.
Abelsen B, Olsen JA. Does an activity based remuneration system attract young doctors to general practice? BMC Health Services Research. 2012;12(1):68.

85. Mitchell AP, Rotter JS, Patel E, Richardson D, Wheeler SB, Basch E, et al. Association Between Reimbursement Incentives and Physician Practice in Oncology: A Systematic ReviewAssociation Between Reimbursement Incentives and Physician Practice in OncologyAssociation Between Reimbursement Incentives and Physician Practice in Oncology. 2019.

86. Graves AJ, O'Neil B, Barocas DA, Penson DF, Resnick MJ, Chang SS. Doing More for More: Unintended Consequences of Financial Incentives for Oncology Specialty Care. JNCI: Journal of the National Cancer Institute. 2015;108(2).

87. Hadley J, Mandelblatt JS, Mitchell JM, Weeks JC, Guadagnoli E, Hwang Y-T, et al. Medicare breast surgery fees and treatment received by older women with localized breast cancer. Health services research. 2003;38(2):553-73.

88. Epstein AJ, Johnson SJ. Physician response to financial incentives when choosing drugs to treat breast cancer. International journal of health care finance and economics. 2012;12(4):285-302.

89. Jacobson M, Earle CC, Price M, Newhouse JP. How Medicare's Payment Cuts For Cancer Chemotherapy Drugs Changed Patterns Of Treatment. Health Affairs. 2010;29(7):1391-9.

90. Conti RM, Rosenthal MB, Polite BN, Bach PB, Shih Y-CT. Infused Chemotherapy Use in the Elderly After Patent Expiration. Journal of Oncology Practice. 2012;8(3S):e18s-e23s.

91. Douven R, Remmerswaal M, Mosca I. Unintended effects of reimbursement schedules in mental health care. Journal of health economics. 2015;42:139-50.

92. Sood N, Buntin, M. B., Escarce, J. J. Does how much and how you pay matter? Evidence from the inpatient rehabilitation care prospective payment system. Journal of health economics. 2008;27(4):1046-59.

93. Biørn E, Hagen TP, Iversen T, Magnussen J. The Effect of Activity-Based Financing on Hospital Efficiency: A Panel Data Analysis of DEA Efficiency Scores 1992–2000. Health care management science. 2003;6(4):271-83.

94. Linna M, Häkkinen U, Peltola M, Magnussen J, Anthun KS, Kittelsen S, et al. Measuring cost efficiency in the Nordic hospitals--a cross-sectional comparison of public hospitals in 2002. Health care management science. 2010;13(4):346-57.

95. Hagen T, Veenstra, M, Stavem, K. Efficiency and Patient Satisfaction in Norwegian Hospitals. Oslo: Health Economics Research Programme (HERO). 2006;2006 Working paper(2).

96. Eldenburg L, Kallapur S. Changes in hospital service mix and cost allocations in response to changes in Medicare reimbursement schemes. Journal of Accounting and Economics. 1997;23(1):31-51.

97. Bellows NM, Halpin HA. Impact of Medicaid reimbursement on mental health quality indicators. Health Serv Res. 2008;43(2):582-97.

98. Januleviciute J, Askildsen, J. E., Kaarboe, O., Siciliani, L., Sutton, M. How do Hospitals Respond to Price Changes? Evidence from Norway. Health economics. 2016;25(5):620-36.

99. Dafny LS. How Do Hospitals Respond to Price Changes? American Economic Review. 2005;95(5):1525-47.

100. Pita Barros P, Siciliani L. Chapter Fifteen - Public and Private Sector Interface. In: Mark V. Pauly TGM, Pedro PB, editors. Handbook of Health Economics. Volume 2: Elsevier; 2011. p. 927-1001.

101. Propper C. Competition in health care: lessons from the English experience. Health Economics, Policy and Law. 2018;13(3-4):492-508.

102. Gaynor M, Town RJ. Competition in Health Care Markets. Handbook of Health Economics2011. p. 499-637.

103. Wright DB, Ricketts Iii TC. The road to efficiency? Re-examining the impact of the primary care physician workforce on health care utilization rates. Social science & medicine. 2010;70(12):2006-10.

104. Fortney JC, Steffick DE, Burgess Jr JF, Maciejewski ML, Petersen LA. Are primary care services a substitute or complement for specialty and inpatient services? Health Services Research. 2005;40(5 I):1422-42.

105. Islam MK, Kjerstad E. The Ambiguous Effect of GP Competition: The Case of Hospital Admissions. Health economics. 2017;26(12):1483-504.

106. Godager G, Iversen, T, Ma, CA. Competition, gatekeeping, and health care access. Journal of health economics. 2015;39:159-70.

107. Gravelle H, Sutton M, Morris S, Windmeijer F, Leyland A, Dibben C, et al. Modelling supply and demand influences on the use of health care: implications for deriving a needs-based capitation formula. Health economics. 2003;12(12):985-1004.

108. Morris S, Sutton M, Gravelle H. Inequity and inequality in the use of health care in England: an empirical investigation. Social science & medicine. 2005;60(6):1251-66.

109. Carlsen B, Frithjof Norheim O. Introduction of the patient-list system in general practice Changes in Norwegian physicians' perception of their gatekeeper role. Scandinavian journal of primary health care. 2003;21(4):209-13.

110. Atella V, Deb P. Are primary care physicians, public and private sector specialists substitutes or complements? Evidence from a simultaneous equations model for count data. Journal of health economics. 2008;27(3):770-85.

111. Brekke KR, Siciliani L, Straume OR. Hospital competition with soft budgets. The Scandinavian Journal of Economics. 2015;117(3):1019-48.

112. Hoxby CM. Does competition among public schools benefit students and taxpayers? American Economic Review. 2000;90(5):1209-38.

113. Gaynor M. Competition and quality in health care markets. Foundations and Trends® in Microeconomics. 2007;2(6):441-508.

114. Iversen T. The effects of a patient shortage on general practitioners' future income and list of patients. Journal of health economics. 2004;23(4):673-94.

115. Iversen T, Lurås H. Economic motives and professional norms: the case of general medical practice. Journal of Economic Behavior & Organization. 2000;43(4):447-70.

116. Iversen T, Lurås H. 10 The importance of micro-data for revealing incomemotivated behaviour among GPs. Individual Decisions for Health. 2002:182.

117. Godager G, Iversen T, Ma C-TA. Service motives and profit incentives among physicians. International journal of health care finance and economics. 2008;9(1):39.

118. Jelovac I. Primary Care, Gatekeeping and Incentives. In A J Culyer (Ed) Encyclopedia of health economics 2014;3:142-5.

119. Starfield B. Is primary care essential? Lancet (London, England). 1994;344(8930):1129-33.

120. Martin DP, Diehr P, Price KF, Richardson WC. Effect of a gatekeeper plan on health services use and charges: a randomized trial. American journal of public health. 1989;79(12):1628-32.

121. Franks P, Clancy CM, Nutting PA. Gatekeeping revisited--protecting patients from overtreatment. The New England journal of medicine. 1992;327(6):424-9.

122. Iversen T. A study of income-motivated behavior among general practitioners in the Norwegian list patient system. . HERO Working Paper 2005:8 Health Economics Research Programme at University of Oslo, Oslo. 2005.

123. Lurås H. General Practice. Four Empirical Essays on GP Behaviour and the Individuals Preferences for GPs.

UiO Heled: Health Economics Research Programme, University of Oslo 2004 (ISBN 82-7756-132-6)2004.

124. Lurås H. The association between patient shortage and patient satisfaction with general practitioners. Scand J Prim Health Care. 2007;25(3):133-9.

125. Kann IC, Biørn E, Lurås H. Competition in general practice: Prescriptions to the elderly in a list patient system. Journal of health economics. 2010;29(5):751-64.

126. Billinghurst B, Whitfield M. Why do patients change their general practitioner? A postal questionnaire study of patients in Avon. The British journal of general practice : the journal of the Royal College of General Practitioners. 1993;43(373):336-8.

127. Cleary PD, McNeil BJ. Patient satisfaction as an indicator of quality care. Inquiry : a journal of medical care organization, provision and financing. 1988;25(1):25-36.

128. Vick S, Scott A. Agency in health care. Examining patients' preferences for attributes of the doctor-patient relationship. Journal of health economics. 1998;17(5):587-605.

129. Scott A, Vick S. Patients, Doctors and Contracts: An Application of Principal-Agent Theory to the Doctor-Patient Relationship. Scottish Journal of Political Economy. 1999;46(2):111-34.

130. Godager G. Birds of a feather flock together: A study of doctor-patient matching. Journal of health economics. 2012;31(1):296-305.

131. Campbell JL, Carter M, Davey A, Roberts MJ, Elliott MN, Roland M. Accessing primary care: a simulated patient study. British Journal of General Practice. 2013;63(608):e171.

132. Iversen T, Lurås H. Patient switching in general practice. Journal of health economics. 2011;30(5):894-903.

133. Marquis MS, Davies AR, Ware JE. Patient satisfaction and change in medical care provider: a longitudinal study. Med Care. 1983;21(8):821-9.

134. Rubin G, Bate A, George A, Shackley P, Hall N. Preferences for access to the GP: a discrete choice experiment. British Journal of General Practice. 2006;56(531):743.

135. Campbell SM, Hann M, Hacker J, Burns C, Oliver D, Thapar A, et al. Identifying predictors of high quality care in English general practice: observational study. Bmj. 2001;323(7316):784-7.

136. Harris MF, Zwar NA. Care of patients with chronic disease: the challenge for general practice. The Medical journal of Australia. 2007;187(2):104-7.

137. Zhang Y. What Characterizes Patients Who Switch General Practitioners? : University of Oslo; 2012.

138. Sato T, Takeichi, M., Shirahama, M., Fukui, T., Gude, J. K. Doctor-shopping patients and users of alternative medicine among Japanese primary care patients. General hospital psychiatry. 1995;17(2):115-25.

139. Wolinsky FD, Steiber SR. Salient issues in choosing a new doctor. Social science & medicine (1982). 1982;16(7):759-67.

140. Gandhi IG, Parle, J. V., Greenfield, S. M., Gould, S. A qualitative investigation into why patients change their GPs. Family practice. 1997;14(1):49-57.

141. Klemperer P. Competition when Consumers have Switching Costs: An Overview with Applications to Industrial Organization, Macroeconomics, and International Trade. The Review of Economic Studies. 1995;62(4):515-39.

142. Pedersen LB, Kjær T, Kragstrup J, Gyrd-Hansen D. Do General Practitioners Know Patients' Preferences? An Empirical Study on the Agency Relationship at an Aggregate Level Using a Discrete Choice Experiment. Value in Health. 2012;15(3):514-23.

143. McGrail MR, Humphreys, J. S., Ward, B. Accessing doctors at times of needmeasuring the distance tolerance of rural residents for health-related travel. BMC Health Serv Res. 2015;15:212.

144. Chan L, Hart, L. G., Goodman, D. C. Geographic access to health care for rural Medicare beneficiaries. The Journal of rural health : official journal of the American Rural Health Association and the National Rural Health Care Association. 2006;22(2):140-6.

145. Pagano E, Di Cuonzo D, Bona C, Baldi I, Gabriele P, Ricardi U, et al. Accessibility as a major determinant of radiotherapy underutilization: a population based study. Health policy (Amsterdam, Netherlands). 2007;80(3):483-91.

146. Arcury TA, Gesler, W. M., Preisser, J. S., Sherman, J., Spencer, J., Perin, J. The effects of geography and spatial behavior on health care utilization among the residents of a rural region. Health Serv Res. 2005;40(1):135-55.

147. Littenberg B, Strauss, K., MacLean, C. D., Troy, A. R. The use of insulin declines as patients live farther from their source of care: results of a survey of adults with type 2 diabetes. BMC public health. 2006;6:198.

148. Skinner J. Causes and Consequences of Regional Variations in Health Care. In Pauly, MV, McGuire, TG, Barros, PP (Eds) Handbook of health economics. 2011;2:45-93.

149. Intrator O, Grabowski DC, Zinn J, Schleinitz M, Feng Z, Miller S, et al. Hospitalization of nursing home residents: the effects of states' Medicaid payment and bed-hold policies. Health Serv Res. 2007;42(4):1651-71.

150. McClellan M, McNeil, B. J., Newhouse, J. P. Does more intensive treatment of acute myocardial infarction in the elderly reduce mortality? Analysis using instrumental variables. Jama. 1994;272(11):859-66.

151. Bynum JP, Bernal-Delgado, E., Gottlieb, D., Fisher, E. Assigning ambulatory patients and their physicians to hospitals: a method for obtaining population-based provider performance measurements. Health Serv Res. 2007;42(1 Pt 1):45-62.

152. Fisher ES, Staiger, D. O., Bynum, J. P., Gottlieb, D. J. Creating accountable care organizations: the extended hospital medical staff. Health affairs (Project Hope). 2007;26(1):w44-57.

153. InfoMap Norge AS [Available from: <u>http://infomap.no</u>.

154. Høydahl E. Standard for sentralitet Statistics Norway2018 [Available from: <u>http://www.ssb.no/klass/klassifikasjoner/128</u>.

155. Mokienko A, Wangen KR. Disenrollment from general practitioners among chronic patients: a register-based longitudinal study of Norwegian claims data. BMC Fam Pract. 2016;17(1):170.

156. Starfield B, Lemke, K. W., Herbert, R., Pavlovich, W. D., Anderson, G. Comorbidity and the use of primary care and specialist care in the elderly. Annals of family medicine. 2005;3(3):215-22.

157. Wallace E, Salisbury C, Guthrie B, Lewis C, Fahey T, Smith SM. Managing patients with multimorbidity in primary care. BMJ : British Medical Journal. 2015;350:h176.

158. Dickens WT. Error Components in Grouped Data: Is It Ever Worth Weighting? The Review of Economics and Statistics. 1990;72(2):328-33.

159. Lee Jin Y, Solon G. The Fragility of Estimated Effects of Unilateral Divorce Laws on Divorce Rates. The BE Journal of Economic Analysis & Policy2011.

160. Wooldridge JM. Econometric analysis of cross section and panel data. 2nd ed. Cambridge, Mass.: MIT Press; 2010. xxvii, 1064 p. p.

161. Public Health Report: Health Status in Norway 2018. Oslo: Norwegian Institute of Public Health; 2018.

162. Bailey, J. How is inequality measured? Quora2017 [Available from: <u>https://www.quora.com/How-is-inequality-measured</u>.

163. University of Texas Inequality Project [updated 16.09.2014. Available from: <u>http://utip.lbj.utexas.edu/tutorials.html</u>.

164. Appleby J, Raleigh V, Frosini F, Bevan G, Gao H, Lyscom T. Variations in health care: the good, the bad and the inexplicable: King's Fund; 2011.

165. Li B, Lingsma HF, Steyerberg EW, Lesaffre E. Logistic random effects regression models: a comparison of statistical packages for binary and ordinal outcomes. BMC Medical Research Methodology. 2011;11(1):1-11.

166. Larsen K, Petersen JH, Budtz-Jørgensen E, Endahl L. Interpreting Parameters in the Logistic Regression Model with Random Effects. Biometrics. 2000;56(3):909-14.

167. Wennberg JE, Fisher, E. S., Skinner, J. S. Geography and the debate over Medicare reform. Health affairs (Project Hope). 2002;Suppl Web Exclusives:W96-114.
168. Damask Trial Team. Influence of magnetic resonance imaging of the knee on GPs' decisions: a randomised trial. Br J Gen Pract. 2007;57(541):622-9.

169. Espeland A, Natvig NL, Løge I, Engebretsen L, Ellingsen J. Magnetic resonance imaging of the knee in Norway 2002–2004 (national survey): rapid increase, older patients, large geographic differences. BMC health services research. 2007;7(1):115.

170. Wylie JD, Crim JR, Working ZM, Schmidt RL, Burks RT. Physician provider type influences utilization and diagnostic utility of magnetic resonance imaging of the knee. JBJS. 2015;97(1):56-62.

171. Grasdal AL, Monstad K. Inequity in the use of physician services in Norway before and after introducing patient lists in primary care. International Journal for Equity in Health. 2011;10(1):25.

172. Fleurbaey M, Schokkaert E. Chapter Sixteen - Equity in Health and Health Care1. In: Mark V. Pauly TGM, Pedro PB, editors. Handbook of Health Economics. Volume 2: Elsevier; 2011. p. 1003-92.

173. Cutler D, Skinner JS, Stern AD, Wennberg D. Physician Beliefs and Patient Preferences: A New Look at Regional Variation in Health Care Spending. American Economic Journal: Economic Policy. 2019;11(1):192-221.

174. Anthony DL, Herndon MB, Gallagher PM, Barnato AE, Bynum JPW, Gottlieb DJ, et al. How Much Do Patients' Preferences Contribute To Resource Use? Health Affairs. 2009;28(3):864-73.

175. Baker LC, Bundorf MK, Kessler DP. Patients' Preferences Explain A Small But Significant Share Of Regional Variation In Medicare Spending. Health Affairs. 2014;33(6):957-63.

176. Mandelblatt JS, Faul LA, Luta G, Makgoeng SB, Isaacs C, Taylor K, et al. Patient and Physician Decision Styles and Breast Cancer Chemotherapy Use in Older Women: Cancer and Leukemia Group B Protocol 369901. Journal of Clinical Oncology. 2012;30(21):2609-14.

177. Wennberg JE, Barnes BA, Zubkoff M. Professional uncertainty and the problem of supplier-induced demand. Social Science & Medicine. 1982;16(7):811-24.

178. Dranove D, Wehner P. Physician-induced demand for childbirths. Journal of health economics. 1994;13(1):61-73.

179. Finkelstein A, Williams H, Gentzkow M. Sources of Geographic Variation in Health Care: Evidence From Patient Migration*. The Quarterly Journal of Economics. 2016;131(4):1681-726.

180. Campbell SM, Braspenning J, Hutchinson A, Marshall M. Research methods used in developing and applying quality indicators in primary care. Quality and Safety in Health Care. 2002;11(4):358.

181. Lawrence M, Olesen F. Indicators of Quality in Health Care. European Journal of General Practice. 1997;3(3):103-8.

182. Donabedian A. Exploratings in quality assessment and monitoring definition of quality and approaches to its assessment. Ann Arbor. 1980.

183. Ovretveit J. Health service quality: an introduction to quality methods for health services: Blackwell Scientific; 1992.

184. McGlynn EA. Six challenges in measuring the quality of health care. Health affairs. 1997;16(3):7-21.

185. Davies HTO, Crombie IK. Assessing the quality of care. British Medical Journal Publishing Group; 1995.

186. Mant J, Hicks N. Detecting differences in quality of care: the sensitivity of measures of process and outcome in treating acute myocardial infarction. Bmj. 1995;311(7008):793-6.

187. Wensing M, Jung HP, Mainz J, Olesen F, Grol R. A systematic review of the literature on patient priorities for general practice care. Part 1: Description of the research domain. Social Science & Medicine. 1998;47(10):1573-88.

188. Campbell SM, Roland MO, Shekelle PG, Cantrill JA, Buetow SA, Cragg DK. Development of review criteria for assessing the quality of management of stable

angina, adult asthma, and non-insulin dependent diabetes mellitus in general practice. Quality in Health Care. 1999;8(1):6.

189. Musich S, Wang S, Hawkins K, Klemes A. The Impact of Personalized Preventive Care on Health Care Quality, Utilization, and Expenditures. Population Health Management. 2016;19(6):389-97.

190. Larsen L, Sonderlund A, Søndergaard J, Thomsen J, Halling A, Hvidt NC, et al. Targeted prevention in primary care aimed at lifestyle-related diseases: A study protocol for a non-randomised pilot study2018.

191. Tran AT, Bakke Å, Berg TJ, Gjelsvik B, Mdala I, Nøkleby K, et al. Are general practitioners characteristics associated with the quality of type 2 diabetes care in general practice? Results from the Norwegian ROSA4 study from 2014. Scandinavian Journal of Primary Health Care. 2018;36(2):170-9.

192. Marshall MN, Shekelle, P. G., Leatherman, S., Brook, R. H. The public release of performance data: what do we expect to gain? A review of the evidence. Jama. 2000;283(14):1866-74.

193. Gudzune KA, Bleich, S. N., Richards, T. M., Weiner, J. P., Hodges, K., Clark, J. M. Doctor shopping by overweight and obese patients is associated with increased healthcare utilization. Obesity (Silver Spring, Md). 2013;21(7):1328-34.

194. Kaplan SH, Greenfield, S., Ware, J. E., Jr. Assessing the effects of physicianpatient interactions on the outcomes of chronic disease. Med Care. 1989;27(3 Suppl):S110-27.

195. Deall C MH. Effect of Cold Weather on the Symptoms of Arthritic Disease: A Review of the Literature. J Gen Pract (Los Angel) 4:275. 2016.

196. Nutting PA, Goodwin MA, Flocke SA, Zyzanski SJ, Stange KC. Continuity of primary care: to whom does it matter and when? Annals of family medicine. 2003;1(3):149-55.

197. Baker R, Boulton M, Windridge K, Tarrant C, Bankart J, Freeman GK. Interpersonal continuity of care: a cross-sectional survey of primary care patients' preferences and their experiences. The British journal of general practice : the journal of the Royal College of General Practitioners. 2007;57(537):283-9.

198. Hetlevik Ø, Gjesdal S. Personal continuity of care in Norwegian general practice: a national cross-sectional study. Scandinavian journal of primary health care. 2012;30(4):214-21.

199. Hjortdahl P, Laerum E. Continuity of care in general practice: Effect on patient satisfaction. British Medical Journal. 1992;304(6837):1287-90.

200. Schers H, van den Hoogen H, Bor H, Grol R, van den Bosch W. Preference for a general practitioner and patients' evaluations of care: A cross-sectional study. British Journal of General Practice. 2004;54(506):693-4.

201. Storvik AG. Sterk økning i refusjon til radiologi – Riksrevisjonen gransker 2016 [Available from: <u>http://www.dagensmedisin.no/artikler/2016/02/24/sterk-okning-i-refusjon-til-radiologi--riksrevisjonen-</u>

gransker/?x=MjAxNi0wNi0xNSAyMDoyMzoyMg==.

Effects of a reimbursement change and travel times on the delivery of private and public radiology services in Norway: A register-based longitudinal study of Norwegian claims data

Anastasia Mokienko¹

Abstract

Background: This article studies the variation in the impact of the reimbursement change for radiology providers in Norway in 2008 depending on the travel times to private and public providers in different municipalities. The activity-based fund allocation for radiology providers was reduced from approximately 50% to 40%, which was compensated by an increased basic grant. The hypothesis is that Norwegian population would be affected by the reimbursement change unevenly depending on the distances to different types of the providers.

Methods: I used panel data at the municipality level with monthly observations for the period 2007–2010. I studied the effect of the reimbursement change and travel time difference between private and public radiology providers (*Time_difference*) on the number of the services using fixed-effects regressions.

Results: After the reimbursement change, the number of private services decreased more than public services. Private services declined after 2008, but the absolute value of the effect was

¹ Correspondence: anastasiya.mokienko@gmail.com Department of Health Management and Health Economics, University of Oslo,

P.O. Box 1089 Blindern, NO-0317 Oslo

smaller as the *Time_difference* became greater. The number of public services increased as the *Time_difference* grew. The total number of services decreased until the *Time_difference* was equal to 40 min, and increased for time differences greater than 40 min.

Conclusions: The messages for policymakers are as follows: Populations that only had private providers nearby were more affected by the reimbursement change in terms of a reduced number of services; The reimbursement change contributed to the reallocation of patients from private to public providers; The reimbursement change reduced the difference between different centralities of municipalities in their consumption patterns and increase the difference between between different Regional Health Authorities.

Trial Registration: not applicable.

Keywords: radiology providers, reimbursement change, institutional settings, travel time to providers.

Background

Motivation

Radiology services are useful tools in the diagnostic process. When physicians suspect a particular diagnosis, they often send patients for further examination. Sometimes a diagnosis is confirmed, and sometimes not. Can we say that the service was unnecessary if the diagnosis was not confirmed? Some policymakers would say yes. However, a negative answer is still an answer when it comes to diagnostics. A big debate exists about efficiency, cost savings, and cutting unnecessary services. An extensive amount of literature supports the idea that rising costs do not necessarily translate into an increase in the quality and efficiency of health care (1-5). However, it is very difficult to assess the benefit of diagnostics and specialist services,
which makes it easy to say that those services are overused (6). Diagnostics may save a significant amount of budgeted money by helping physicians to assess diagnoses more precisely. When the benefit of diagnostics is unclear, physician discretion is involved, and practice variation should be expected, which could result in patients being overtreated or undertreated.

In addition to practice differences, variation comes from regional differences in distribution of the providers. For instance, the number of radiology services in Norway varies in different regions and municipalities. Research on the accessibility of medical service providers has demonstrated that greater travel distances to providers lead to reduced utilisation of health-care services (7-10) and that patients consume more services from nearby providers (11). There is also a difference in the geographical tolerance of highly versus sparsely populated communities: Residents of closely settled areas are much less willing to travel to access a general practitioner (GP) than people in sparsely populated areas (12). Thus, people who live in remote areas travel to health-care institutions less frequently than those in populated areas, but they are willing to travel much longer distances than people residing in population centres.

Equal access to good quality care is one of the top priorities of health care in Norway (13, 14). Understanding how the variation in the provision of health care services works and how it affects the consequences of political and financial changes helps policymakers to take more thorough decisions. To be able to reach this goal, there is a need for evidence on what contributes to regional differences. The aim of this paper is to provide more evidence on what may add to regional differences in provision of health services using the example of the radiology services in Norway. It examines how the change in the remuneration system in 2008 for radiology providers contributed to a change in the radiology supply in the different geographical regions depending on the travel times to private and public providers.

This topic is important because Norway is a vast country with a small population, and it therefore has many remote municipalities. Not all municipalities have radiology providers, and from some municipalities, the travel time can reach several hours. Some municipalities (medium and large ones) have private and/or public providers, but others do not. When private and public providers react differently to financial changes, it could result in a variation in the supply to patients who have a particular kind of provider available.

Public and private providers

Norway has four regional health authorities (RHAs) named after their locations (Southeast, Central, North, and West). There are two types of radiology providers in Norway: private and public. Private providers operate as for-profit institutions that can have contracts with RHAs and deliver radiology services on public terms (Patients only pay the laboratory a patient co-payment, while the rest is paid by the state and the RHA.). Each RHA chooses a number of private radiology providers through a tendering process and by signing contracts with them for a specific number of services. This option is sometimes associated with wait times for patients. Private providers also deliver radiology services on private terms (when patients pay the full fee directly to the laboratory); this option is not associated with wait times for patients.

The contracts with RHAs specify the volume of and reimbursement for examinations, the maximum number of services, and the total costs. Some contracts specify only an aggregated budget for services (15). Other contracts are very detailed and specify the budget for each type of service, such as ultrasound imaging (UI), magnetic resonance imaging (MRI), computed body tomography (CBT or CAT scans), and radiography (X-rays) (16).

The provider's revenue includes three components: the fee-for-service from the National Health Insurance (NHI) scheme, patient co-payments (the same for both private and public providers when received through NHI), and the invariable component (a basic allocation that is independent of the number of the services provided). 'The size of the basic allocation [was] determined by a number of factors, including the number of inhabitants living in the region and the demographics of the population.' (17)

Public providers are hospital radiology departments that deliver radiology services to the population on public terms; that is, they accept both patients from hospitals and outpatients referred to them by GPs and specialists. The revenue of public providers also includes three components equivalent to those of the private providers. Visits to a public or private laboratory require a referral from a general practitioner (GP) or a specialist to be covered by NHI (18). In theory, radiology laboratories can decline to make an appointment in general, but in practice, this does not happen often because GPs already act as gatekeepers, as illustrated in (19).

The RHA and NHI do not reimburse public radiology laboratories directly; instead, they reimburse the hospital affiliated with the laboratory. This implies that public outpatient providers are not as restricted by contracts as their private counterparts; thus, they have softer budget constraints than private providers do. Soft budget constraints are often related to a poor ability to balance budgets and providers with the tendency to increase activity or costs to a level above the one preferred by the principle stakeholder (20-22). In contrast, in the private sector, the number of services is controlled by hard budget constraints to maintain positive profits because contracts include specified volumes.

The 2008 reimbursement change

Reforms in the financing of specialist health care were carried out since 1997 and activitybased funding (ABF) was introduced to encourage the achievement of activity targets ((23), p 69). If these targets were not met, the RHAs lost income. If the activity levels were higher than targeted, then the costs would be only partially compensated. Hence, in short, ABF was not intended to cover marginal costs or to encourage activity beyond the target ((17), p. 13). In the period from 2005 to 2008, the proportions of activity-based and basic allocation were approximately equal. Figure 1 demonstrate that the prior to 2008 spending for private radiology continually increased. The reimbursement change of 2008 has changed radiology funding accordingly: the ABF part decreased from 50% to 40%, and the basic allocation increased from 50% to 60% to compensate for it. The reimbursement scheme was changed to cut on spending and to harmonise the financial scheme of radiology providers with the general system for financing outpatient medical services in Norway (17, 24) 'Somatic specialist care is financed partly through block grants (60%) and partly through activity-based financing from the central government to the RHAs (40%), with the latter component based on diagnosis-related groups (DRGs). The financing structure is aimed at both containing costs and giving providers sufficient flexibility to assure the best mix of services for patients.' (23)



Figure 1 Market share in the costs between private (blue) and public (red) radiology providers in percent (Figure 3.4 in (25))

Methods

Data

I used claims data obtained from the Norwegian Directorate of Health. The dataset (aggregated at the municipality level) contained the number of radiology services (CAT scans, MRIs, X-rays, and ultrasounds) reimbursed per month by NHI from 2007 to 2010, the travel times from the municipality of the patient to the municipality with closest private or public provider, the number of inhabitants, the centrality of the municipalities, and the RHAs to which they belonged. I monitored 422 municipalities in 48 different periods (monthly observations from 2007 to 2010), for a total of 19,867 observations.

Variables

Travel times

Table 1 in Appendix contains an overview of the variables. The travel times were measured in hours according to driving time by car (provided by Info Map Norway [(26)]) between a patient's residential municipality (approximated by the municipality of the patient's GP) and the municipality of the public radiology provider (*Pubtime*) or the private radiology provider (*Privtime*). If patients have a radiology provider in their own municipality, then the travel time was set to zero by definition in the dataset. The difference in travel time between the nearest private provider and nearest public provider is represented by *Time_difference = Privtime - Pubtime*. The difference in travel time is included as the main independent variable because when deciding between two providers in the settings of unevenly distributed providers, patients often choose the more available provider in terms of proximity. Since private and public providers have different institutional settings, this choice affects outcome.

Centrality

'*Centrality*' of the municipality (*Centrality* 1–7, dummy variables) is a variable from Statistics Norway that classifies every municipality in Norway according to one of seven categories, where 7 represents the most central type of municipality (e.g. Oslo), and 1 denotes the least central ones (e.g. small remote villages). '*Centrality*' indicates the location of municipalities in relation to urban settlements of various sizes (27, 28). It reflects the travel time from an urban settlement to a centre with well-developed infrastructure, including banks, post offices, and so forth, as well as the number of inhabitants and public services available (see (29, 30) for details). Since research indicates that residents of closely settled areas are much less willing to travel to access a health care provider than people in sparsely populated areas (12), '*Centrality*' might not only reflect type of municipality but as well be correlated with patients' willingness to travel.

Regional health authorities

Region 1–4' are dummy variables describing whether the municipality belongs to 1) South East, 2) West, 3) Central, or 4) North RHAs.

Centrality and *'Region 1–4'* are time invariant. They are part of the fixed effects and are therefore cancelled out in the model, but they are used for descriptive statistics.

Number of services

The dependent variable is the number of services provided at private (*Priv_Serv*), Public (*Pub_Serv*), or both types of providers together (*Total_Serv*) per month. It was calculated by accumulating claims in every municipality. If a patient from municipality A goes to municipality B to receive a radiology examination, that service is classified as a service delivered to municipality A. The measurement of this variable reflects the number of services per 1,000 inhabitants in the municipality.

Hypotheses

Patients who live in the centres have better access to both public and private providers, while those who live remotely have to travel up to several hours to reach a provider. I was interested in studying the interaction between patients' travel times and the reimbursement change of 2008 in terms of the number of services consumed. I expected that the Norwegian population would be affected by the reimbursement change unevenly depending on the distances to different types of the providers.

I based my hypotheses on two assumptions (A1 and A2). A1) There is stream of patients who need services, and if one source reduces the offer, the patients will switch to another more readily available source. Under availability, I considered both capacity and travel time to the provider. A2) As mentioned in the background section, public providers have softer budget constraints and can thus stretch their capacity outside of the limits set by budgets compared to private providers, which have hard budget constraints.

Thus, I formulated the following hypotheses.

Hypothesis 1

There will be a larger decrease in number of private services than public services based on the differences in their budget constraints.

Hypothesis 2

The stream of patients who move between providers and the effect on the total number of services will be different depending on the difference in the proximity of private and public radiology providers. The changes at private, public and both providers will be following:

2A) Patients use private radiology more when these providers are relatively closer (i.e. *Time_difference* is negative or equal to zero), which means that, after 2008, the greatest

9

reduction in the *Priv_Serv* will be in these areas. The reduction diminishes with the increase in *Time_difference*.

2B) The change for public providers consists of two effects. The first involves a reduction in the original public service users. The greater usage was before 2008, the greater the reduction in the number of services will become after 2008. In general, patients use public radiology more when these providers are closer (that is, when *Time_difference* is zero or positive). The second effect relates to users switching from private radiology. These patients are more likely to switch the closer they live to a public provider compared to a private provider (i.e. the greater the value of *Time_difference*). Depending on what effect is greater, the change will be positive, negative, or equal to zero.

2C) Since private providers are more affected, the greatest reduction in the total number of services occurs in the areas with negative *Time_difference*. This reduction will diminish with an increase in *Time_difference* because patients can more easily switch to a public provider.

Figure 2 represents a visual explanation of the hypotheses in terms of *Time_difference*—how the consumption of services would change when we move on the scale of *Time_difference* from negative to positive values. Figure 2 makes use of three states: negative, equal to zero, and positive values of *Time_difference*. The text boxes indicate what I expected in each of the three states and why.

10

Time_difference<0 Closest radiology is private After 2008, I expected Priv_Serv to decrease, Pub_Serv to increase least, and Total_Serv to decrease the most.

Reason: The more negative Time_difference is, the less likely it is that patients will move to public providers to compensate for the reduction in private services. Time_difference=0

Equal distance to both types of radiology. After 2008, I expected Priv_Serv to decrease, Pub_Serv to increase, and Total_Serv may not change. Reason: The likelihood of patients moving from private to public radiology to compensate for the reduction in private services is highest. Time_difference>0 Closest radiology is public. After 2008, I expected Priv_Serv to decrease and Pub_Serv to increase or stay the same, and Total_Serv may not change Reason: Patients use more public radiology since it is closer. The likelihood that patients will move from private to public providers increases with the increase in Time_difference.

Figure 2 Three states on the axis of *Time_difference* (the difference in travel time between the nearest private provider and the nearest public provider) and the hypotheses regarding the consumption of radiology services

If we follow Figure 2, the first text box indicates that the closest radiology provider is private $(Time_difference < 0)$. I expected the total number of services to decrease due to reduced offerings from private providers. Since there is a longer travel time to the public provider, fewer patients would move to the public provider due to time costs compared with the other two cases (when the public provider is closer or equally close). Therefore, more patients would rather not take the radiology examination at all or do it out of pocket. Thus, the total number of services would decrease more than if the closest radiology provider were public.

The second box indicates that the distance between private and public radiology providers is small (*Time_difference* \rightarrow 0). In this situation, patients can change providers more easily. The likelihood that patients will switch from private to public radiology provider is higher. Thus, I expect a substantial drop in *Priv_Serv* and an increase in *Pub_Serv*, while the total number of the services may not even change.

In the third textbox, the closest radiology provider is public ($Time_difference > 0$). I expected that patients would use the public provider more than private ones. Since public providers have softer budget constraints, I predicted that the total number of services would be less affected by the reimbursement change. However, some patients who used private providers before 2008 would move to public providers due to the reduced offerings of private providers after 2008. Therefore, I expected that the total number of services would stay the same, public services would increase or stay the same, and private services would decrease or stay the same.

Model

I estimated how *Time_difference* would affect number of services at private, public, and both providers after the reimbursement change. Using the panel data, we can control for time-invariant heterogeneity without observing it. I proceeded using a fixed effects model because it is more robust and needs fewer assumptions fulfilled compared to a random effects model. The fixed effects model is based on the assumption that the errors are uncorrelated with the independent variables and that the errors are conditionally homoscedastic and not serially correlated (31).

The model includes the differences in the time it takes for patients to reach a private provider minus time it takes for them to reach a public provider (*Time_difference = Privtime - Pubtime*). After trying several polynomial functions, I used a quadratic function because I expected that the relationship between number of the services and the *Time_difference* would not be completely linear. A regression model was estimated separately for each of the samples of private and public providers, as well as for the sample including both type of providers:

 $Y_{it} = B_0 + B_1 \text{ post08}_t + B_2 \text{ post08}_t \cdot \text{Time_difference}_{it} + B_3 \text{ post08}_t \cdot \text{Time_difference}_{it}^2 + e_i + u_{1it}$

where Y_{it} denotes the number of services (*Priv_Serv*, *Pub_Serv*, *Total_Serv*) to municipality *i* (*i*=1,...,422) in period *t* (*t*=1,...,48), *post08*_t is a dummy equal to 0 prior to 2008 and 1 after

January 1, 2008, and B_k (k=0...3) are the regression coefficients; e_i is a provider specific fixed effect, and u_{1it} is an error term.

*Pubtime*_{it}, *Privtime*_{it} and *Time_difference*_{it} do not vary much over time for the same municipalities, but I still used 'it'-indexes to indicate even a very small variation (although the variation is not enough to keep them as independent variables in the fixed-effects model without interaction effect with *post08*).

Results

Descriptive statistics

Tables 1A, 1B, and 1C in Appendix display descriptive statistics at different levels for the whole period of 4 years for the whole country: before and after the change for the whole country (Table 1A); before and after the change according to each level of centrality (Table 1B); and before and after the change for each RHA (Table 1C).

Average driving time to the nearest private provider is 3 h: It can be in the same municipality (as little as 0 h away) or in another region (up to 18 h away). The average driving time to the nearest public provider was a little over 1 h and had a smaller range than the driving time to a private provider, ranging from being in the same municipality to being almost 6.5 h away (see Table 1A).

Figure 3 illustrates the distribution of the variable *Time_difference*, which is continuous, concentrated around zero, and mostly to the right-hand side of zero. It has a long right tale and a left-sided truncation. There are many municipalities in which both types of providers were equally close (30%). In addition, in many municipalities, public providers were much closer than private providers (i.e. observations to the right of zero, 65%). In a few municipalities,

private providers were closer than public providers (i.e. observations to the left of zero, 5%). The average *Time_difference* was about 1.5 h.



Figure 3 Distribution of *Time_difference* (the difference in travel time between the nearest private provider and nearest public provider) in hours

The summary statistics for the whole Norwegian population (Table 1A) indicates that the number of examinations per 1,000 inhabitants in the municipality conducted at private providers per month has decreased by 1.6 (from 9.6 to 8), while the corresponding number for public providers has increased by 1.8 (from 24.4 to 26.2) after the change.

Table 1B illustrates the stratified values of travel times according to the different centralities. It indicates that depending on the centrality of the municipality, patients may have both providers available in the same municipality or a short distance away, or they may have to drive at least 1 - 2 h to the closest private or public provider. Table 1B demonstrates that the population in the least central places use most frequently public providers, while population in the most central locations use most frequently private providers. The most central

municipalities saw a reduction in the number of private examinations by around four examinations per 1,000 inhabitants in the municipality (from 18.4 to 14.8); the reduction in the least central municipalities was a little less than 1 (from 6.2 to 5.6). A different picture emerged for examinations at public providers, with an increase of 3 at the least central locations (from 26.8 to 29.7) and 1.5 in the most central municipalities (from 17.2 to 18.9). In total, there was a slight increase in the numbers in the least central municipalities and a slight decrease in the most central municipalities.

Table 1C illustrating the summary statistics stratified by the RHA, demonstrates that private services (range 2–11 examinations/1,000 inhabitants x municipality after the reform) were most strongly represented in the South East, followed by the West and Central municipalities, with the smallest number located in the North. All of them decreased by 2, but in terms of the percentage of change from the original number, the largest and smallest reductions occurred in the North and the South East, respectively. Public services (range 22–32 examinations/1,000 inhabitants x municipality after the reform) were most strongly represented in the North, followed by the Central and South East areas, with the smallest number being located in the West. They increased by 1–3 services after the reform (Central, North, South East, and West, in descending order). The total number was more or less the same in the South East and the North, and it decreased by 1 in the West and increased by 2 in the Central RHA.

Regressions

The models were estimated using xtreg in Stata 13. To test the first hypothesis, I studied the coefficients next to '*post08*' in Table 2. I found that, after the reimbursement change, patients received fewer services at private providers (coefficient = -1.913) and more services at public providers (coefficient = 1.439) than before 2008. The total number of examinations has

declined (coefficient = -0.474). The descriptive statistics in Tables 1A, 1B, and 1C complete the picture, especially the overviews stratified by *Centrality* and RHA.

Thus, hypothesis 1 is supported: The number of services offered by private providers has declined more than the number of services at public providers. Furthermore, *Pub_Serv* has increased, which indicates that the stream of the patients switching from private to public providers is greater than the reduction in public services due to the 2008 reimbursement change.

	Private	Public	Total
Post08	-1.913***	1.439***	-0.474***
	(0.070)	(0.141)	(0.164)
<i>Time_difference</i> x <i>post08</i>	0.328***	0.348***	0.676***
	(0.049)	(0.100)	(0.116)
<i>Time_difference</i> ² x <i>post08</i>	-0.016***	-0.003	-0.020**
	(0.004)	(0.008)	(0.010)
Cons	9.579***	24.273***	33.852***
	(0.048)	(0.097)	(0.113)
R^2	0.043	0.017	0.005

Table 2 Fixed-effects linear regression models for the number of private, public, and total services

Coefficients (standard errors): * p < 0.10, ** p < 0.05, *** p < 0.01. n = 422, N = 19,867, n - number of municipalities, N - number of obs. (n = 422; N = 19,867). Xtreg procedure in Stata 13 was used for estimations.

To test the second hypothesis, I studied the combined coefficients for '*post08*' + '*Time_difference* x *post08*' + '*Time_difference*² x *post08*' together (see Table 2). Figures 4A, 4B, 4C represent quadratic functions based on the coefficients in Table 2 after 2008 for easier understanding of the results. I created three separate graphs for 'private', 'public', and 'total

number' of services. All the curves are described completely; however, the study's main interest is to look at the values within the minimum and maximum range of *Time_difference*, which are -1.92 and 17.53 hours, respectively. The curves describe the change in the number of services after 2008.

Private providers (Figure 4A). The curve is below 0. It indicates a reduction in services after 2008, but the reduction diminished along *Time_difference* until *Time_difference* = 10. Then, the reduction increased again in absolute value.



Figure 4A Curve representing the results for the number of services at private providers and the minimum and maximum *Time_difference* (black lines). The number of services was measured in services/1,000 inhabitants. The *Time_difference* was measured in hours

2) Public providers (Figure 4B). Within our minimum and maximum range of *Time_difference*, the curve is positive, increasing, and almost linear. It means that the public services increased in conjunction with *Time_difference* in our minimum and maximum ranges.



Figure 4B Curve representing the results for the number of services at public providers and the minimum and maximum *Time_difference* (black lines). The number of services was measured in services/1,000 inhabitants. The *Time_difference* was measured in hours

3) Total number of services (Figure 4C). Within our minimum and maximum range of *Time_difference*, for the values of *Time_difference* below 40 min, the curve is below the x-axis (i.e. there, the number of services decreases). For the values of *Time_difference* greater than 40 min, there was an increase in the services after 2008, and it increased along with *Time_difference*. If *Time_difference* equalled 40 min, then there was zero change in the number of total services after 2008.



Figure 4C Curve representing the results for the total number of services, as well as the minimum and maximum *Time_difference* (black lines). The number of services was measured in services/1,000 inhabitants. The *Time_difference* was measured in hours

Here, we take a look at the curves when $Time_difference = 0$. This condition only leaves us with coefficients next to '*post08*' in Table 2. I expected that I would observe negative coefficients for private services but positive coefficients for public services with roughly equal numbers of each (i.e. patients would shift from the private to the public sector, and the number of total services would not change much). The results indicate that the coefficient for private providers was -1.913, the coefficient for public provider was 1.439, and the coefficient for the total number of services was -0.474. The coefficient for the total number was small (-0,474) but still different from 0. Therefore, even in areas with an equal distance to both types of providers, there was a small reduction in the total number of services. Hence, the effect of the reduction in private services was stronger than the compensation in public services in the municipalities with euqual distances to both types of providers.

To sum up, from the graphs, it appears that private services decreased, and they reduced more for negative values of *Time_difference*, with a diminishing reduction until *Time_difference* reached approximately 10 h. Public services increased along with *Time_difference*, and total services decreased until *Time-difference* equalled 40 min and increased for *Time_difference* values above 40 min. The hypotheses 2A, 2B, 2C about private and public services are supported by the results.

Variation measure

To elaborate and evaluate the results, I wanted to include a variation measure of the change based on the summary statistics results. The inspiration behind introducing a variation measure came from the inequality measure techniques, which were adapted to this particular case (32-34). I compared minimum and maximum values of means before and after 2008 for different centralities and RHAs for private, public, and both services together. The results are presented in Tables 3A and 3B in Appendix. The way I conducted it was as following. I looked on the summary statistics that are displayed in Tables 1B and 1C. For every line for private, public, and both services, I found minimum and maximum means before and after 2008 and transferred them to Tables 3A and 3B, respectively. Afterwards, I identified the difference between the minimal and maximal means (column 'Max-Min'). Then, I determined whether that difference increased or decreased after the reform and put the results in percentage form in the column 'Change, %'. For the purposes of the evaluation of the reform, the row with total number of services is of greatest interest.

After 2008, the range for the total numbers of services according to centralities has decreased from range [30.43-36.74] to range [32.46-35.50] (services per capita), indicating that less variation occurred between municipalities belonging to different centralities. It is interesting to note that the largest municipalities (highest centrality) are not the greatest consumers of radiology services per capita but are just average. The variation comes from the small and middle-sized municipalities. If we compare different RHAs, then we observe increased variation. The range has increased from [31.20 - 36.34] before 2008 to [30.05 - 36.19] after 2008 (services per capita).

The results reveal that variation has reduced between different centralities by 51.8% and increased between different RHAs by 19.5%. It is, however, curious that, for example, in the Central RHA, the total number has increased from 32 to 35, whereas in the West RHA, the number has decreased from 31 to 30. These findings do not paint very uniform picture of the consumption of radiology services according to location.

Discussion

To explain the results for private providers, I wanted to look at the travel times and *Time_difference*. Distance to the health care providers is an important factor for patients (35). Demand for healthcare services changes amongst other factors due to variations in the travel time required to receive services, so service utilisation is inversely related to travel times (7-10, 36). Thus, the closest providers are used most frequently. Since private providers had harder budget constraints, they reduced their supply of services more after the reimbursement change. Thus, the patients who had private providers closer to them were affected by the change more than those who had public providers nearby. The number of the private services dropped by 15%–20% after 2008.

It is worth noting that only 5% of the municipalities had shorter distances to private providers than to public providers (21 municipalities with an average population of 13,000 people, ranging from 2,500 to 45,000). Most of the municipalities had equal distances to providers (124 municipalities with an average population 19,500 people, ranging from 345 to 573,000 people).

Next, we turn to an explanation of the results for public providers. Public providers had soft budget constraints, so they were not restricted by the 2008 change as much as their private-sector counterparts. Thus, first, when public providers were closer, patients would use more public services. Second, the smaller the distance between providers was, the easier it was for patients to 'transfer' to providers with more availability (i.e. to a public provider). In other words, patients tend to go to the provider where it is easiest to access services in terms of both travel times and availability. Thus, with the increase in *Time_difference* (i.e. a public provider becomes relatively closer than a private one), the number of the public services increases because it is easier for patients to transfer to the provider that is both available and relatively closer than a private provider. A report from the Norwegian Health Economics Administration

(25) supports these findings. The researchers reported a shift in the number of services towards public providers after the reimbursement change (see p. 27, Figure 5.1 in the report (25)). The report also indicates that refunds for private and public providers were 49%/51% in 2005–2007; after 2008, they became 46%/54%.

In general, the fact that in some areas many patients transferred from private to public providers is not nessesery efficient, because it indicates that patients that are rationed by private providers are now treated by public providers, shifting the market from private to public providers and implying that not only patients with the lowest expected benefit were rationed from private providers. If policy-makers want a coherent effect across providers, all providers should have hard budget constraints.

Thus, after uniting the outcomes for both types of providers, we observed that the smaller *Time_difference* becomes (i.e. a private provider is closer than a public provider), the greater the reduction in the number of total services is. Two causes brought about this change. First, patients used the closest provider (in this case, a private provider) more often, so proportionally the reduction became greater. Second, a possible transfer to a public provider is connected with time costs since private providers are not as far away. Therefore, more patients fall off, (i.e. they either chose not to have certain examinations, paid out of pocket, or used private health insurance). The total numbers depend on the composition of the private and public numbers. In areas where public providers dominate, the total numbers increased (i.e. centralities 1, 2, and 4). The opposite picture emerged in municipalities where private providers are more dominant; private services reduce and drag down the number of total services (*Centrality* 7) (see Table 1B).

Further research is required regarding whether the reduction in total services nationally was due to some of these services being unnecessary in the start (especially in areas where patients had providers close by) or whether people stopped waiting or paid for radiology examinations privately out of pocket (11, 19, 37). The number of patients using private health insurance went up 12 times from 2006 to 2016, and 30% of such plans were used for specialists and diagnostics (38). Thus, it would be valuable to investigate whether the reimbursement change added to the general equity in healthcare access: whether it increased offer of radiology services in areas with underused services and reduced that offer in areas of overuse taking in account individual need for services (39, 40). In the context of regional variation for radiology services, further examination to which degree the overuse and underuse of the services come from GP preferences (41), patients' specific characteristics (42-44) or purely from organizational structure (41, 45) would add clarity to understanding of the variation in radiology services. This study limits are missing data regarding wait times, examinations paid fully out of pocket, examinations covered by private health insurance, the number of dropped examinations, and provider capacities (i.e. the optimal work load for the providers in terms of efficiency).

It is also important to identify the criteria used to decide whether or not a service is necessary. In general, all diagnostics are necessary, and a negative answer is a great answer. However, from a health economics perspective, we should measure the marginal health benefits (improvement in health) gained from services received. However, diagnostics do not constitute a procedure to improve health but rather a step in order to know how to do achieve better health outcomes.

It would have been beneficial to have data for more years prior to the change in order to have more material with which to study the effects of the change. I have accomodated this limitation by including information on radiology development for 2002-2009 (Figure 1) to the present descriptional study. It would have been useful to control for life-expectancy, income level, variation in education in the municipalities. However, these characteristics were included in municipality fixed effects.

Conclusion

The geographical distribution of the providers and the different ways that providers react to changes in the reimbursement system affect the implications of the reimbursement change for publicly reimbursed providers in 2008. Policymakers can take away three messages from these findings: 1) Populations that only had private providers nearby were more affected than others by the reimbursement change in terms of the reduced number of services; 2) the reimbursement change contributed to the reallocation of patients from private to public providers; and 3) the reimbursement change reduced the difference between different centralities of municipalities in their consumption pattern and increased the difference between different RHA regions.

Abbreviations

- ABF Activity-based funding
- CBT or CAT scan Computed body tomography
- GP General practitioner
- HELFO Norwegian Health Economics Administration
- MRI Magnetic resonance imaging
- NHI National Health Insurance
- RHA Regional Health Authority

UI – Ultrasound imaging

Declarations

Ethics approval and consent to participate

The Norwegian Health Directorate has provided data and ethics approval.

Consent for publication

Not applicable.

Availability of data and materials

These data are sensitive without general access, so I applied for an exemption from the obligation to release the data publicly.

Competing interests

There are no competing interests.

Funding

Funding was provided by the Research Council of Norway, project no. 204677/H10.

Authors' contributions

This is one-author paper; hence, all the work was completed by Anastasia Mokienko.

Acknowledgements

I am grateful to Tor Iversen and Knut Reider Wangen (University of Oslo) for their valuable comments, the Norwegian Health Directorate for the data used, and the Research Council of Norway for providing funding.

Authors' information

Anastasia Mokienko,

PhD Candidate

Department of Health Management and Health Economics,

University of Oslo,

P.O. Box 1089 Blindern, NO-0317 Oslo.

Correspondence: anastasiya.mokienko@gmail.com

References

1. Baicker K, Chandra A. Medicare spending, the physician workforce, and beneficiaries' quality of care. Health Aff (Millwood). 2004;Suppl Web Exclusives:W4-184-97.

2. Yasaitis L, Fisher ES, Skinner JS, Chandra A. Hospital quality and intensity of spending: is there an association? Health Aff (Millwood). 2009;28(4):w566-72.

3. Sirovich BE, Gottlieb DJ, Welch HG, Fisher ES. Regional variations in health care intensity and physician perceptions of quality of care. Ann Intern Med. 2006;144(9):641-9.

4. Fisher ES, Wennberg DE, Stukel TA, Gottlieb DJ, Lucas FL, Pinder EL. The implications of regional variations in Medicare spending. Part 2: health outcomes and satisfaction with care. Ann Intern Med. 2003;138(4):288-98.

5. Skinner JS, Stalger DO, Fisher ES. Is technological change in medicine always worth it? The case of acute myocardial infarction. Health Affair. 2006;25(2):W34-W47.

6. Chandra A, Sabik L, Skinner J. COST GROWTH IN MEDICARE: 1992-20062019.

7. Pagano E, Di Cuonzo D, Bona C, Baldi I, Gabriele P, Ricardi U, et al. Accessibility as a major determinant of radiotherapy underutilization: a population based study. Health policy (Amsterdam, Netherlands). 2007;80(3):483-91.

8. Chan L, Hart, L. G., Goodman, D. C. Geographic access to health care for rural Medicare beneficiaries. The Journal of rural health : official journal of the American Rural Health Association and the National Rural Health Care Association. 2006;22(2):140-6.

9. Arcury TA, Gesler, W. M., Preisser, J. S., Sherman, J., Spencer, J., Perin, J. The effects of geography and spatial behavior on health care utilization among the residents of a rural region. Health Serv Res. 2005;40(1):135-55.

10. Littenberg B, Strauss, K., MacLean, C. D., Troy, A. R. The use of insulin declines as patients live farther from their source of care: results of a survey of adults with type 2 diabetes. BMC public health. 2006;6:198.

11. Iversen T, Ma CT. Market conditions and general practitioners' referrals. International journal of health care finance and economics. 2011;11(4):245-65.

12. McGrail MR, Humphreys, J. S., Ward, B. Accessing doctors at times of need-measuring the distance tolerance of rural residents for health-related travel. BMC Health Serv Res. 2015;15:212.

13. National Strategy to Reduce Social Inequalities in Health, Report No. 20. Ministry of Health and Care Services. 2007.

14. Meld. St. 44 (1995-1996) Ventetidsgarantien - kriterier og finansiering. Sosial- og helsedirektoratet. [White Paper no. 44 (1995-1996) The Waiting-time Guarantee - Criteria and Financing. Ministry of Health and Care Services (ed.)]. 1995.

15. Northern Norway Regional Health Authority. Contract of Purchize of Policlinic Radiology Services between Curato Røntgen AS and North RHA. In Norwegian [Helse Nord Rammeavtale Kjøp av polikliniske radiologiske tjenester mellom Curato Røntgen AS og Helse Nord RHF] 2013 [

16. South east Norway Regional Health Authority. Contract of Purchize of Policlinic Radiology Services with Unilabs. In Norwegian. 2013 [

17. The Norwegian Directorate for Health and Social Affairs. Activity-Based Funding of Health Services in Norway. An Assessment and Suggested Measures.; 2007.

18. South East Norway Regional Health Authority. Contract of Purchase of Outpatient Radiology Services. In Norwegian. 2014 [19. Iversen T, Mokienko A. Supplementing gatekeeping with a revenue scheme for secondary care providers. International Journal of Health Economics and Management. 2016:1-21.

20. Kornai J. Resource-Constrained versus Demand-Constrained Systems. Econometrica. 1979;47(4):801-19.

21. Kornai J. "Hard" and "soft" budget constraint. Acta Oeconomica. 1980;25(3/4):231-45.

22. Tjerbo T, Hagen TP. Deficits, Soft Budget Constraints and Bailouts: Budgeting after the Norwegian Hospital Reform. Scandinavian Political Studies. 2009;32(3):337-58.

23. Ringard Å, Sagan, A., Saunes, I.S., Lindahl, A.K. Norway Health system review. Health systems in Transition. 2013. Contract No.: No 8 2013.

24. The Royal Proposition on State Budget for Budget Year 2008. St.prp. nr. 1 (2007–2008). In Norwegian.: Hearing before the The Royal Finance Ministry(2008).

25. Norwegian Health Economics Administration. Analysis of Total Spending on Outpatient Laboratory and Radiology Services. Analysis Report 4 2010. In Norwegian: Analyse av samlede utgifter til polikliniske laboratorieog røntgentjenester. Analyserapport r 4 2010. Versjon 1.0. 1 juni 2010.

26. InfoMap Norge AS [Available from: <u>http://infomap.no</u>.] Accessed 02.03.2019.

27.Statistics Norway. One in five regular GPs are immigrants. 2009 [13.08.2018]. Available from:

http://www.ssb.no/en/helse/artikler-og-publikasjoner/one-in-five-regular-gps-are-immigrants.]Available from:
Accessed
02.03.2019.

28. Statistics Norway. Classification of centrality 2018 [Available from: <u>https://www.ssb.no/a/metadata/conceptvariable/vardok/927/en</u>.] Accessed 02.03.2019.

29. Høydahl E. Standard for sentralitet Statistics Norway2018 [Available from: <u>http://www.ssb.no/klass/klassifikasjoner/128</u>.] Accessed 02.03.2019.

 30.
 Høydahl E. Ny sentralitetsindeks for kommunene. Notater. Documents. 2017/40. Statistics Norway

 2017
 [Available
 from:
 <u>http://www.ssb.no/befolkning/artikler-og-publikasjoner/attachment/330194? ts=15fdd63c098</u>.] Accessed 02.03.2019.

31. Wooldridge JM. Econometric analysis of cross section and panel data. 2nd ed. Cambridge, Mass.: MIT Press; 2010. xxvii, 1064 p. p.

32. Bailey, J. How is inequality measured? Quora2017 [Available from: <u>https://www.quora.com/How-is-inequality-measured</u>.] Accessed 02.03.2019.

33. University of Texas Inequality Project [updated 16.09.2014. Available from: <u>http://utip.lbj.utexas.edu/tutorials.html</u>. Accessed 02.05.2019.

34. Appleby J, Raleigh V, Frosini F, Bevan G, Gao H, Lyscom T. Variations in health care: the good, the bad and the inexplicable: King's Fund; 2011.

35. Pedersen LB, Kjær T, Kragstrup J, Gyrd-Hansen D. Do General Practitioners Know Patients' Preferences? An Empirical Study on the Agency Relationship at an Aggregate Level Using a Discrete Choice Experiment. Value in Health. 2012;15(3):514-23.

36. Cauley SD. The Time Price of Medical Care. The Review of Economics and Statistics. 1987;69(1):59-66.

37. Storvik AG. Sterk økning i refusjon til radiologi – Riksrevisjonen gransker 2016 [Available from: http://www.dagensmedisin.no/artikler/2016/02/24/sterk-okning-i-refusjon-til-radiologi--riksrevisjonen-gransker/?x=MjAxNi0wNi0xNSAyMDoyMzoyMg==] Accessed 02.05.2019.

38. Christiansen H. 12 ganger så mange private helseforsikringer på 10 år. Aftenposten. 19.07.2017.

39. Fleurbaey M, Schokkaert E. Chapter Sixteen - Equity in Health and Health Care1. In: Mark V. Pauly TGM, Pedro PB, editors. Handbook of Health Economics. Volume 2: Elsevier; 2011. p. 1003-92.

40. Grasdal AL, Monstad K. Inequity in the use of physician services in Norway before and after introducing patient lists in primary care. International Journal for Equity in Health. 2011;10(1):25.

41. Cutler D, Skinner JS, Stern AD, Wennberg D. Physician Beliefs and Patient Preferences: A New Look at Regional Variation in Health Care Spending. American Economic Journal: Economic Policy. 2019;11(1):192-221.

42. Anthony DL, Herndon MB, Gallagher PM, Barnato AE, Bynum JPW, Gottlieb DJ, et al. How Much Do Patients' Preferences Contribute To Resource Use? Health Affairs. 2009;28(3):864-73.

43. Baker LC, Bundorf MK, Kessler DP. Patients' Preferences Explain A Small But Significant Share Of Regional Variation In Medicare Spending. Health Affairs. 2014;33(6):957-63.

44. Mandelblatt JS, Faul LA, Luta G, Makgoeng SB, Isaacs C, Taylor K, et al. Patient and Physician Decision Styles and Breast Cancer Chemotherapy Use in Older Women: Cancer and Leukemia Group B Protocol 369901. Journal of Clinical Oncology. 2012;30(21):2609-14.

45. Wennberg JE, Barnes BA, Zubkoff M. Professional uncertainty and the problem of supplier-induced demand. Social Science & Medicine. 1982;16(7):811-24.

•	IX	
	0	
	G	
	Q	
	Q	

Table 1 Variables and Summary statistics: A) Nationally, B) stratified by centrality, C) stratified by Regional Health Authority before/after 2008 (n is the number of groups (i.e. municipalities), and N is the number of observations)

Variable	Explanation of variables	Mean, N=19867 n=422	Std. Dev.	Min	Max	Mean before 2008 N=4944,	Std Dev.	Mean after 2008, N=14923,	Std. Dev
τ Α			č	c		n=415***	1 1 1	n=422***	
Priv_Serv	Number of services per 1,000 inhabitants	8.39	6.91	0	47.60	9.59	7.55	8.00	6.64
	in the municipality done at private providers per month, services/1,000 inhabitants								
Pub_Serv	Number of services per 1,000 inhabitants	25.71	10.43	0	86.90	24.35	9.86	26.16	10.58
	in the municipality done at public providers per month, services/1,000 inhabitants								
$Total_Serv$	Total number of services per 1,000	34.11	9.98	1.39	96.86	33.94	9.88	34.16	10.01
	inhabitants in the municipality per month, services/1,000 inhabitants x month								
Privtime	The travel time from the municipality of	2.57	3.04	0	18.41	2.56	3.01	2.57	3.05
	the patient** to the nearest municipality with a private radiology provider in hours								
Pubtime	The travel time from the municipality of	1.11	1.11	0	6.37	1.11	1.11	1.12	1.12
	the patient** to the nearest municipality with a public radiology provider in hours								
Time_differen	Privtime-Pubtime in hours	1.46	2.56	-1.92	17.53	1.45	2.55	1.46	2.56
се									
Centralities	Dummies (1 is the least central			0	1				
1–7 Centralitv 1	municipality, 7 the most central) Least central: Centrality 0A*	0.21	0.40	0	1	0.21	0.41	0.20	0.40

Table 1A Variables and summary statistics nationally before/after 2008

0.34	0.36	0.38	0.39	0.25	0.27	0.49	0.40	0.39	0.40
0.13	0.16	0.17	0.19	0.07	0.08	0.41	0.20	0.19	0.20
0.33	0.36	0.38	0.39	0.26	0.28	0.49	0.40	0.39	0.40
0.13	0.15	0.17	0.19	0.07	0.08	0.40	0.20	0.19	0.20
1	1	1	1	1	1	1	1	1	1
0	0	0	0	0	0	0	0	0	0
0.34	0.36	0.38	0.39	0.26	0.27	0.49	0.40	0.39	0.40
0.13	0.16	0.17	0.19	0.07	0.08	0.41	0.20	0.19	0.20
	1,000 to	1,000 to	1 15,000	15,000 to	pulation	Health	Health	Health	Health
$y 0B^*$	tion from	ttion from	ation fron	tion from 1	ity 3*, pc	Regional	Regional	Regional	Regional
Centralit	[*] , popula	, popula	*, popul	, popula	Central	the th East	the	the tral	the th
central:	lity 1A*	lity 1B*	lity 2A ³ 00	lity 2B*	central: 0,000	iy for ity Sout	ity Wes	ity Cen	ity Nort
Least c	Centra 15,000	Centra 15,000	Centra to 50,0	Centra 50,000	Most 6 from 56	Dumm Author	Dumm Author	Dumm Author	Dumm Author
Centrality 2	Centrality 3	Centrality 4	Centrality 5	Centrality 6	Centrality 7	Region I	Region 2	Region 3	Region 4

* Centralities classified by Statistics Norway. ** Municipality of the GP of the patient was used as a proxy of the municipality of the patient, as patients usually choose a GP in the same municipality, except in municipalities that are so small that the nearest GP is in another nearest municipality. *** The difference in the number of municupalities in the data is due to structure of claims data. The total number of municipalities is 422.

Mariable	N=1037,	N=3053,	N=624,	N=1937	N=760,	N=2337	N=843,	N=2559,	N=924,	N=2770,	N=348,	N=1043	N=408,	N=1224
variable	<i>n</i> =88	<i>n=</i> 90	<i>n</i> =52	, <i>n</i> =55	<i>n</i> =64	, <i>n</i> =65	<i>n</i> =71	n=72	n=77	n=77	<i>n</i> =29	, <i>n</i> =29	<i>n</i> =34	, <i>n</i> =34
	1 before	1 after,	2 before	2 after	3 before	3 after	4 before	4 after	5	5 after	9	6 after	7	7 after
									before		before		before	
Deire Conn	6.23 (6.44)	5.58	8.28	7.64	11.00	8.76	6.60	4.94	13.91	11.97	4.08	2.30	18.43	14.83
		(5.79)	(99.9)	(6.29)	(6.50)	(5.64)	(5.10)	(4.43)	(7.19)	(6.48)	(4.51)	(3.86)	(7.20)	(6.55)
	26.77	29.72	22.15	25.52	23.13	23.70	27.28	29.09	22.83	23.52	28.98	30.87	17.21	18.88
rup_dur	(11.17)	(11.28)	(90.6)	(9.55)	(8.46)	(6.31)	(10.07)	(10.33)	(8.15)	(10.26)	(9.46)	(8.28)	(7.02)	(8.20)
Total Coni	33.06	35.30	30,43	33.15	34.13	32.46	33.89	34.03	36.74	35.50	33.07	33.84	35,63	33.70
	(11.70)	(11.81)	(6.67)	(9.32)	(8.78)	(0:50)	(10.05)	(10.19)	(8.80)	(16.6)	(9.54)	(8.20)	(90.9)	(7.21)
Drivtimo	3.52 (2.08)	3.48	2.26	2.25	1.51	1.51	4.14	4.29	0.66	0.66	6.35	6.34	0.38	0.38
		(2.10)	(1.59)	(1.58)	(0.94)	(0.93)	(4.49)	(4.60)	(0.40)	(0.40)	(4.42)	(4.40)	(0.29)	(0.29)
0	1.78 (1.20)	1.77	1.22	1.22	0.94	0.94	1.44	1.49	0.35	0.35	1.49	1.48	0.27	0.27
rubune		(1.19)	(0.75)	(0.74)	(0.55)	(0.54)	(1.37)	(1.45)	(0.29)	(0.29)	(1.54)	(1.54)	(0.25)	(0.25)
Time_diffe	1.74 (1.94)	1.71	1.04	1.03	0.57	0.57	2.71	2.80	0.30	0.31	4.86	4.85	0.11	0.11
rence		(1.93)	(1.43)	(1.42)	(0.89)	(0.88)	(4.03)	(4.06)	(0.38)	(0.38)	(3.59)	(3.59)	(0.22)	(0.22)
Region 1	0.23 (0.42)	0.24	0.25	0.26	0.35	0.35	0.31	0.31	0.79	0.79	0.03	0.03	0.82	0.82
		(0.43)	(0.43)	(0.44)	(0.48)	(0.48)	(0.46)	(0.46)	(0.41)	(0.41)	(0.18)	(0.18)	(0.38)	(0.38)
Region 2	0.18 (0.39)	0.19	0.31	0.31	0.36	0.35	0.19	0.18	0.14	0.14	0.03	0.03	0.06	0.06
		(0.39)	(0.46)	(0.46)	(0.48)	(0.48)	(0.39)	(0.38)	(0.35)	(0.35)	(0.18)	(0.18)	(0.24)	(0.24)
Region 3	0.29 (0.45)	0.29	0.37	0.35	0.22	0.23	0.13	0.13	0.06	0.06	0.14	0.13	0.08	0.09
		(0.46)	(0.48)	(0.48)	(0.42)	(0.42)	(0.33)	(0.33)	(0.25)	(0.25)	(0.34)	(0.35)	(0.28)	(0.28)
Region 4	0.30 (0.46)	0.28	0.08	0.07	0.06	0.06	0.37	0.39	0	0	0.79	0.79	0.03	0.03
		(0.45)	(0.27)	(0.26)	(0.24)	(0.24)	(0.48)	(0.49)			(0.40)	(0.41)	(0.17)	(0.17)
*The first nur	nber is mean.	the number	in parenthesi	s is the star	ndard deviati	on. N is nu	imber of obs	ervations, n	is number	of municipal	lities, and T	is mean n	umber of p	eriods.
T hefore=12.	T after=36.		4							•			•	
	1 41 WT													

Table 1B Summary statistics stratified according to centrality and before/after 2008^*

30

Variable	RHA1, before	RHA1 after	RHA2, before	RHA2 after	RHA3, before	RHA3 after	RHA4, before	RHA4 after
Priv_Serv	12.66 (7.76)	10.87 (6.80)	10.08 (6.96)	8.05 (6.13)	8.08 (5.76)	7.78 (5.79)	4.44 (5.79)	2.33 (2.70)
Pub_Serv	23.67 (9.53)	25.31 (10.64)	21.12 (9.38)	22.00 (9.60)	24.11 (7.47)	27.03 (8.93)	29.04 (11.17)	31.17 (10.7
Total_Serv	36.34 (9.62)	36.19 (10.18)	31.20 (8.91)	30.05 (8.14)	32.19 (7.82)	34.81 (8.86)	33.48 (11.75)	33.50 (11.0)
Privtime	1.23 (0.98)	1.25 (0.99)	1.91 (1.37)	1.91 (1.36)	1.78 (1.36)	1.79 (1.37)	6.56 (4.31)	6.67 (4.40)
Pubtime	0.79 (0.82)	0.79 (0.82)	0.83 (0.60)	0.84 (0.60)	1.04 (0.68)	1.05 (0.70)	2.08 (1.62)	2.10 (1.37)
Time_difference	0.44 (0.57)	0.45 (0.58)	1.08 (1.29)	1.07 (1.29)	0.74 (1.11)	0.75 (1.13)	4.48 (4.08)	4.57 (4.10)
Centrality 1	0.12 (0.32)	0.12 (0.32)	0.19 (0.40)	0.19 (0.40)	0.31 (0.46)	0.31 (0.46)	0.31 (0.46)	0.29 (0.45)
Centrality 2	0.08 (0.27)	0.08 (0.27)	0.20 (0.40)	0.20 (0.40)	0.24 (0.43)	0.24 (0.43)	0.05 (0.21)	0.05 (0.21)
Centrality 3	0.13 (0.34)	0.13 (0.34)	0.28 (0.45)	0.28 (0.45)	0.19 (0.39)	0.18 (0.39)	0.05 (0.21)	0.05 (0.21)
Centrality 4	0.13 (0.34)	0.13 (0.34)	0.16 (0.37)	0.15 (0.36)	0.11 (0.32)	0.11 (0.32)	0.31 (0.46)	0.33 (0.47)
Centrality 5	0.37 (0.48)	0.36 (0.48)	0.13 (0.34)	0.13 (0.34)	0.06 (0.24)	0.06 (0.24)	0 (0)	(0) 0
Centrality 6	0.01 (0.08)	0.01 (0.08)	0.01 (0.11)	0.01 (0.11)	0.05 (0.22)	0.05 (0.22)	0.27 (0.45)	0.28 (0.45)
Centrality 7	0.17 (0.38)	0.17 (0.37)	0.02 (0.15)	0.02 (0.15)	0.04 (0.19)	0.04 (0.19)	0.01 (0.11)	0.01 (0.11)
	N=2002,	N=6078,	N=982, n=82,	N=2967,	N=949, <i>n</i> =80,	N=2878,	N=1011,	N=3000,
	<i>n</i> =168,	<i>n</i> =170,	<i>T</i> =11.98	n=83, T=35.75	<i>T</i> =11.86	<i>n</i> =82, <i>T</i> =35.1	n=85, T=11.89	n=87, T=37.
	T=11.92	T=35.75						

31.17 (10.71) 33.50 (11.02)

Table 1C Summary statistics stratified by RHA*

*The first number is mean, the number in parenthesis is the standard deviation. *N* is number of observations, *n* is number of municipalities, and T is mean number of periods. T before=12, T after=36.

n=87, T=37.48

Table 3 Minimum and maximum values of means before and after 2008 (services per capita) and their percentage change A) among centralities and **B**) among Regional Health Authorities

	Before 2008			After 2008				
	Min mean	Max mean	Max-Min	Min mean	Max mean	Max-Min	Change, %	
Priv_Serv	4.08	18.43	14.35	2.30	14.83	12.53	-12.7	
Pub_Serv	17.21	28.98	11.77	18.88	30.87	11.99	1.9	
Total_Serv	30.43	36.74	6.31	32.46	35.50	3.04	-51.8	

3A Minimum and maximum values of means before and after 2008 (services per capita) and their percentage change among centralities

| 3B Minimum and maximum values of means before and after 2008 (services per capita) and their percentage change among Regional **Health Authorities**

	Before 2008			After 2008			
	Min mean	Max mean	Max-Min	Min mean	Max mean	Max-Min	Change, %
Priv_Serv	4.44	12.66	8.22	2.33	10.87	8.54	3.9
Pub_Serv	21.12	29.04	7.92	22.00	31.17	9.17	15.8
Total_Serv	31.20	36.34	5.14	30.05	36.19	6.14	19.5

BMC Family Practice

RESEARCH ARTICLE

Open Access



Disenrollment from general practitioners among chronic patients: a register-based longitudinal study of Norwegian claims data

Anastasia Mokienko^{*} and Knut Reidar Wangen

Abstract

Background: Norwegian general practitioners (GPs) consult on a variety of conditions with a mix of patient types. Patients with chronic diseases benefit from appropriate continuity of care and generally visit their GPs more often than the average patient. Our aim was to study disenrollment patterns among patients with chronic diseases in Norway, because such patterns could indicate otherwise unobserved GP quality. For instance, higher quality GPs could have both a greater share of patients with chronic diseases and lower disenrollment rates.

Methods: Data on 384,947 chronic patients and 3,974 GPs for the years 2009–2011 were obtained from national registers, including patient and GP characteristics, disenrollment data, and patient list composition. The birth cohorts from 1940 and 1970 (146,906 patients) were included for comparison. Patient and GP characteristics, comorbidity, and patient list composition were analyzed using descriptive statistics. Patients' voluntary disenrollment was analyzed using logistic regression models.

Results: The GPs' proportion of patients with a given chronic disease varied more than expected when the allocation was purely random. The proportions of patients with different chronic diseases were positively correlated, partly due to comorbidity. Patients tended to have lower disenrollment rates from GPs who had higher shares of patients with the same chronic disease. Disenrollment rates were generally lower from GPs with higher shares of patients with arthritis or depression, and higher from GPs who had higher shares of patients. This was the same in the comparison group.

Conclusion: Patients with a chronic disease appeared to prefer GPs who have higher shares of patients with the same disease. High shares of patients with some diseases were also negatively associated with disenrollment for all patient groups, while other diseases were positively associated. These findings may reflect the GPs' general quality, but could alternatively result from the GPs' specialization in particular diseases. The supportive findings for the comparison group make it more plausible that high shares of chronic patients could indicate GP quality.

Keywords: Chronic patients, Switching, Primary health care, Schizophrenia, Epilepsy, Diabetes type 1, Diabetes type 2, Asthma, Arthritis, Depression

* Correspondence: anastasiya.mokienko@gmail.com;

anastasiya.mokienko@gmail.com

Department of Health Management and Health Economics, University of Oslo, P.O. Box 1089, Blindern, Oslo 0318, Norway



© The Author(s). 2016 **Open Access** This article is distributed under the terms of the Creative Commons Attribution 4.0 International License (http://creativecommons.org/licenses/by/4.0/), which permits unrestricted use, distribution, and reproduction in any medium, provided you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons Public Domain Dedication waiver (http://creativecommons.org/publicdomain/zero/1.0/) applies to the data made available in this article, unless otherwise stated.

Background

The quality of care for people with chronic diseases often relies on appropriate primary care. Some such patients may need continuous, long-term follow-up and motivation in order to maintain a favorable lifestyle. Others, who experience a condition associated with social stigma, may need time to develop trust in their care providers. Early detection of the chronic disease and its subsequent routine monitoring is also very important to save patients from acute hospitalization and complications from the disease [1]. Comorbidity is a good reason for primary care providers to be better able to manage chronic diseases [2, 3].

Previous studies have found that long-term physicianpatient relationships are beneficial for patients [4, 5] and that patients disenroll from their general practitioner (GP) when they are not satisfied with their GP-patient relationship [6–10]. Patients may also disenroll from their GP if they perceive insufficient quality of care. Accessibility factors, such as adequate time for consultations [11] and availability of appointments [12] are predictors of good quality. Booking intervals for consultations and duration of the consultations themselves are correlated with good management of chronic diseases; the effect was greater for patients with asthma than for those with diabetes or angina, possibly because primary care providers deal more with asthma than diabetes or angina [13].

When it comes to accessibility, earlier research shows that longer patient lists are associated with negative evaluations of accessibility and that the GP's age has a negative association with the evaluation of all aspects, except accessibility [14]. Longer patient lists are also associated with better illness detection [15], which may suggest that practices detecting a higher number of chronic conditions have greater demand from patients due to their systematic chronic disease management [15–18].

A strong connection between patient choice and higher quality of practice, as measured by studying the publicly available data on practice performance, has been reported [19]. A review study found that patients were weakly influenced by publicly available information about provider quality [20]. On the provider side, only hospitals seemed to improve quality as a response to quality indicators being made publicly available [21]. For GPs, patient shortage has been found to correlate with patient dissatisfaction, the GP's communication skills, and other GP characteristics [22–24].

Interaction between chronically ill patients and their GPs has not been given specific attention in previous literature, but a previous study of obese patients may contain clues for generalizable results: reportedly, obese patients avoided physicians they perceived as sources of stigma and searched for providers who were "obese friendly" [25].

If patients switch between GPs until their demands are met, we would expect these patients to be disproportionally distributed across GPs. Similar trends could be expected if the GPs intentionally specialize, formally or not, in a given patient group. However, neither of these mechanisms has obvious implications for the provider choices made by other groups of patients. For example, a GP who is popular among patients with diabetes type 2 (DT2) may also be popular among patients with depression, whereas patients without chronic diseases may be indifferent to this GP's motivational skills. Older patients and patients with chronic diseases have generally higher care continuity, whereas patients with lower care continuity are those living in rural areas, employed, with higher education, or with poorer mental health [26].

Our aim is to investigate patterns of chronic patient disenrollment. This type of study is required because there are no published indicators of GP quality, and therefore these indicators need to be identified through patient actions (such as disenrollment). Moreover, specialized patient choice patterns might suggest an extra argument for using more fee-for-service reimbursement or risk-adjusted capitation for GPs in order to compensate for varying expected workloads depending on their patient list composition. Primary care in Norway is publicly funded with a capitation and fee-for-service system, and patients have to consult their GPs in order to see a specialist. Each individual GP has a patient list and can decide the maximum number of patients that can be enrolled on their list. Patients can switch between available GPs up to three times a year, according to their own preference.

Methods

Data sources and study populations

This is a retrospective study using data from two national registers in Norway, administrated by the Norwegian Directorate of Health, from 2009–2011. Our GP data were obtained from the national register of regular GPs, which covers the entire GP population, and merged with patient data using the GPs' IDs. Our patient data were based on claims data obtained from the KUHR registry (Kontroll og Utbetaling av HelseRefusjon), which covers the entire Norwegian patient population. This registry records claims data continuously but for our analysis, the sample period 2009–2011 was divided into six semiannual intervals. The individual level data included patient characteristics, their consumption of primary care, and the GP with which they were enrolled.

Two samples of patients were selected among patients who visited a GP at least once from 2009–2011. Most of our analysis is based on sample 1, which consisted of patients registered with one or more of the following seven diagnoses at least once during the period 2006–2011: DT1, DT2, asthma, arthritis, schizophrenia, depression, and epilepsy. These patient groups were chosen because they are known to vary substantially both in the number of patients in the population, and in the utilization of primary care services. For instance, patients with DT2 constitute almost 5% of the population and receive most of their health care from their GP, while patients with schizophrenia are fewer and receive more specialist care in a hospital setting.

Our analysis also included a comparison group, sample 2. This group consisted initially of the entire birth year cohorts from 1940 and 1970, but we excluded patients already included in sample 1. Obviously this selection yielded an age distribution different from that in sample 1, but the selection of one elderly and one younger birth year cohort should provide a good basis for comparison.

Initially, the two samples combined contained 988,483 patients (Fig. 1). We excluded 34,189 cases where the disenrollment was likely to be due to causes not relevant for our purpose; that is, when patients moved to another municipality, or when a GP moved, retired, or died. For the logistic regressions, we excluded patients living in municipalities with less than 50,000 inhabitants in order to focus on patients who could choose from several GPs. This left us with 316,636 patients in sample 1 and 32,311 patients in sample 2 (348,947 in total). Finally, we excluded patients with irregular medical records, mainly missing birth year or sex, yielding 313,659 patients in sample 1 and 30,212 patients in Sample 2 (343,871 in total).

Measures

Our main outcome variable, 'SwitchOut', measured whether a patient disenrolled from a GP from one semiannual period to the subsequent period. Definitions of independent variables are summarized in Table 1. Information about the GPs' age, sex, specialization, and list length, and patients' sex, birth year, and number of visits was obtained directly from the data registries. The variable 'Pat comorb' was given the value 0 for patients in sample 2, while for each patient in sample 1 we counted the number of registered diseases (1-7) and subtracted 1 from this number. This yielded a variable with a range between 0 and 6. The variables 'Diab2_share' and 'Epil_share' measure a GP's share of patients with the respective chronic disease, but with a slight adjustment: if shares were calculated straightforwardly, they could potentially be influenced by the health status of a single patient, because some chronic diseases are relatively rare and some GPs had fewer patients (shorter lists). To illustrate, consider a GP who has 100 patients, of which one has epilepsy. If we take the perspective of the GP, the share of patients with epilepsy is slightly above average (Table 1). However, this measure is of little relevance if we take the perspective of the patient with epilepsy: the GP has no other patients with epilepsy. To avoid interpretational ambiguity, we chose to take the patients' perspective. For each patient-GP pair, we excluded the patient from the calculation of the GP's share. Thus, the share variables mostly showed the variation between GPs but also some variation within a GP practice.



Variable	Definition	Sample 1	(N = 313,659))	Sample 2	(N = 30,212)	
		Median	Mean	St.dev	Median	Mean	St.dev
DT1_share	The share of a GP's patients with diabetes type 1	0.005	0.007	0.006	0.005	0.006	0.005
DT2_share	The share of a GP's patients with diabetes type 2	0.042	0.046	0.027	0.036	0.040	0.021
Arth_share	The share of a GP's patients with arthritis	0.014	0.016	0.010	0.013	0.015	0.009
Asthm_share	The share of a GP's patients with asthma	0.020	0.023	0.015	0.018	0.020	0.013
Depr_share	The share of a GP's patients with depression	0.107	0.112	0.042	0.094	0.100	0.038
Schi_share	The share of a GP's patients with schizophrenia	0.004	0.005	0.003	0.004	0.005	0.003
Epil_share	The share of a GP's patients with epilepsy	0.008	0.009	0.004	0.008	0.008	0.004
ListLength	The number of patients on a GP's list	1423	1444.0	367.8	1439	1453.4	367.8
Ln_ListLength	The natural logarithm of Listlength	7.261	7.240	0.277	7.272	7.248	0.270
GP_Age	The GP's age	52	50.358	9.120	51	49.744	8.989
GP_Sex	=1 if the GP is male, =0 otherwise	1	0.706	0.455	1	0.673	0.469
GP_age · GP_Sex	The product of GP_Age and GP_Sex	48	36.473	24.718	45	34.266	25.016
GP_Specialist	=1 if the GP has a specialist degree in general medicine; =0 otherwise	1	0.707	0.455	1	0.702	0.457
Pat_Sex	=1 if the patient is male; =0 otherwise	0	0.426	0.494	0	0.494	0.500
Pat_BirthYear	The patient's year of birth	1959	1958.6	19.1	1970	1961.5	13.5
Pat_Comorb	Sample 1: No. of chronic diseases minus one. Sample 2: Not defined	0	0.148	0.405	-		
Pat_Visits	The patient's number of visits to primary care	3	4.662	5.268	1	2.227	3.369
Pat_Visits_win	Winsorized Pat_Visits at 99^{th} percentile (max = 23)	3	4.570	4.626	1	2.205	3.107
Pat_Visits_dum	=1 if Pat_Visit >23, =0 otherwise	0	0.10	0.98	0	0.002	0.047

Table 1 Variable definitions and descriptive statistics on the patient level¹

¹Municipalities over 50 000. First half of 2009

In order to avoid highly influential outliers, we transformed two variables. The distribution of GPs' list length was skewed so we transformed the variable using the natural logarithm. The distribution of patients' number of visits to primary care was also skewed, and for this variable, we winsorized the distribution at the 99th percentile (23 visits per period) and included a dummy variable for observations that exceeded this limit.

Statistical analyses

We inspected the data numerically and graphically at both the patient and GP levels. This included graphs intended to reveal whether the distribution of chronic patients seemed disproportionate across GPs. On the GP level, the mean proportion of patients with DT2 was 4.5% in the first half of 2009. If patients were allocated by pure chance, a randomly selected GP's share of patients with DT2 would have the expected value of about 4.5%, and be approximately normally distributed for a sufficiently long patient list (>60 patients). For data at the GP level, we calculated Spearman's correlation coefficients for the various GP-related variables, including the shares of patients with different diagnoses, the GP's age and sex. We defined sub-samples of patients from sample 1 based on the seven chronic diseases. These sub-samples partly overlapped due to comorbidity. For each sub-sample, the shares of patients with 1 of the other six diseases were calculated.

We then used logistic regressions to model patients' disenrollment from their GP. The modeling was performed for each patient category separately: on the subsamples from sample 1, as defined above, and sample 2. Because the dependent variable (SwitchOut) was based on observations from two consecutive periods, we had up to five effective observations for each patient. For the independent variables, we used observations from the first five periods. The set of independent variables included those from Table 1, and an interaction term between GPs' age and sex. We incorporated the longitudinal data structure by including patient-specific effects (intercepts) in the models. Patient-specific effects can account for unobserved factors, such as ethnicity or educational background, as long as these factors remain constant throughout the sample period. The models were estimated using xtlogit in Stata 13, under the standard assumptions that the patient-specific effects were normally distributed and did not correlate with the independent variables. Fixed effect models, which allow the patient-specific effects to be non-normally distributed or correlated with the independent variables, were also
considered. However, in fixed effect models the timeinvariant patient variables for sex and birth year would, by construction, be excluded from the estimations.

Results

Descriptive statistics

According to Table 1 and Fig. 2, the proportion of patients with DT2 varied substantially among GPs. If these patients had been allocated purely by chance, about 95% of the proportions would lie between the red curves in Fig. 2, but this was not the case. In fact, only 46.5% of the proportions were positioned within the red curves. For the other diagnosis groups, the corresponding patient shares also seemed disproportionally distributed.

Overall, 4.5% of chronic patients disenrolled from their GP from one period to the next, but the share varied from 3.7% among patients with DT2 to 6.2% among patients with schizophrenia (Table 2). Among patients in sample 2, the share that disenrolled was 3.7%.

Descriptive statistics for the independent variables used in the logistic regressions are reported in Table 1, separately for samples 1 and 2. Due to the sample selection procedure, the average GP characteristics differ somewhat from those obtained for the full GP population, where 66% were men, the average age was 48 years, and the average patient list length was 1200 (N = 3940).

The distribution of the variable 'ListLength' appeared continuous but was somewhat skewed to the right. The distribution of 'Pat_visits' was markedly right-skewed, and the distribution's tail was rather scattered: for sample 2, the 75th, 95th, and 99th percentiles were 6, 14, and 23, respectively, but the maximum value was as high as 219.

Table 3 presents the sizes of the sub-samples defined for the seven chronic diseases. The most frequent of the diseases was depression (N = 488,686), while schizophrenia



Table 2 Share of patients who voluntarily disenrolled from their GPs, between the 1^{st} and 2^{nd} halves of 2009.¹

Sample	Sub set	Ν	%
Sample 1	Full sample	313,659	4.52
	DT1	11,292	4.99
	DT2	74,473	3.75
	Schizo	8,316	6.29
	Depr	186,415	5.00
	Arthr	27,157	4.00
	Asthm	37,110	4.16
	Epil	15,403	4.86
Sample 2	Full sample	30,212	3.76

¹Municipalities over 50 000

was the least frequent (N = 21,368). In the sub-sample of patients with depression (third column from the left), 1.3% also suffered from schizophrenia. Among patients with schizophrenia (rightmost column), 28.7% also suffered from depression. A substantial number of patients were recorded with both DT1 and DT2, likely due to registration errors or diagnostic uncertainty.

We calculated Spearman's rank correlation coefficients for the GP proportion of patients with a given chronic disease and other patient proportions and GP characteristics, as shown in Table 4. The correlation coefficient of 'Asthm_share' and 'DT1_share' was 0.648, indicating that GPs with a high proportion of patients with asthma also tended to have a high proportion of patients with DT1. All variables related to the GPs' proportions of patients were significantly different from zero. The proportion of patients with chronic diseases were all positively correlated, and negatively correlated with the proportion of other patients ('Other_Share'). 'Other_Share' was negatively correlated with 'GP_Age' and 'GP_Sex', indicating that older GPs and male GPs tended to have fewer patients without our seven chronic diseases.

Logistic regression analysis

Table 5 shows the estimated parameters of the logistic regressions where 'SwitchOut' is the dependent variable, the independent variables are those listed in Table 1, and Sigma_u denotes the standard deviation of the patient-specific intercepts. The first seven columns show results based on sample 1 according to patient diagnosis group; the last column is based on sample 2. In logistic regressions, the coefficients can be used to compare the difference in log-odds ratios between groups, so that a patient sex coefficient of -0.188 (arthritis patients) represents the difference in log-odds ratios between male and female patients. The corresponding difference in odds ratios is obtained by taking the anti-log, exp(-0.188) = 0.829.

	Arthritis	Asthma	Depression	DT2	DT1	Epilepsy	Schizophrenia
Arthritis		4.4	2.7	3.9	4.0	2.0	1.3
Asthma	6.1		4.5	6.7	5.8	3.6	6.0
Depression	14.5	17.6		13.5	15.2	15.8	28.7
Diabetes type 2	10.0	12.4	6.4		77.8	5.7	12.0
Diabetes type 1	1.6	1.7	1.1	12.0		1.3	1.9
Epilepsy	1.0	1.3	1.5	1.1	1.6		3.1
Schizophrenia	0.3	1.0	1.3	1.1	1.1	1.4	
Ν	90,095	124,776	488,686	232,383	35,887	46,145	21,368

 Table 3
 Percent of patients with a chronic disease (column) that have another chronic disease (row)

³First half year of 2009. Patient level data. Sample 1 without restrictions (neither on municipality size, data irregularity or moving). N is the number of patients with the chronic disease

The statistical inference for this type of model is based on large-sample theory and coefficient estimates are approximately normally distributed. Thus, to simplify the presentation, we do not report p-values as they can be derived from the estimated standard errors.

Some of the estimated effects of the patient share variables were relatively robust across patient groups. For 'Arth_share', all coefficients were significantly negative, implying that all patient groups tended to have lower disenrollment from GPs with relatively high shares of patients with arthritis. For 'Asthm_share' and 'Depr_share', all of the significant coefficients were also negative. In contrast, for 'DT1_share', 'Epil_share' and 'Schi_share', almost all significant effects were positive.

We can distinguish two main effects. First, the "own share effect," namely, all patient groups tended to remain with GPs who had a high share of patients with the same diagnosis. Second, the "cross share effect," where, for

Table 4 GP characteristics. Spearman's correlation coefficients with two-sided *p*-values.²

	Arth_ share	Asthm_ share	Depr_ share	DT1_ share	DT2_ share	Epil_ share	Schi_ share	Other_ share	GP_ age	GP_ sex	List Length
Asthm_share	0.488										
	0.000										
Depr_share	0.195	0.264									
	0.000	0.000									
DT1_share	0.519	0.648	0.221								
	0.000	0.000	0.000								
DT2_share	0.232	0.310	0.121	0.332							
	0.000	0.000	0.000	0.000							
Epil_share	0.270	0.298	0.205	0.335	0.177						
	0.000	0.000	0.000	0.000	0.000						
Schi_share	0.045	0.175	0.227	0.135	0.183	0.162					
	0.005	0.000	0.000	0.000	0.000	0.000					
Other_share	-0.562	-0.683	-0.762	-0.712	-0.362	-0.406	-0.285				
	0.000	0.000	0.000	0.000	0.000	0.000	0.000				
GP_Age	0.203	0.137	0.064	0.213	-0.047	0.091	-0.028	-0.174			
	0.000	0.000	0.000	0.000	0.003	0.000	0.077	0.000			
GP_Sex	0.181	0.293	0.077	0.318	0.101	0.205	0.135	-0.265	0.249		
	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000		
ListLength	-0.020	-0.069	0.041	-0.032	-0.145	-0.040	-0.033	0.035	0.166	0.172	
	0.205	0.000	0.010	0.046	0.000	0.011	0.038	0.026	0.000	0.000	
GP_Specialist	0.008	0.017	0.030	0.037	-0.133	0.067	-0.003	-0.018	0.365	0.098	0.226
	0.618	0.275	0.063	0.020	0.000	0.000	0.860	0.250	0.000	0.000	0.000

 2 GP level data for first quarter of 2009, N = 3974. Correlation coefficients with two-sided p-values less than 1% are in boldface

	Arthritis	Asthma	Depression	Diabetes type 2	Diabetes type 1	Epilepsy	Schizophrenia	Others
Arth_share	-15.032	-10.550	-16.792	-9.506	-16.905	-16.495	-20.113	-15.310
	(1.611)	(1.597)	(0.815)	(1.194)	(3.116)	(2.836)	(3.925)	(2.185)
Asthm_share	-4.381	-10.406	-2.117	1.883	-1.624	-0.188	-3.895	0.093
	(1.598)	(1.309)	(0.636)	(0.934)	(2.494)	(2.262)	(2.922)	(1.799)
Depr_share	-1.915	-2.343	-5.377	-2.781	-0.484	-2.029	-1.095	-0.220
	(0.445)	(0.392)	(0.165)	(0.278)	(0.648)	(0.590)	(0.752)	(0.457)
DT2_share	-0.875	1.260	-0.534	-4.117	-0.499	-0.886	2.397	0.112
	(0.855)	(0.738)	(0.349)	(0.459)	(1.347)	(1.207)	(1.524)	(0.986)
DT1_share	16.725	11.661	15.525	7.841	-20.177	15.491	10.100	15.962
	(3.049)	(2.576)	(1.147)	(1.691)	(4.069)	(4.042)	(5.592)	(3.473)
Epil_share	9.578	11.917	4.069	4.048	-9.185	-13.955	-1.462	-0.165
	(4.637)	(3.910)	(1.695)	(2.815)	(6.681)	(5.882)	(7.709)	(4.754)
Schi_share	23.551	28.298	37.453	39.029	21.821	39.502	1.307	29.586
	(5.265)	(4.248)	(1.810)	(3.082)	(7.191)	(6.259)	(7.663)	(5.136)
Ln_ListLength	-0.702	-0.631	-0.405	-0.658	-0.346	-0.489	-0.205	-0.623
	(0.053)	(0.047)	(0.019)	(0.033)	(0.076)	(0.069)	(0.090)	(0.052)
GP_Age	0.032	0.029	0.033	0.035	0.032	0.033	0.033	0.033
	(0.003)	(0.003)	(0.001)	(0.002)	(0.005)	(0.004)	(0.006)	(0.003)
GP_Sex	-0.367	-0.512	-0.202	-0.234	-0.138	-0.108	-0.390	-0.317
	(0.189)	(0.166)	(0.065)	(0.118)	(0.265)	(0.235)	(0.306)	(0.175)
GP Age ^a Sex	0.010	0.013	0.006	0.007	0.009	0.004	0.010	0.008
	(0.004)	(0.003)	(0.001)	(0.002)	(0.005)	(0.005)	(0.006)	(0.004)
GP_Specialist	-1.148	-1.271	-1.145	-1.288	-1.119	-1.236	-1.189	-1.242
	(0.035)	(0.030)	(0.012)	(0.021)	(0.050)	(0.044)	(0.056)	(0.033)
Pat_Sex	-0.188	-0.090	-0.100	-0.082	-0.133	0.015	-0.163	0.040
	(0.035)	(0.028)	(0.012)	(0.020)	(0.047)	(0.041)	(0.055)	(0.032)
Pat_BirthYear ^a	0.007	0.007	0.012	0.007	0.003	0.007	0.012	0.195
	(0.001)	(0.001)	(0.000)	(0.001)	(0.001)	(0.001)	(0.002)	(0.036)
Pat_Comorb	0.135	0.103	0.096	0.162	0.169	0.193	0.214	
	(0.027)	(0.022)	(0.013)	(0.016)	(0.034)	(0.035)	(0.038)	
Pat_Visits_win	0.042	0.046	0.049	0.041	0.035	0.049	0.046	0.057
	(0.003)	(0.003)	(0.001)	(0.002)	(0.005)	(0.004)	(0.005)	(0.004)
Pat_Visits_dum	0.046	-0.208	-0.270	-0.327	-0.141	-0.157	-0.212	-1.019
	(0.116)	(0.087)	(0.046)	(0.083)	(0.158)	(0.134)	(0.143)	(0.333)
Cons	-12.977	-13.367	-25.141	-13.052	-7.795	-14.111	-26.068	-0.306
	(1.863)	(1.736)	(0.738)	(1.283)	(2.508)	(2.090)	(3.519)	(0.402)
Sigma_u	0.718	0.784	0.773	0.747	0.755	0.809	0.922	0.662
	(0.040)	(0.032)	(0.013)	(0.024)	(0.051)	(0.043)	(0.049)	(0.042)
No. obs	130,690	175,010	890,215	357,153	53,206	73,419	39,535	146,906
No. patients	27,157	37,110	186,415	74,473	11,292	15,403	8,316	30,212

Table 5 Logistic regression for patients' voluntary disenrollment from GPs, separate for patient groups.⁴Estimated parameters (standard errors)

⁴Dependent variable: 'SwitchOut'. Only patients living in cities with more than 50,000 inhabitants were included. The seven left columns are from sample 1, the far right column is from sample 2. ^aFor 'Others', 'Pat_BirthYear' was replaced with a dummy variable equal to 0 for patients born in 1940 and equal to 1 for patients born in 1970. Each patient was observed up to five times. Sigma_u denotes the estimated standard deviation of the random patient-specific constant terms. Stata 13, the xtlogit procedure, was used in the estimations. Estimates with two-sided *p*-values < 1% are in boldface

instance, a high share of DT1 patients increased the switch-out for patients with arthritis (meaning, patients with arthritis were more likely to switch-out if their GPs had more patients with DT1). The cross share effect was not generally symmetric as a high share of patients with arthritis reduced the switch-out for patients with DT1.

For all GP and patient characteristics, the significant coefficients had the same sign across all patient groups. Patients tended to switch less often from GPs who had long patient lists ('Ln_ListLength') or who were specialists in general medicine ('GP_Specialist'). For older, female GPs, patients tended to switch out more often ('GP_Age'). This effect was even stronger for male GPs, for which the full effect of age is obtained by adding the coefficients of 'GP_age' and the interaction between a GP's age and sex ('GP_Age*GP_Sex').

Patients born more recently (i.e., lower 'Pat_BirthYear') or who had more comorbidities ('Pat_Comorb') tended to switch GPs more often. The 1% of patients who most frequently used primary care (i.e., 'Pat_Visits_dum' = 1) tended to switch less often than patients who had fewer visits. However, among the remaining 99% of patients, those with a higher number of primary care visits ('Pat_visits_win') tended to switch more often.

The patient-specific effects are assumed to be normally distributed, with a zero mean and an estimated standard deviation, Sigma_u. For patients with arthritis, the value of Sigma_u can be interpreted as the difference in logodds between a patient who has a patient-specific intercept one standard deviation from the mean (0.718) and a patient with an intercept equal to the mean value (zero). This is about four times the numerical value of the coefficient for patient sex, and it corresponds to a difference in odds ratio equal to 2.050. In all patient groups, the estimated value for Sigma_u indicates that the unobserved patient characteristics have a comparably large influence on disenrollment.

Discussion

Our data indicate that patients with chronic diseases are not allocated to GPs by chance alone (Fig. 2). One explanation could be that some GPs informally specialize, for example in DT2, and thus are able to establish and maintain a "stock" of such patients. In so doing, the patient comorbidity shown in Table 3 would imply a tendency for these GPs to also have relatively higher shares of patients with arthritis and asthma. Moreover, patients with chronic diseases tend to have comorbidities, contributing to their GPs having shares of patients with different diagnoses. This could partly explain why the proportions of chronic disease types are all positively correlated, as shown in Table 4.

The coefficients in Table 5 suggest that chronic patients disenroll less often from GPs who have a high share of

patients with the same diagnosis; for instance, 'Arth share' has a negative effect (-15.032) for patients with arthritis, and 'Asthm_share' has a negative effect (-10.406) for patients with asthma. Again, this may be the result of GPs informally specializing in certain types of patients with chronic diseases. It may also result from the GPs' general qualities such as organizational skills, communication abilities, or empathic attitudes. It has been suggested that such patterns may result from patients' negative interactions with healthcare providers, so that, for instance, obese patients search for "obese friendly" physicians [25]. Patients could also make use of informal conversations (word-of-mouth) with family, friends, or colleagues that recommend one GP or another, which seems to have a greater effect on the choice of GP than public information disclosure [20]. The relationship between the GP and patient could also be a factor in patient choice, since chronic patients spend more time in primary care and would change their GP if they were not satisfied [3, 4]. We can assume that GPs who have high numbers of patients with a particular disease might have a particular practice style, which also attracts these patients, but these mechanisms may be complex, for instance for patients with schizophrenia. In Table 5, the only exception from the general pattern is for patients with schizophrenia, for which the effect of 'Schi_share' is insignificant. However, all other patient groups tend to disenroll more from GPs with high shares of patients with schizophrenia, potentially suggesting that these GPs are less popular in general, and this may perhaps counter the "own share effect" among patients with schizophrenia.

We find that all or most patient groups tend to disenroll less from GPs who have high shares of patients with arthritis, depression, and asthma. We assume that this disenrollment pattern happens due to qualities of GPs that attract most patients, such as good communication and care coordination skills. For chronic patients who are intensive users of primary care it is important to find a GP that fits their needs, so they might change until they find the right match. Patients in the comparison group have, per se, no obvious reason to prefer GPs who specialize in any chronic disease, but it is likely they have preferences regarding GP qualities. Thus, our finding that in some cases the preferences of the comparison group and of the patients with chronic diseases align suggests that GPs' shares of chronic patients reveals information about these GPs' general qualities.

A puzzling finding is that all or most patient groups tend to disenroll more from GPs who have high shares of patients with DT1 and schizophrenia. According to Norwegian guidelines, these two patient groups' follow-up happens in secondary care, in contrast to our other patient groups. Patients who receive follow-up in secondary care could perhaps be more indifferent to which GP they visit for other acute illnesses. If so, they may be satisfied with GPs who have a practice style favoring patients who can be treated expediently over patients who need long-term follow-up. With this interpretation, the high disenrollment among patients with schizophrenia (Table 2) can be interpreted not necessarily as a search for a GP who is well-suited for handling issues related to schizophrenia but perhaps as an expression of other, shorter-term considerations.

GP specialization in general medicine has a negative relationship with disenrollment, suggesting that patients prefer to stay with specialized GPs. List length also has a negative relationship with disenrollment for all patient groups, except for patients with schizophrenia. Previous studies have found that non-chronic patients stay with GPs with shorter patient lists, meaning that they value accessibility [10-12], in contrast to chronic patients who value long patient lists, which is associated with higher disease detection [13]. GP's age is positively related with disenrollment for all patient groups, suggesting that patients in general may prefer younger GPs. This effect of age is supported by earlier findings [12]. For patients with arthritis, asthma, depression or DT2, this tendency is stronger for male than female GPs, perhaps because there are fewer women among older GPs than among younger GPs. In most patient groups, disenrollment was not significantly associated with GP sex, except patients with asthma and depression, who tend to less often disenroll from male GPs.

In all groups of patients with chronic disease, disenrollment increased with the number of comorbidities. This is consistent with the discussion above, given that management of patients with comorbidities is challenging for primary care providers [27]. Our selection of patient groups was not, however, designed to investigate the effect of comorbidities in particular. Future studies should consider including other diagnoses, such as cardiovascular disease and cancer. A higher number of visits to primary care also tended to increase disenrollment, but the negative coefficients for the dummy variable, identifying patients who had more than 23 visits in a six month period, may indicate that the relationship between disenrollment and the number of visits is not linear. Younger patients generally disenroll more often and, except for patients with epilepsy and other patients (sample 2), male patients disenroll less often.

This study has three main imitations: first, although the majority of the numerical data seemed reliable, we found that as many as 77.8% of patients with DT1 were also registered as having DT2. Such "double diabetes" cases are not uncommon [28, 29], but it is likely that most of the cases in our data are due to diagnostic uncertainty or registration errors. This may affect both the results related to the share of patients with diabetes ('DT1 share' and 'DT2 share'), and the results for subsamples defined for patients with DT1 and DT2. Second, our data did not include potentially relevant patient variables such as cultural background, native language, income, educational background, or marital status. Disease severity and proper control of symptoms could also influence disenrollment behavior. To an extent, our random effect logistic regressions can account for timeinvariant patient variables, but future studies should consider including more variables in order to assess their influence. Additional information about the GPs, such as cultural background, length of time in practice, and professional interests would also have been of interest. Third, the age distribution differs between our selected comparison group, sample 2, and our main sample of interest, sample 1. Sample 2's age distribution also differs from the age distribution across all groups in the full population without our specified chronic diseases. This means that the estimates for sample 2 in Tables 2 and 5 are likely to be biased, if interpreted as estimates for the full population. We believe that the qualitative aspects of these results would not be very different in the full population, but this is of course a conjecture. Future register-based studies should consider obtaining a comparison group with similar age distribution as the sample of main interest, for instance by drawing patients randomly from the entire population.

The data sets used in our logistic regressions were restricted with respect to municipality size. In smaller municipalities, patient options for disenrollment will be more limited by the fact that there are fewer local GPs to choose from. It is likely that including patients irrespective of municipality size would yield estimated effects less pronounced than those reported here - that is, compared to the full population, our result are likely to be biased away from zero. We also excluded observations where observed disenrollment seemed to be due to causes other than patients' preferences for GPs. Patients and GPs who move, or GPs who retire or die, are likely to have demographic characteristics (e.g., age) that differ systematically from the distributions in the full patient and GP populations. It is more difficult to predict how including these observations would have influenced our results, but it would at least have complicated the interpretations.

Conclusions

The following conclusions can be drawn from our findings: 1) patients with chronic diseases are not allocated to GPs only by chance; 2) chronic patients that use primary care intensively disenroll less often from GPs who have a high share of patients with the same diagnosis; and 3) most patient groups tend to remain with GPs with a greater share of arthritis, asthma, and

depression patients, which can indicate better quality care for these and other patient groups. These conclusions are distinct from the findings in the literature.

To investigate this further, more objective quality measurements should be obtained, such as adherence to treatment guidelines, surveillance of treatment outcomes for chronic patients, and user satisfaction in general. If objective quality differences are found, further assessments could be warranted, for instance, whether the current reimbursement system has an appropriate balance between capitation and fee-for service, or whether capitation should be risk-adjusted based on shares of patient types.

Abbreviations

DT1: Type 1 diabetes; DT2: Type 2 diabetes; GP: General practitioner; KUHR: Control and payment of reimbursements to health service providers (Kontroll og Utbetaling av HelseRefusjon)

Acknowledgements

We wish to thank Tor Iversen for many valuable comments. The Norwegian Directorate of Health provided data.

Funding

The Research Council of Norway, project no. 204677/H10, provided funding.

Availability of data and materials

The datasets analyzed during the current study were obtained from the Norwegian Directorate of Health. Due to legal restrictions we are prevented from making the data publicly available or otherwise sharing individual level data. For access to such data for research purposes, please contact the Norwegian Directorate of Health directly.

Authors' contributions

AM conceived the study, prepared the data and performed the statistical calculations. Both AM and KRW participated in the study design, interpretation of results, and writing of the manuscript. The final manuscript has been read and approved by both AM and KRW.

Competing interests

There author declare that there have no competing interests.

Consent for publication

The patient data were obtained from public registers and patient consent for publication was not required.

Ethics approval and consent to participate

The patient data were obtained from public registers and patient consent to participate was not required. The project has been approved by the Regional Committees for Medical and Health Research Ethics (REC South East), project no. 2011/1708.

Received: 21 June 2016 Accepted: 2 December 2016 Published online: 15 December 2016

References

- Starfield B. Commentary on regular primary care lowers hospitalisation risk and mortality in seniors with chronic respiratory disease. J Gen Intern Med. 2010;25(8):758–9.
- Harris MF, Zwar NA. Care of patients with chronic disease: the challenge for general practice. Med J Aust. 2007;187(2):104–7.
- Starfield B, Lemke KW, Herbert R, Pavlovich WD, Anderson G. Comorbidity and the use of primary care and specialist care in the elderly. Ann Fam Med. 2005;3(3):215–22.
- Kaplan SH, Greenfield S, Ware Jr JE. Assessing the effects of physicianpatient interactions on the outcomes of chronic disease. Med Care. 1989; 27(3 Suppl):S110–27.

- Saultz JW, Lochner J. Interpersonal continuity of care and care outcomes: a critical review. Ann Fam Med. 2005;3(2):159–66.
- Safran DG, et al. Switching doctors: predictors of voluntary disenrollment from a primary physician's practice. J Fam Pract. 2001;50(2):130–6.
- Nagraj S, et al. Changing practice as a quality indicator for primary care: analysis of data on voluntary disenrollment from the English GP Patient Survey. BMC Fam Pract. 2013;14:89.
- Kasteler J, et al. Issues underlying prevalence of "doctor-shopping" behavior. J Health Soc Behav. 1976;17(4):329–39.
- Gray PG, Cartwright A. Choosing and changing doctors. Lancet. 1953; 265(6799):1308–9.
- Marquis MS, Davies AR, Ware Jr JE. Patient satisfaction and change in medical care provider: a longitudinal study. Med Care. 1983;21(8):821–9.
- 11. Rubin G, et al. Preferences for access to the GP: a discrete choice experiment. Br J Gen Pract. 2006;56(531):743–8.
- Campbell JL, et al. Accessing primary care: a simulated patient study. Br J Gen Pract. 2013;63(608):e71–6.
- Campbell SM, et al. Identifying predictors of high quality care in English general practice: observational study. BMJ. 2001;323(7316):784–7.
- Heje HN, et al. Doctor and practice characteristics associated with differences in patient evaluations of general practice. BMC Health Serv Res. 2007;7:46.
- Anwar MS, et al. Chronic disease detection and access: does access improve detection, or does detection make access more difficult? Br J Gen Pract. 2012;62(598):e337–43.
- Soljak M, et al. Variations in cardiovascular disease under-diagnosis in England: national cross-sectional spatial analysis. BMC Cardiovasc Disord. 2011;11:12.
- 17. Norbury M, et al. Time to care: tackling health inequalities through primary care. Fam Pract. 2011;28(1):1–3.
- Mercer SW, Watt GC. The inverse care law: clinical primary care encounters in deprived and affluent areas of Scotland. Ann Fam Med. 2007;5(6):503–10.
- Santos R, Gravelle H, and Propper C. Does quality affect patients' choice of doctor? Evidence from England. Econ J. 2016. doi:10.1111/ecoj.12282.
- Marshall MN, Shekelle PG, Leatherman S, Brook RH. The public release of performance data: what do we expect to gain? A review of the evidence. Jama. 2000;283(14):1866–74.
- 21. Marshall M, McLoughlin V. How do patients use information on health providers? BMJ. 2010;341:c5272.
- Lurås H. The association between patient shortage and patient satisfaction with general practitioners. Scand J Prim Health Care. 2007;25(3):133–9.
- 23. Williams SJ, Calnan M. Key determinants of consumer satisfaction with general practice. Fam Pract. 1991;8(3):237–42.
- Sitzia J, Wood N. Patient satisfaction: a review of issues and concepts. Soc Sci Med. 1997;45(12):1829–43.
- Gudzune KA, Bleich SN, Richards TM, Weiner JP, Hodges K, Clark JM. Doctor shopping by overweight and obese patients is associated with increased healthcare utilization. Obesity. 2013;21(7):1328–34.
- Kristjansson E, et al. Predictors of relational continuity in primary care: patient, provider and practice factors. BMC Fam Pract. 2013;14:72.
- 27. Wallace E, et al. Managing patients with multimorbidity in primary care. BMJ. 2015;350:h176.
- Pozzilli P, et al. Obesity, autoimmunity, and double diabetes in youth. Diabetes Care. 2011;34 Suppl 2:S166–70.
- Cleland SJ, et al. Insulin resistance in type 1 diabetes: what is 'double diabetes' and what are the risks? Diabetologia. 2013;56(7):1462–70.